

https://theses.gla.ac.uk/

Theses Digitisation:

https://www.gla.ac.uk/myglasgow/research/enlighten/theses/digitisation/

This is a digitised version of the original print thesis.

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission in writing from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten: Theses <u>https://theses.gla.ac.uk/</u> research-enlighten@glasgow.ac.uk

#### SUMMARY

This Thesis is divided into four distinct and self-contained parts whose unifying theme is implicit in the general title "Chemical Studies of Some Natural Products".

Part I discusses possible biogenetic routes to the vesicant principle cantharidin in which it is considered that this compound is terpenoid in origin. It indicates the expected radio-active labelling patterns that would be encountered should feeding experiments with <sup>14</sup>C labelled mevalonic acid to the beetle <u>Meloe proscarabeus L</u>. lead to the incorporation of radio-activity in the cantharidin shown to be aynthesised by this insect. Chemical degradations of the theoretically possible labelled cantharidins designed to distinguish between them are discussed and a survey of the occurrence of cantharidin within the family Meloidae is given.

Part II describes an attempt to prepare, from a consideration of certain theoretical factors which are briefly discussed, a compound with potentially ProQuest Number: 10662714

All rights reserved

INFORMATION TO ALL USERS

The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10662714

Published by ProQuest LLC (2017). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346

high neuromuscular blocking activity, 3a, 17c-bis (trimethylammonium)-5-a-androstane, by  $S_N^2$  replacement of the methanesulphonyloxy groups of 3 $\beta$ , 17 $\beta$ -dimethanesulphonyloxy-5a-androstane with trimethylamine.

Part III describes a physico-chemical approach to the elucidation of the structure of the sesquiterpene  $\gamma$ -lactone, aristolactone occurring in <u>Aristolochia</u> <u>reticulată</u> L and <u>A. serpentaria</u> L. Successful degradation to the parent hydrocarbon germacrane (as a mixture of diastereoisomers) is described -these results permitting unsquivocal assignment of the basic carbon skeleton of aristolactone. Re-examination of accumulated chemical evidence together with a number of physical measurements (I-R and n.m.r.) are employed to give a probable structure for aristolactone. A cyclisation of methyl oxoaristate, a  $\gamma$ -keto ester derived from aristolactone, has been effected giving rise to a new derivative possessing an as yet undetermined bicyclic system.

Part IV describes the elucidation of the structures of the quaternary alkaloid petaline chloride and its methine base, lecuticine - compounds isolated from

Leontice leontopetalum L. An attempt to synthesise one of the degradation products derived from the methine base leonticine is described but the projected synthesis failed at the last stage.

and a gradient and and an and an

and the second second

Chemical Studies of Some Natural Products

A Thesis Submitted To The University of Glasgow

For the degree of

DOCTOR of PHILOSOPHY

in the

Faculty of Science

by

Sidney J. Smith, B.S.P., M.Sc. (Sask.)

December, 1963.

Department of Pharmacy Royal College of Science and Technology, Glasgow, C.1.

#### ACKNOWLEDGEMENTS

I wish to thank Professor J.B. Stanlake and Emeritus Professor J.P. Todd for the opportunity to carry out this research. I also wish to thank Professor P. Pauson for making freely available the facilities of the Department of Chemistry of the Royal College of Science and Technology.

To Professor J.B. Stenlake and Dr. M. Martin-Smith, I express my appreciation for suggesting the problems investigated in this Thesis. The constant encouragement generously given by Dr. M. Martin-Smith in the course of the work is gratefully appreciated.

I wish to acknowledge the assistance kindly given by Dr. P. Bladou of this college and by Dr. A. Melera of Varian A.G., Switzerland in measuring nuclear magnetic resonance spectra; by Mrs. F. Lawrie of the Department of Chemistry, University of Glasgow for measuring infra-red spectra; by Dr. G. Eglinton also of the Chemistry Department, University of Glasgow for certain spectral interpretations and for permission to use the high efficiency gas-liquid chromatography equipment; by Professor V. Herout of the Caschoslovak Academy of Sciences, Pregue, in very kindly supplying a sample of Buthentic germacrane; by Dr. R. Reed of the Chemistry Department, University of Glasgow, for running mass spectral molacular weight determinations; and by Dr. T.C. Muir of the Experimental Pharmacology Division, Institute of Physiology, University of Glasgow, for performing the pharmacological assays. My thanks are due to all my colleagues in the Division of Pharmaceutical Chemistry of the Department of Pharmacy of the Royal College of Science and Technology, especially to Dr. W.D. Williams, for their helpful criticisms and discussions, to the technical staff of this Division and to Miss J. Murdoch for providing food for the body whilst my supervisors provided food for thought.

Finally, I wish to thank the Board of Governors of the Royal College of Science and Technology for College Residue Fund Scholarships held in the period 1960-1962, the trustees of the Professor J.P. Todd Research Fund for a scholarship held in the year 1962-1963, and to the Pfizer Company for the Pfizer Research Prize.

#### SUPPLARY

This Thesis is divided into four distinct and self-contained parts whose unifying thems is implicit in the general title "Chemical Studies of Some Natural Products".

Part I discusses possible biogenetic routes to the vesicant principle cantharidin in which it is considered that this compound is terpenoid in orighn. It indicates the expected radio-active labelling patterns that would be encountered should feeding corporiments with <sup>11</sup>C labelled nevalenic acid to the bactle <u>Melce proscarabaus</u> I. lead to the incorporation of radic-activity in the cantharidin shown to be synthemised by this insect. Chemical degradations of the theoretically possible labelled cantharidins designed to distinguish between them are discussed and a survey of the occurrence of cantharidin within the family Heloides is given.

Part II describes an attempt to propare. from a consideration of certain theoretical factors which are briefly discussed, a compound with potentially high neuromuscular blocking activity, 3a, 17a-bis (trimethylammonium)-5-a-androstane, by  $S_N^2$  replacement of the methanesulphonyloxy groups of 3 $\beta$ , 17 $\beta$ -dimethanesulphonyloxy-5a-androstane with trimethylamine.

Part III describes a physico-chemical approach to the elucidation of the structure of the sesquiterpene y-lactone, aristolactone occurring in Aristolochia reticulata L and A. serpentaria L. Successful degradation to the parent hydrocarbon germacrane (as a mixture of disstereoisomers) is described these results permitting unequivocal assignment of the basic carbon skeleton of aristolactone. Re-examination of accumulated chemical evidence together with a number of physical measurements (I-R and n.m.r.) are employed to give a probable structure for aristolactone. A cyclisation of methyl oxoaristate, a v-keto ester derived from aristolactone, has been effected giving rise to a new derivative possessing an as yet undetermined bicyclic system.

Part IV describes the elucidation of the structures of the quaternary alkaloid petaline chloride and its methine base, leonticine - compounds isolated from Leontice leontopetalum L. An attempt to synthesise one of the degradation products derived from the methine base leonticine is described but the projected synthesis failed at the last stage.

### TABLE OF CONTENTS

Part I	The Biogenesis of the Vesicant	
	Principle Cantharidin.	4
	Introduction	5
	Discussion	22
	Experimental	37
	References	39
Part II	Attempted Synthesis of	
	3a,17a-Bis(trimethylessonium)-5a-	
	Androstane Dimethanesulphonate by	
	Direct Reaction of 38,178-Dimethane-	
	sulphonyloxy-5a-Androstane with	
	Trimethylamine	48
	Reportmental	61
	References	63

PAGE



Part III	The Structure of Aristolactone, A		
	10 membered Carbon Ring Sesquiterpene		
	Lactone from Aristolochia reticulata L.		
	and Aristolochia sorpentaria L.	68	
	Introduction	69	
	Discussion	101	
	Experimental	155	
	References	165	
Part IV	The Structure of Petaline Chloride,		
	an alkaloid isolated from		
	Leontice Icontopetalum L.	175	
	Introduction	176	
	Discussion	190	
	Experimental	205	
	References	219	

	E A LED
Appendix I	224
Appendix II	225

-3-

# The Biogenesis of the Vesicant Principle Cantharidin

#### INTRODUCTION

- Sen

In 1956 the important observation was made by Tovormina. Gibbs and Hoffl that mevalonic acid lactone (I) labelled in the 2 - position with <sup>1)</sup>C ( in all formulae an asterisk denotes a 14 C isotope) gave rise to 14 C labelled cholesterol (II) in rat liver homogenates. Shortly afterwards Folkers' group<sup>2</sup> reported that mevalonic acid lactone could replace acetate as the growth factor for Lactobacillus acidophilus, and within a very short time work by several schools, namely those of Block, Rabinowitz, Cornforth and Popjak and Lynen<sup>6</sup>, conclusively demonstrated that mevalonic acid is a key precursor of the triterpene squalene (II) from which all naturally occurping steroids are derived.



HO



- C13

#### LLT

Accordingly a great deal of attention was focussed on terpenoid biogenesis<sup>7-10</sup> and, as a result of a number of important tracer experiments, a rational basis was provided for the isoprene rule<sup>11</sup>. This was then modified and expanded into the biogenetic isoprene rule<sup>12</sup> which states that the carbon skeleton of terpenoids is either formally divisible into "isoprene units" linked in a regular "head-to-tail" manner or is derived in a simple fashion from such a skeleton or skeletons through processes of condensation, bond migration, or loss of carbon atoms.



Figure 1.

Satisfactory rationalizations of the elaboration in nature of the major groups of terpenoids have now been put forward 13,14 Thus, condensation of 3 acetate units through a mechanism involving malonate yields phosphorylated mevalonic acid which in turn gives rise to the two key intermediates isopropenylpyrophosphate (IV) and the isomericad-dimethylallylpyrophosphateVahich may be identified as the 5-carbon fragments responsible for the validity of the isoprene rule. Condensation of these compounds<sup>15</sup>, as shown in figure 1, then affords geranylpyrophosphate 6,16,17/VI) which is thus the monoterpone prototype in nature. Geranylpyrophosphate can then either undergo further condensation with isoprenoid pyrophosphates to form higher terponoids, or it can undergo a variety of blochemical transformations to give the various classes of monoterpenoids.

Although the evidence concerning the biogenesis of monoterpenoids belonging to the long established groups is unequivocal, there exist in nature certain compounds for which conclusive proof of anisoprenoid biogenesis has still not been adduced. One such group

-7-

is the cyclopentanoid nonoterpenes". This includes the plant glycosides verbenalin(VII)<sup>10</sup>, aucubin(VIII)<sup>19</sup>, and asperuloside(IX)<sup>20</sup>; the sapogenin genipin(X)<sup>21</sup>; nepetalactone(XI) and nepalic acid(XII) isolated from catnip<sup>22</sup>; and the structurally related lactones iridomyrmecin(XIII)<sup>23-25</sup> isoiridomyrmecin (XIV)<sup>24,25</sup>, and iridodial XV<sup>24-26</sup> which occur in ants belonging to the genus <u>Iridomyrmex</u>.

HO

-8-





TIV











ズ

XI

XII



-9-

XIII XIY XV

As has been pointed out by Wiesner and his co-workers<sup>27</sup>, the fact that these compounds obey the isoprene rule could possibly be fortuitous, and they could well arise in nature in a manner other than via the isoprene route. Wiesner's group suggest as a possible pathway a Woodward fission<sup>28</sup> of an alkylated phenylacetaldehyde as indicated in scheme A (formulae XVI to XVIII),

SCHEME A.









XXIV

Figure 2

However other possibilities also exist as suggested by schemes which were originally put forward in connection with the biogenetic origin of the non-tryptamine portion of the complex indole alkaloids<sup>29</sup>. Although these schemes are now known not to be involved in indole alkaloid biogenesis<sup>30</sup> it is still possible that similar mechanisms could be involved in the formation of the "cyclopentanoid monoterpenes". For example, Wenkert<sup>31</sup> has suggested that the glycosides VII toIX (and by implication the lactones and derivatives X. to XV) could be formed from the unit XXIV which in turn could arise from the rearrangement, hydration and retro-aldolization of prephenic acid 31,32, followed by condensation with a formaldehyde unit, as outlined in figure 2. Wenkert<sup>31</sup> however emphasises that although there is great structural similarity between the prephenic acid derivatives and the "cyclopentanoid monoterpenes", this similarity may be only fortuitous. Radio-tracer studies would be expected to throw more light on the biosynthetic pathway of these interesting compounds, but up to the present no such studies

e-100

would appear to have been reported.

A further compound which may be monoterpenoid in nature is cantharidin (CXVII) the vesicant principle occurring in certain beetles belonging to the family Meloidae. Inspection of the formula of this compound would certainly show it to be consistent with a tail-to-tail condensation of isoprene units. However, a mechanism which would explain this unusual carbon skeleton in terms of normal head-to-tail coupling of isoprene units followed by a methyl group migration has been very recently put forward by Martin-Smith and Khatoon<sup>33</sup>. These authors suggest that cantharidin could conceivably arise from a suitably substituted A-cyclocitral derivative by some such process as is shown in sequence XXV to XXVII.1, 2-Methyl shifts of the type postulated have many analogies in nature8,34,32 It is to be noted that formula XXVII shows the relative stereochemistry of cantharidin with the anhydride ring cis to the oxide bridge - not an absolute storeochemistry - since cantharidin is a non-optically active molecule, having a plane of symmetry.

-11-



XXV

ZY.V.I.

XXVII

It is to be noted that very recently Dean<sup>36</sup> has suggested that the oxide bridge in cantharidin could arise in nature <u>via</u> peroxide formation (as in ascaridole) followed by reduction to the tetrahydrofuran ring system.

That cantharidin might indeed arise by the tail-to-tail condensation of two isoprene units can perhaps be considered from analogy with the tail-to-tail condensation, either between one molecule of farmesylpyrophosphate (XXVIII) and one molecule of nerolidylpyrophosphate (XXIX) or between two molecules of farmesylpyrophosphate

(XXVIII) to form squalene (III; XXX,  $R_1=R_2=$ CH3 CH3-C = CH-CH2-CH2-CH-CH2-) 6,9,10,37-41







XXIX

XXVIII



XXX

The Cornforth and Popjak school<sup>41</sup> has postulated that the tail-to-tail condensation of the two fifteen carbon units takes place <u>via</u> a nucleophilic process, analogous to that involved in the biosynthesis of geranyl- and farnesyl- pyrophosphates<sup>37</sup>. Thus nerolidyland farnesyl- pyrophosphates are considered to condense to form the cyclic phosphate ester XXXIII, as shown in figure 3. Following an elimination reaction the resulting intermediate XXXIV is postulated to suffer reduction by reduced nicotinamide-adenine dinucleotide phosphate (H-NADP) to form squalene and it would seem quite



Squalone biogenesis:  $R_1 = R_2 = CH_2 + C = CH_2 - CH_2$ 

Figure 4

possible that an acyclic procursor of cantharidin could be formed in much the same way.

The scheme outlined in figure 3 fully covers the known experimental observations with respect to the biogenesis of squalene although, as has been pointed out<sup>41</sup>, there is some doubt as to the formation of nerolidylpyrophosphate in the liver<sup>42,43</sup>. Nevertheless the occurrence of optically active nerolidol in plants supports the possibility of enzymatic isomerization of a farnesol derivative to this tertiary alcohol.

The scheme shown in figure 3 assumes the elimination of a proton (in XXXIII) attached to what was originally C-1 of the farmesylpyrophosphate half of the molecule (i.e. a carbon atom originally derived from C-5 of mevalonic acid) but this does not necessarily have to be the case. The proton loss could occur from what was originally C-2 of the farmesyl mojety (XXXIII,figure 4)<sup>41</sup>. After condensation of the nerolidylpyrophosphate (XXXI) with farmesylpyrophosphate (XXXII) to yield XXXIII, the molecule could undergo cyclopropane ring formation (XXXV) with the proton elimination

million



Squalene bicgenesis: R<sub>1</sub> = GR<sub>3</sub>~C-CH-CH<sub>2</sub>-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-C-CH<sub>2</sub>-CH<sub>2</sub>-C-CH<sub>2</sub>-CH<sub>2</sub>-C-CH<sub>2</sub>-CH<sub>2</sub>-C-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>

occurring from the carbon atom  $\beta$  to the pyrophosphate bearing carbon, rather than from the edjacent carbon as proposed in figure 3. Reductive cleavage of this cyclopropanoid intermediate by H-NADP with the concerted elimination of the pyrophosphate anion would give rise to squalene (XXX), as shown in figure 4.

An alternative mechanism for the tail-to-tail coupling of sesquiterpene units in the biogenesis of squalene which does not involve nerolidylpyrophosphate was suggested to Cornforth and Popjak41 by Professor R.B. Woodward. This mechanism (Figure 5) involves the coupling of farmesylpyrophosphate (XXII) with the yilde of thiamine pyrophosphate giving XXVI, which would be followed by the elimination of a proton to yield the complex YXXVII. The condensation of this intermediate with another molecule of farnesylpyrophosphate(XXII) would result in the formation of the intermediate XXXVIII. This mechanism is analogous to the acyloin type of condensation involved in the elaboration of Reduction of XXXVII by the reduced acetoin

-15-

form of nicotinamide - adenine dinucleotide phosphate (H-NADP) would Yield squalene (XXX) and at the same time regenerate the yaine of thiamine pyrophosphate.

Popjak, Low and Moore<sup>45</sup> proposed that the condensation of the two CL5 units was a stereospecific process, and partial proof of their hypothesis was soon adduced<sup>46</sup> from the observation that during "hydrogen transfer" only the hydrogen atom from the "5" side of the reduced nicotinamide coenzyme was involved<sup>47</sup> (H<sub>b</sub> in partial structure XXXIX). Complete proof was provided by the elegant correlation of the stereochemistry at C-ll and C-l2 in trideuterated squalene (partial structure XL) with that of S-trideuterated succinic acid (XLI)



-16-

It is to be anticipated that direct experimental evidence will eventually indicate which of the mechanisms shown in figures 3,4, and 5 is actually involved in squalene biogenesis.

Should a tail-to-tail condensation parallel to one of those shown in figures 3,4, or 5, be occurring in the biogenesis of cantharidin, condensation between one molecule of 3 - methyl butenyl-3-pyrophosphate (XLII) and one molecule of  $\mathcal{O}$ ,  $\mathcal{O}_{1}$ - dimethylallylpyrophosphate or between two molecules of  $\mathcal{O}_{1}$ -dimethylallylpyrophosphate (V) would be expected to give rise to the possible intermediates XLIII or XLIV, where "A" is an attacking nucleophilic species.



The alternative condensation of the butenylpyrophosphate XLII with isopropenylpyrophosphate (IV)

-17-

would give the corresponding double bond isomer of XLIII or XLIV. Ring closure of XLIII or XLIV could then afford cyclohexanoid monoterpenes with a tail-to-tail isoprene coupling.

-18-

Radio-active tracer studies employing 2-14C mevalonic acid would be expected to distinguish between the head-to-tail (with rearrangement) and the tail-to-tail mechanisms just outlined as possible biosynthetic routes for cantharidin. Accordingly attention was focussedon approaches to this end.

It was first necessary to secure a suitable organism with which to undertake the proposed research. A survey of the literature revealed that cantharidin is of widespread occurrence in the family Meloidae, especially within the genera <u>Cyaneolytta, Epicauta, Lytta, Meloe.</u> and <u>Mylabris</u>, the more well authenticated sources of cantharidin being shown in Table 1. For reasons to be outlined in the discussion, Meloe proscarabeus was selected as the most likely potential candidate for the proposed research.

### 102.9in

## TABLEI

### SOURCES OF CANTHARIDIN

Organism	Percentage Canthari	din	Reference
Cissites cephalotes 01.	~		49
( <u>C. maxillosa</u> ) Cyaneolytta gigas F.	-		50
(Lytta gigas) C. signifrons Fahr.	1.89		51
(Lytta coelestina)			50
(Lytta violacea)			
(Aylabris lunata)	1.0		21
Fletica wahlbergia Fahr. Epicauta adspersa Klug.	0.32 0.38 - 0.41		51 52
E. fenoralis Fr.	1.7 - 3.5		53
Korhani Mars.	0.40 - 1.75		54
(Cantharis hirticornis)	2.02		22
(Lytta strata)			50
<u>E. velata Gerst.</u>	2.73		56 51
F. vittata F.	0.4 - 1.33		50, 57, 58
Cantharis vittata)			
Lydus trimaculatus Fischer	-		49 50
Lytta conspicua Waterh. (Mylabris conspicua)	-		50
L. sanguinea Haag (Hugchys sanguinea)	-		59
L. vesicatoria L. (Cantharis vesicatoria)	wings & alytra head & antennac legs	).082 ).088 ).091	58,60
Macrobasis albida Saw	thorax & abdomen	). 24	61
N. cinerea F. (Lytta cinerea)			50

# +20-

### TABLEI

## SOURCES OF CANTHARIDIN

Organism	Percentage Cantharidin	Reference
Noloe angusticollis Say		50
M. maialis L.	-	50,62
<u>A. proscarabeus</u> L.	-	50,62
<u>A. variegatus</u> Donov.	-	50
( <u>mylabris variezata</u> )		
A. violaceus Jarsh	-	50
Mylabris balteata Pall.	0.193	60
( <u>A. punctun</u> )		
<u>A bifasciata</u> De Geer	1.02	50,63
.4. calida Pall.		50
( <u>.4. maculata</u> )		
<u>A. cichorii</u> L.	0.40 - 1.5	51, 55, 58, 64, 66
<u>4. collizata</u> Redt.		50
<u>M. Crocata</u> Pall.	-	50
( <u>M. duodecimpunctata</u> )		-
<u>A dicincta</u> Bertol.	-	59
M. holosericea (lug.	1.3	51
M. macilenta dars.	-	67
M. oculata Thumb	0.615	51
1. phalerata Pall.	1.0 - 1.2	59
( <u>1. sidae</u> )		~ ~
1. pustulata Thumb	0.33- 2.9	64.68
4. <u>nuadripunctata</u> L.	9.2 (dry)	69.70
(i. melanura)		
4. quatuordecimpunctata Pall.	0.49	71
1. tripartita Gerst.	-	59
M. variabilis Pall.	19.28 (dry)	70
The earlier interest in the occurrence of cantharidin can be traced to its former medicinal application as a topical vesicant and counter-irritant but due to its pronounced renal toxicity72,73,74,75, it is no longer employed in this way. The exact function of cantharidin in the insect has not been established although its insecticidal properties may point to a defensive function 50, 75, 77 Volker 78 has pointed out that in the case of Lytta vesicatoria appreciable concentrations of cantharidin are present only in sexually mature individuals and so this fact may indicate that it has a sexual role. However. cantharidin is known to be distributed throughout the wings, head, legs, and abdomen of this species 60 and so it is not confined to the sex organs. At present cantharidin is neither of chemical nor pharmaceutical interest. Nevertheless recently a renewed interest has been shown in the biochemical mechanism of cantharidin acantholysis<sup>79</sup> and also some synthetic analogues of the type XLV have been claimed to exhibit anti-hypertensive properties 80.

N-CHo-CH Re

-21-

# Discussion

As outlined above it first became necessary to obtain a species of beetle suitable for radiotracer experiments. The beetle Meloe proscarabeus L. was chosen since it was large enough for intraperitoneal injection of radio-active material, and was of local occurrence. The early literature makes reference to the isolation of cantharidin from this beetle and from the reddish droplets it secretes when handled 50,62 In view of the inaccessibility of some of the original work, and in view of the fact that no recent investigations have been reported on this beetle, together with the ambiguity surrounding certain early identifications of cantharidin (especially the confusion of this compound with pederin<sup>31</sup>), it was deemed necessary to first confirm the occurrence of cantharidin in Meloe proscarabeus.

. Adult specimens of M. proscarabeus collected at Loch Ardinning, Starlingshire, in May 1962, were killed in the laboratory by means of chloroform. The red droplets which were secreted just prior to death were collected and examined separately. The droplets were taken up in hot ethenol and after removal of the solvent the oily residue was sublimed by heating to 90°/0.05 mm. thus affording a colourless crystalline sublimate. The material was shown to be cantharidin by comparison with an authentic sample, there being no melting point depression (sealed tube) on admixture of the two specimens. The infra-red spectra in K Cl disc were completely superposable and identical with the published spectrum<sup>82</sup> Cantharidin was also obtained from both freshly killed and pulverized and from dried and pulverized bodies of the beetles. It was found to be present to the extent of 0.137% of the total body weight and 0.364% of the dried body weight.

Radio-active tracer studies on <u>Meloe proscarabeus</u> employing 2-14C mevalonic acid would be expected to

-23-

pin-point the cantharidin biosynthetic pathway. Incorporation of the labelling would indicate that cantharidin is indeed terpenoid in origin, rather than being formed, for instance, from shiki.mic acid <u>via</u> the theoretical <u>seco</u>-prephenateformaldehyde intermediate XXIV postulated by Wenkert<sup>31</sup>. Degradation of the labelled cantharidin molecule would then be expected to give a distinction between a head-to-tail or a tail-to-tail coupling of isoprene units. Although the symmetry of the cantharidin molecule (XXVII) does not make it strictly an ideal compound with which to conduct tracer biogenetic experiments, the labelling patterns which would arise from the alternative coupling pathways would be different, as seen below.

-24-

# HEAD-TO-TAIL CONDENSATION AS A POSSIBLE ACCHANISM FOR CANTHARIDIN BIDSYNTHESIS

If the biogenesis of Cantharidin were to involve a normal head-to-tail condensation of isoprenoid units as in the biogenesis of geraniol<sup>6,15,16,17</sup>, injection of <u>Meloe proscarabeus</u> with  $2-{}^{14}$ C mevalonic acid should after its 2- stage phosphorylation to XLVI<sup>83</sup> give rise to  $1-{}^{14}$ C-isopropenylpyrophosphate(XLVII) and

· · · · · ·



4-14C-3-methylbut-2-enyl-1-pyrophosphate (C(- methyl-A -14C-methyl-allylpyrophosphate) (XLVIII) which would be expected to condense to give the isotopically labelled geranylpyrophosphate (XLIX) by the accested mechanism<sup>10</sup>, 37, 42 as shown in figure 6. This compound(XLIX) or subsequent derivatives might then be expected to undergo specific hydroxylations at allylic positions, the particular sites of attack presumably being uniquely determined by specific enzymes.

Cyclization of geranylpyrophosphate itself, geraniol or some allylically hydroxylated derivative to the  $\beta$ -cyclocitral skeleton can be envisaged as occurring by attack of an electrophilic species on the isopropylidene double bond of XLIX, with concerted attack from the TT electrons of the second double bond, effecting ring closure. The resulting compound would belong to one of four possible types L to LIII, where "X" represents the attacking electrophilic species, "Y" an oxygen function resulting from an earlier allylic hydroxylation, and "A" a nucleophilic species (e.g. hydroxyl ion) completing the cyclization sequence.

-25-



In the cyclizations of geranyl- and farnesylpyrophosphates in plants to yield cyclic mono- and sesquiterpenes respectively, the initiating electrophilic species """ would appear to be invariably a protonll, 13, 14, 84. On the other hand in the cyclizations of squalene giving rise to tetracyclic, pentacyclic, or onocerin types of triterpenes, the electrophilic species "X" would appear invariably to be the equivalent of OH , in both plants and animals<sup>85</sup>, this species being known to involve molecular oxygen36. The cyclizations of geranylgeranylpyrophosphate leading to the various types of diterpenoids appear to involve both mechanisms. A proton would seen to be involved in most cases but certain diterpendids appear to utilize OHDas evidenced by the occurrence in nature of a number of 3- oxygenated diterpenoids. Examples are gibberellic acid (LIV)<sup>87</sup>, the grayanotoxins<sup>38-90</sup> (e.g. andromedotoxin, LV) calestol (LV191, and certain diterpenoid alkaloids such as aconitine LVII92 and the Erythrophleum alkaloids93-95 an example of the last group being cassaine (LVIII).

102600

It is also to be noted that Cross<sup>96</sup> has suggested that the sesquiterpenoid picrotoxinin (LXI) could arise by attack on LIX by a hydroxonium ion, followed by hydroxide ion attack on the product LX causing a concerted double 1,2-methyl shift with elimination.



Should beetles belonging to the family Meloidae possess an enzyme system utilizing ON as the electrophilic cyclizing agent in cantharidin biogenesis, a compound such as LXII might well be formed which on conversion into LXIII could readily give rise to the biogenetic sequence XXV to XXVII proposed by Martin-Smith and Khatoon<sup>33</sup>. At the same time intermediate LXIII could be derived from compounds such as LXIV where the cyclization has been initiated by a proton and followed by dehydrations, allylic oxidations, reductions, etc. From the number of

-27-

steps required it night appear that such a route was less likely than the one involving initial stack by  $OH \oplus$  but the possibility cannot be disregarded.



WTIT VIXIV LATT It can be seen that compounds L and LI or LII and LIII differ from each other only with respect to whether it is the unlabelled or the labelled carbon atom which suffers oxidative allylic attack in the proposals outlined. Since it may be reasonably assumed that the isomerization of XLVII to XLVIII (Figure 6) is stereospecifically controlled by an enzyme, the labelled methyl group in XLVIII would be expected to be solelycis or solelytrans to the acthylene groups bearing the pyrophosphate function, and randomization of the isotope label would not Similarly any allylic hydroxylation be expected. (in XLIX for example) under enzymatic control would be expected to give rise to soldy an L or LII type, as distinction would be made between the methyl

-280

group cis to the long chain at the other end of the double bond and the methyl group trans to it. Thus a mixture of the L and LII type would not be expected and one or the other would be expected as the sole product. However there is no a priori method of predicting which would be formed. The labelling pattern in cantharidin itself would then be LXV or LXVI.



It should be noted, however, that Arigoni<sup>13</sup> has stressed the point that the specificity of enzyme systems appears to decrease rapidly with the number of carbon atoms present in the substrate molecule and so the validity of the above assumptions must remain in some doubt until further biochemical information becomes available.

-29-



# TAIL-TO-TAIL CONDENSATION AS A POSSIBLE MECHANISM FOR

#### CANTHARIDIN BIOSYNTHESIS

Should cantharidin arise in nature through a tail-to-tail coupling mechanism such as those previously indicated in figures 3,4, and 5, any cyclization step must involve an anti-Markownikoff attack. Such an anti-Markownikoff addition would seem highly unlikely. However it is to be noted that the formation of the tetra- and penta- cyclic triterpenes from squalene does involve an anti-Markownikoff addition in the formation of ring C, although here the concerted nature of the total electron movements obviously provides sufficient driving force to over-ride the one unfavourable ring closure. It may also be noted that an anti-Markownikoff cyclization has been postulated in the proposed mechanism of picrotoxinin biogenesis (LIX to LXI)<sup>96</sup>.

Several intermediates are possible from the condensation of LXVII with LXVIII, or of two molecules of LVII by the Woodward mechanism, one of which is shown in LXIX (figure 7). However, all would be expected to lead, by anti-Markownikoff cyclization, to one of the four basic types shown in LXX to LXXIII. where "Y" is an oxygen function resulting from an earlier allylic oxidation. Once again no prior prediction of the actual position adopted by the <sup>14</sup>C label is possible and the resulting cantharidin would be one of LXXIV to LXXVII.



## Proposed Research

Unfortunately the extreme winter of 1962-1963 caused the non-appearance of <u>Meloe proscarabeus</u> in May 1963 and despite many field trips no specimens could be found upon which to carry out labelling experiments. In outline, the approach was to have been as follows.

Live adult specimens of <u>Meloe proscarabeus</u> were to have been made to deplete their body stores of cantharidin through induction of its secretion by irritation of the beetles. Aqueous 2-14C mevalonic acid lactone was then to have been injected intraperitoneally and after a suitable interval the beetles milked for newly synthesised cantharidin, the process being repeated successively. The collection of the cantharidin - containing droplets and their subsequent concentration and sublimation, was to have been performed in an identical manner to that used in the identification of cantharidin already achieved on the specimens collected in May 1962. Later the beetles would have been killed and their bodies worked up for further quantities of cantharidin. It would then have been established whether or not the cantharidin had incorporated 14C from the mevalonic acid lactone. If so, the level of radio-activity would have been ascertained so that suitable dilution with unlabelled cantharidin could be carried out. Then degradative experiments based on the known chemistry of cantharidin would have been performed on the diluted material.

Examination of formulae LXV and LXVI shows that a head-to-tail coupling of isoprene units in cantharidin would result in half the radio-activity being located in the cyclohexane ring and half in the

-32-

ring substituents. Tail-to-tail coupling on the other hand would result in all the activity being located in the ring substituents (formulae LXXIV to LXXVII). Thus the degradative scheme must include a procedure for distinguishing between the cyclohexane ring and the substituent carbon atoms.

Kuhn-Roth oxidation of cantharidin would <u>a oriori</u> be expected to yield two moles of acetic acid and two moles of carbon dioxide with removal of all ring substituents. Unfortunately, however, this method has been found in general to give unsatisfactory results with highly substituted C-methyl compounds<sup>97,98</sup>. For example the substituted methyl succinic anhydride LXXVIII a degradation product of vitamin B<sub>12</sub> gives values consistent with only one C-methyl group<sup>99</sup>.



During work on the structural elucidation of cantharidin, Piccard<sup>100</sup> and later Gadamer<sup>101</sup> heated

-33-



gure 8.







cantharidin with phosphorus pentasulphide and obtained as product <u>e-xylene (LXXIX)</u> (numbering corresponds to the cantharidin molecule).



Should this degradative schele be followed, however, the symmetry of the cantharidin molecule coupled with the symmetry of the g-xylone molecule would necessitate additional experiments. This is apparent when it is noted (Figure 8) that the six possible labelled cantharidins give rise to five labelled and one isotope-free s-xylenes (see LAXR ~ a to f). Production of the isotope-free form would conclusively point to pattern LMXVI in cantharidin and thus a tail-to-tail coupling occurring in Meloe proscarabeus. Production of labelled o-xylene would not per se be unambiguous. Thus double labelling in g-xylene (LXXX-a or-d) would point to either LXV or LXXV leaving the biosynthetic pathway undefined. Further the presence of single labelling in the g-xylene (as in LXXX b.c. or f) would not distinguish between LXVI, L'XIV, or L'XVII.

- 34-

Similarly the modified degradation procedure devised by Piccard<sup>1,))</sup> and Gadamer<sup>101</sup> of dry distilling cantharidin with soda-lime leading to the formation of centharene (dihydro-2-xylene; LXXXI) would also provide an ambiguous result.



A considerably nore useful degradative pathway, especially if employed in conjunction with 6-xylene formation, would seem to be catalytic fission (reversed Diels-Alder reaction) of cantharidin. Heating cantharidin with 20% palladium on charcoal is reported<sup>102</sup> to result in the formation of furan and dimethyl aleic anhydride, which in the case of labelled cantharidin would give rise to one of LEERIT a to f and LXXXIII a to f, respectively (Maure 8). Presence of labelling in the furan (LXYXII a LXXII b) would conclusively indicate that the head-to-tail coupling route was involved in cantharidin biosynthesis. This information coupled with the number of isotopic carbon labels found in the o-xylene, would pin-point the labelling pattern, two isotopes indicating LXV, and one, LXVI. Similarly the absence of 14C in the

-35-

furan indicates that a tail-to-tail mechanism is occurring in cantharidin biogenesis. This conclusion together with knowledge of the number of isotope labels in <u>o</u>-xylene will show the exact route, no label in the aromatic compound indicating LXXVI, and two labels, LXXV. Should the <u>o</u>-xylene contain one label a distinction in the fine mechanism could not be made, but the tail-to-tail biosynthetic route for cantharidin would be confirmed.

-36-

## Experimental

-37-

Melting points were taken in sealed capillary tubes in a heated block, and are uncorrected. Sublinations were carried out in horizontal tubes employing a Towers heating unit. Infra-red spectra were measured on a Perkin-Elmer Infracord. <u>Cantharidin from the secretion of Meloe proscarabeus L.</u>

The droplets (5 m.g.) secreted by the <u>A</u>. proscarabeus prior to death were dissolved in boiling ethanol and transferred to a sublimation tube. Removal of the solvent under reduced pressure and sublimation of the oily residue at  $90^{\circ}/0.05$  mm afforded white crystalline needles (1.3 mg.). The needles were washed with petroleum ether (40-60°) and dried <u>in vacuo</u> to give material m.p. 212-214° (sealed tube) identical with that obtained below from the body of the same insect and identical with authentic cantharidin.

Cantharidin from the body of . 1. Proscarabeus.

a. A freshly killed adult speciman (1.07g.) was frozen with dry ice and pulverized in a mortar with a pestle. A small quantity of 10% w/v HCI was added to ensure liberation of all combined cantharidin<sup>103</sup> and the remains were transferred to a sublimation tube with the aid of chloroform. After first heating to  $60^{\circ}/0.05$  mm Hg for 1 hour to remove highly volatile matter, the temperature was raised to  $90^{\circ}$  whereupon cantharidin (2 mg.) sublimed as ailky white needles. After washing with petroleum ether ( $40-60^{\circ}$ ) these had m.p. 214-216° (sealed tube), undepressed on admixture with authentic cantharidin. The infra-red spectrum in KCl disc showed prominent peaks at 1840, 1770, 1235, 1005, 995, 690 and 850 cm<sup>-1</sup>, and was completely superposable with that of authentic cantharidin.

b. A freshly killed specimen of <u>M</u>. <u>proscarabeus</u> weighing 0.626g. immediately after death was dried at 50° for 3 hours and then stored in a vacuum desiccator for 3 days. The dried carcass (0.346g.) was pulverized and treated as above, to afford, on sublimation, cantharidin (1 mg.) again having m.p. 214-216° and the correct infra-red spectrum.

-38

#### R E F E R E N C E S

- 1. Tovormina, Gibbs and Hoff, J. Amer. Chem. Soc., 1956, 72, 4+98, 6210.
- Wolf, Hofmann, Aldrich, Skeggs, Wright and Folkers,
   J. Amer. Chem. Soc., 1957, <u>79</u>, 1486.
- Admur, Rilling and Block, J. Amer. Chem. Soc.,
   1957, 79, 2640.
- 4. Dituri, Gurin and Rabinowitz, J. Amer. Chem. Soc., 1957, 79, 2650.
- 5. Cornforth, Cornforth, Popjak and Gore, <u>Blochem</u>. J., 1958, <u>69</u>, 146.
- Lynen, Eggerer, Henning and Kessel, <u>Angew. Chemie</u>, 1958, <u>70</u>, 738.
- 7. Inter. alla.
  Childs and Bloch, J. Biol. Chem., 1962, 237, 62.
  Popjak, Cornforth, Cernforth, Ryhage and Goodman,
  J. Biol. Chem., 1962, 237, 56;
  Baisted, Capstack and Nes, Biochemistry, 1962, 1, 537;
  Knauss, Porter and Wasson, J. Biol Chem. 1959, 234, 2835;
  Chaykin, Law, Phillips, Tschen and Bloch,
  Proc. Nat. Acad. Sci., U.S.A., 1958, 14, 998;
  Arigoni, Experientia, 1958, 14, 153;
  Birch, English, Lassey, Westropp and Smith,
  Proc. Chem. Soc., 1957, 233.

-39-

8.	Cornforth, Cornforth, Pelter, Horning and Popjak
	Proc. Chem. Soc., 1958, 112.
9.	Lynen, Agranoff, Eggerer, Henning and Moslein,
	Angew. Chemie, 1959, 71, 657.
10.	Cornforth, Cornforth and Popjak, Tetrahedron
	Letters, 1959, No. 19, 29.
11.	Ruzicka, Experientia, 1953, 2, 357.
12.	Ruzicka, Proc. Chem. Soc., 1959, 341.
13.	Arigoni, "Steric Aspects of the Chemistry and
	Biochemistry of Natural Products,"
14.	Hendrickson, Tetrahedron, 1959, 7, 82.
15.	Arganoff, Eggeroff, Henning and Lynen,

J. Amer. Chen. Soc., 1959, 81, 1254.

- 16. Rilling, Tchen and Bloch, Proc. Nat. Acad. Sci., U.S.A., 1958, 44, 167.
- 17. Bloch, Ciba Foundation Symposium, "The Biogenesis of Terpenes and Sterols," J. & A. Churchill, London, 1958, p.4.
- 18. Buchi and Manning, Tetrahedron Letters, 1960, No.26, 5.
- 19. Haegle, Kaplan and Schmid, <u>Tetrahedron</u> Letters. 1961, No. 3, 110.
- 20. Grinshaw, Chem. and Ind., 1961, 403.
- 21. Ujerassi, Nokano, James, Zalkow, Eisenbaum Shoolery, J. Org. Chem., 1961, 26, 1192.

-40-

2

- 22. AcElvian and Risembraun, J. Amer. Chem. Soc., 1955, 77, 1599.
- 23. Pavin, Chem. e Industr, 1955, 37, 714.
- 24. Cavill, Ford and Locksley, Chem. and Ind., 1956, 465.
- 25. Cavill, Ford and Locksley, <u>Aust. J. Chem.</u>, 1956, 2, 288.
- 26. Cavill and Hinterberger, <u>Aust. J. Chem.</u>, 1960, <u>13</u>, 296.
- 27. Valenta, Miesner, Babin, Bogri, Forrest, Fried, and Reinshagen, Experientic, 1962, 18, 111.
- 28. Woodward, <u>Nature</u> (London), 1948, 162, 155.
- 29. Thomas, Tetrahedron Letters, 1961, No. 16, 544.
- 30. Leete, Ghosal and Edwards, J. Amer. Chem. Soc., 1962, 84, 1068.

Edwards and Leete, Chem. and Ind., 1961, 1666

- 31. Wenkert, J. Amer. Chem. Soc., 1962, 84, 98.
- 32. Wenkert and Bringi, J. Amer. Chem. Soc., 1959, 81, 6535, 1474.
- 33. Martin-Smith and Khatoon, Progress in Drug Research, Vol. 6, ed. Jucker, Birkhauser Verlag, Basle, 1963, pp. 279-346.

-41-

- 34. Birch, Rickards, Smith, Harris, and Mhalley, <u>Proc. Chem. Soc.</u>, 1958, 223. Britt and Arigoni, <u>Proc. Chem. Soc.</u>, 1958, 224.
- 35. Jones and Lowe, J. Chem. Soc., 1960, 3959.
- 36. Dean, Naturally Occurring Oxygen Ring Compounds, Butterworths, London, 1963, p.32.
- 37. Rilling and Bloch, J. Biol. Chem., 1959, 234, 1424.
- 38. Cornforth and Popjak, Blochem. J., 1954, 58, 403.
- 39. Ruzicka, Tschemmosen and Hauser, Movementia, 1993, 2, 362.
- 40. Crabbe and Jurisson, Ind. Chim. Belge., 1957, 22, 1309.
- 41. Popjak, Goodman, Cornforth, Cornforth and Ryhage, J. Biol. Chem., 1961, 236, 1934.
- 42. Goodman and Popjak, J. Lipid. Research. 1960,1,286.
- 43. Popjak, Tetrahedron Letters, 1959, No.19, 19.
- 44. Breslow, J. Amer. Chem. Soc., 1958, 80, 3719.
- 45. Popjak, Love and Moore, <u>J. Lipid. Research.</u> 1962, <u>3</u>, 364.
- 46. Popjak, Schroepfer and Cornforth, <u>Biochem</u>. Biophys. Res. Comm., 1962, 9, 371.

- 47. cf Cornforth, Ryback, Popjak, Donniger, Schroepfer, Bioches. Biophys. Res. Cons., 1961-1962, 6, 438.
- 48. Cornforth, Cornforth, Donniger, Ryback and Schroepfer, Biochem. Biophys. Res. Comm., 1963, 11, 129.
- 49. van Zijp, Pharm. Weekblad, 1922, 59, 285.
- 50. Kobert, Lehrbuch der Intoxikationen vol. II Enke, Stuttgart, 1906, pp. 434-447.
- 51. Colledge, Pharm. J., 1910, 84, 674.
- 52. Coll, Rev. Farm. (Buenos Aires), 1931, 73, 17.
- 53. Pfister, <u>Anales Quim. Farm</u>. (Chile) 1940, 26 (through <u>Chem. Abs.</u>, 1941, <u>35</u>, 6061.)
- 54. Shimano, Hizuno and Boto, Ann. Proc. Gifu. Coll. Pharm., 1953, no. 3, 44, (through Chem. Abs., 1956, 20, 13308).
- 55. Hooper, Pharm. J., 1913, 82, 391.
- 56. van Zijp, Pharm. Weekblad., 1917, 54, 295.
- 57. Fahnestock, Amer. J. Pharm., 1857, 6, 86.
- 58. Warner, <u>Viert</u>. fur. prakt. Pharm., 1857, 6, 36.
- 59. Wallis, Textbook of Pharmacognosy, J. and A. Churchill, London, 1955, pp. 336-339.
- 60. Blyth and Blyth, Poisons: Their Effects and Detection, 5th Ed., Charles Griffin, London, 1920, pp. 500-506.

- 61. Viehoever and Capen, J. Amer. Off. Agri. Chem., 1923, 6, 489.
- 62. Schroff, Lehrbuch der Pharmacologie, Braumuller, Vienna, 1856, p. 374.
- 63. United States Dispensatory, 25th Edia Lippincott, New York, 1955, pp. 236-239.
- 64. Chopra, Indigenous Drugs of India, 2nd Ed. Dhur and Sons, Calcutta, 1958, pp. 472-473.
- 65. Fwe, J. Amer. Pharm. Assoc., Sci. Ed., 1920, 2, 257.
- 66. Siering, Suddeut. Apoth. Ztg., 1949, 89, 41.
- 67. Martindale, "The Extra Pharmacoepia," vol. I, 24th Ed., The Pharmaceutical Press, London, 1958, p. 352.
- 68. Iyer and Guha, J. Ind. Instit. Sci., 1931, 14A, 31.
  69. Bluhn, Z. fur Chemie., 1865, 675.
- 70. Cotte, Compt. Rend. Soc., 3101., 1920, 83, 106.
- 71. Bluhm, Pharm. Z. fur. Russl., 1866, 4, 160.
- 72. Sakamoto, Proce Jap. Pharmacol. Soc., 1933, 7, 118, (through Chem. Abs., 1935, 29, 2605.)
- 73. Sarada and Toholau, J. Exper. Med., 1936, 29, 156, (through Chem. Abs., 1936, 30, 8373).
- 74. Pearce, J. Fxper. Med., 1913, 17, 542.

alpha

- 75. Azzi, Arch. Sci. Med., 1917, 40, 125.
- 76. Gowitz, <u>Arb. Physiol. Angew. Ent.</u>, <u>Berl.</u>, 1937, 4, 116, (through <u>Chem. Abs.</u>, 1937, <u>31</u>, 6764).
- 77. Sato, <u>Okayama Igakkai Zasshi</u>, 1941, <u>53</u>, 679. (through <u>Chem. Abs.</u>, 1943, <u>37</u>, 2072).
- 78. Volker, Frohners Lehrbuch. der Toxicologie fur Tierartzte, Enke, Stuttgart, 1950.
- 79. Weakley and Finbinder J. Invest. Dermatol., 1962, 39, 39 and refs. theirein.
- 80. British Patents, 770, 624; 770, 625, (through Chem. Abs., 1957, <u>51</u>, 16557).
- 81. Pavan, <sup>M</sup>Proceedings of the Fourth International Congress of Biochemistry,<sup>4</sup> 1958, vol. XII, Biochemistry of Insects pp. 15-35.
- 82. Stork, van Tamelen, Friedman and Burgstahler, J. Amer. Chem. Soc., 1953, 75, 384.
- 83. Inter Alia.

Loomis and Battaile, <u>Federation Proc.</u>, 1960,<u>19</u>, 240; Waard and Popjak, <u>Biochem</u>. J., 1959, <u>73</u>, 410; Henning, Moslein and Lynen, <u>Arch. Biochem</u>. <u>Biophys</u>. 1959, <u>83</u>, 259; Tchen, J. Biol. Chem. 1958, 233, 1100; Tchen, J. Amer. Chem. Soc., 1957, <u>79</u>, 6344.

- 84. Barton and de Mayo, Quart. Rev., 1957, 11, 189.
- 85. Eschenmosser, Ruzicka, Jeger and Arigoni, Helv. Chim. Acta, 1955, 38, 1850.
- 86. Tchen and Bloch, J. Biol. Chem., 1957, 266, 931, 921.
- 87. McCapra, Scott, Sim and Young, Proc. Chem. Soc., 1962, 185.
- 88. Kakisawa, Yannai, Kozima and Nakanishi, <u>Tetrahedron Letters</u>, 1962, No. 6, 215 Kakisawa, Kurono, Takahashi and Herata, <u>ibid</u> 1961, No. 2, 59.
- 89. Tallent, J. Org., Chem., 1962, 27, 2968.
- 90. Isawa, Kumazawa and Nakajima, <u>Chem and Ind.</u>, 1961, 511. Isawa, Kumazawa and Nakajima, <u>Agr. Biol. Chem</u>. (Japan), 1961, <u>25</u>, 793, 782.
- 91. Cais, Djerassi and Mitscher, J. Amer. Chem. Soc., 1958, 80, 247.
- 92. Wiesner, Gotz, Simmons, Fowler, Bachelor, Brown and Buchi, <u>Tetrahedron Letters</u>, 1959, <u>42</u>, 1127.
- 93. Engel, Helv. Chim. Acts , 1959, 42, 1127.
- 94. Arya, Engel and Ronco, Helv. Chim Acta, 1961,44, 1645.
- 95. Arya and Engel, <u>Helv. Chim. Acta</u>, 1961, <u>44</u>, 1650. Arya, J. Ind. Chem. Soc., 1961, <u>38</u>, 419.

- 96. Cross, Quart. Rev., 1960, 14. 317.
- 97. Kirsten and Stenhagen, Acta. Chem. Scand., 1952, 6, 682.
- 98. Tashinian, Baker, and Koch, Anal. Chem., 1956, 28, 1304.
- 99. Garbers, Schmid and Karrer, Helv. Chim. Acta, 1954, 37, 1336.
- 100. Piccard, Ber., 1886, 19, 1404; 1879, 12, 577; 1878, 11, 2122; 1877, 10, 1504.
- 101. Gadamer, Arch. Pharm., 1917, 255, 315.
- 102. Bruchhausen and Bersch, Arch. Pharme, 1928, 266, 697.
- 103. cf. Viehover and Capen, J. <u>Amer. Off. Agr. Chem</u>., 1923, 6, 489.

#### ATTEMPT TO PREPARE

3a,17a-Bis(trimethylamonium)-5a-Androstane Dimethanesulphonate By Direct Reaction of 38,178-Dimethanesulphonyloxy -5a-Androstane With Trimethylamine

The development and refinement within recent years of such concepts as the receptor theory of drug action1, the metabolite displacement theory2, the concept of bioisosterism<sup>3</sup>, the supporting molety theory<sup>4</sup>, and the concept of drug latentiation<sup>5</sup>, coupled with an increased understanding of the ultimate biochemical, physiological, and pharmacological mechanisms involved in drug action, holds out promise of a possible departure from the empiricism traditionally inherent in the synthesis of new drugs required for specific clinical purposes, with an accompanying increase in the rationality of approach to this problem. Indeed the planned synthesis of antimetabolites has been termed "the revolution in pharmacology, while conclusions drawn from the receptor theory recently led to the introduction of a new class of anabolic steroids . The presently described attempt to prepare 3% 170 -bis (trimethylammonium) - 50 - androstane (I) represents an application of certain theoretical deductions in an attempt to achieve a rational approach to the synthesis of new neuromuscular blocking agents as outlined below.

-49-



Although neuromuscular blocking properties are exhibited by a wide variety of quaternary salts as first shown by the classical work of Grum Brown and Fraser in 1869<sup>8</sup>, potent activity would appear to be characteristic of certain compounds in which there are two or more quaternary centres. This fact has given rise to proposals of a "two-point attachment" theory<sup>9,10</sup>, in which it is postulated that bisquaternary compounds interact simultaneously with two anionic receptor sites - sites which are normally concerned in the translation of nerve impulses into muscular contraction through the action of acetylcholine on the outer surface of the muscle end plate of voluntary muscle<sup>11</sup>. A somewhat modified version of

-50-

the two-point attachment idea has been advanced by Waser<sup>12</sup> who regards the receptor as a roughly circular pore with anionic centres around the inner rim. Bisquaternary salts are then assumed to straddle this pore and achieve two point attachment at opposite ends of the chord which the molecule thus makes with the cross section of the circular pore. At the same time, however, not all workers accept a two point attachment and it has been proposed<sup>13</sup> that only one cationic head of a bisquaternary salt is actually involved in interaction with the acetylcholine receptors, the second cationic centre serving to repell incoming acetylcholine molecules. These ideas have become known as the "adumbration theory".

Should the "two-point" attachment theory be correct, it is of great importance that the interanionic distance of the receptor system be accurately determined since such information would prove invaluable in the design of new molecules capable of acting as potent neuromuscular blocking agents. Indeed considerable attention has been devoted to this problem, all deductions being necessarily made from considerations of the molecules of compounds known to be potent

-51-

neuromuscular blocking agents, since direct study of the receptors themselves has not proven to be feasible although claims have been made for the isolation of a protein material showing many of the characteristics to be expected of the acetylcholine receptor14. Unfortunately, however, the great bulk of such studies has been with conformationally non-rigid molecules. such as the polymethylene bisammonium compounds, d-tubocurarine, and their closely related analogues. It was assumed in these studies that the flexible pharmacon was adsorbed at the receptor in the conformation showing minimal non-bonded interactions within the molecule and maximal charge separation that is in the case of the polymethylene bisammonium salts, the fully staggered conformation. The fact that decamethonium (II) was the most potent of the polymethylene series was then taken as an indication that the anionic sites in the receptor were spaced at a distance of ca 14A apart as this was the interonium distance in the fully. staggered conformation of decamethonium. Moreover Paton and Zaimis<sup>10</sup> claimed that this distance of 14Å would also accomodate the flexible d-tubocurarine

-52-
molecule (III) despite the fact that examination of molecular models reveals that this molecule can have interonium distances varying from 6Å to only 12Å.



The different mechanisms of action of the two compounds - depolarizing block in the case of decamethonium and non-depolarizing (formerly termed competitive) block in the case of d-tubocurarine werenot considered significant in terms of the inter-anionic site distance.

There is no a <u>priori</u> reason for assuming that bisquaternary compounds are adsorbed on the receptor in their thermodynamically most favoured conformation. Indeed the influence of entropy would be expected to ensure that within a given population of bisquaternary molecules a number would exist in other conformations and these could easily be the active species. Again although little is known concerning the exact nature of the forces involved in drug-receptor interaction it would seem safe to assume (by analogy with known physical and chemical processes) that energy must be supplied to the system drug and receptor in order to form the drugreceptor complex. One way in which the supplied energy could be taken up would be for the drug molecule to adopt a conformation of higher energy than the fully extended conformation.

That the originally proposed distance of 14% between adjacent anionic sites in the receptor system cannot be correct is made quite apparent by the potent neuromuscular blocking properties of the fully rigid C-curarine I and toxiferine I<sup>15</sup>, in which the interonium distance is fixed at 9.7Å <sup>16</sup> and of compounds such as cyclooctadecane - 1,10-bis (trimethylamnonium) iodide<sup>17</sup> and the polymethylene bis (tropinium) halides<sup>16</sup> in which the maximal interonium distance is only just over 9Å as seen by inspection of models. Moreover, recent deductions as to the interonium distances of the polymethylene bisquaternary salts from studies of their conductance in water has led to the conclusion that the internitrogen distance in decamethonium (II) is ca 9.5Å <sup>18</sup>.

-54-

These results are of considerable interest in terms of the identity distance 19,20, or the distance between peptide bonds in a maximally extended peptide, which has the value of 3.61A<sup>19</sup>, and in terms of the distance between two turns of an C-protein helix which has a value of 5.5A<sup>21</sup>. Making allowance for slight adaptation of the shape of the cellular proteins during drug receptor interaction (compare the induced fit theory of Koshland<sup>22</sup>), the interonium distances of the very active neuromuscular blocking agents can be seen to correspond closely to multiples of these distances. Thus considering the identity distance, the interonium distance of hexamethonium at 6. 3A18 is somewhat less than 2 x 3.61, that of decamethonium at 9.5A 18 is somewhat less than 3 x 3.61, whilst that of octadecamethonium, at which a second maximal neuromuscular blocking potency is reached 14 should be ca 14A18 which is somewhat less than 4 x 3.61Å.

It would appear that an exact fit at the anionic sites by the molecule of the neuromuscular blocking agent is necessary for depolarizing activity, for when steric hinderance to the approach of the cationic heads

-55-

to the receptor is increased by increasing the bulk of the substituents on the nitrogen atom, change to non-depolarizing activity is observed. In view of the non-depolarizing activity of hexamethonium it may therefore be that the molecules of this compound are less capable of an exact fit at the receptor system than are those of decamethonium. It is also of considerable interest that replacement of the trimethylammonium functions in the polymethylene bis (trialkylammonium)-series by triethylammonium groups leads to an increase in interonium distance as shown by conductance experiments<sup>23</sup>.

In view of the fact that the conductance measurements show the interonium distance in decamethonium to be <u>ca</u>  $9.5^{\circ}$   $1^{\circ}$  whilst the rigid toxiferine I has an interonium distance of  $9.7^{\circ}$ , it seemed of considerable importance to prepare other rigid bisquaternary salts having an interonium distance of this order and test whether or not such compounds would be potent neuromuscular blocking agents. Construction of molecular models shows that  $3^{\circ}$   $1^{\circ}$   $0^{\circ}$  bis (trimethylammonium) - $5^{\circ}$  and rostane (I) has an interonium distance of  $9.5^{\circ}$ 

-56-

so this compound seemed an obvious compound to synthesise and test biologically. Moreover that it could be expected to possess a favourable combination of lipophilic to hydrophilic properties, upon which great stress has been laid as a factor in the determination of neuromuscular blocking activity<sup>24</sup>, would be suggested by the potent activity present in the steroidal alkaloid malouetine (3 $\beta$ , 20 G bis (trimethylammonium) - $5\alpha$  - pregnane) (IV)<sup>25</sup> which has an interonium distance of ca 11.5 ± 1.5Å, the range being due to free rotation of the bond between C-17 and C-20.



The obvious route to the desired  $3\alpha', 17\alpha'$ -bis (trimethylammonium)-50% androstane appeared to be by way of nucleophilic displacements of the methane sulphonyloxy groups of  $3\beta', 17\beta'$ - dimethanesulphonyloxy 50%- androstane by suitable nitrogenous nucleophilic

-57-

species as this would be expected to give inversion of configurations at both C-3 and C-17 if the reaction were SN<sup>2</sup> at both centres. Nucleophilic displacement with inversion of the methanesulphonyloxy group in the  $3\beta$ position would not be expected to involve any complications in view of earlier work<sup>26</sup> but S<sub>N</sub><sup>2</sup> attack at C-17 could be expected to involve some difficulties as this position is a neopentyl position. However, recently, through the use of N-methyl-pyrrolidone 3 tertiary-butyl alcohol (19:1) as a solvent, a number of successful  $S_N^2$ replacements of 17 ß- substituents have been reported in the steroid field<sup>27</sup>. Accordingly it was decided to employ a modification of this procedure in an endeavour to replace both methanesulphonyloxy groups in 3 $\beta_2$ , 17 $\beta$ -dimethanesulphonyloxy - 5 $\alpha$ -androstane by 3 and 17 Atrimethylammonium functions through the direct action of trimethylamine. Carlier attempts employing dimethylamine on the 33, 173- dimethanesulphonyloxy- compound had failed to give the corresponding bis tertiary amine, 3 00,17 of bis (dimethylamino) - 5%androstane - replacement of the 3 B-methanesulphonyloxy group only being achieved<sup>28</sup>

In the event a number of experiments employing different reaction conditions failed to give the required product. Water soluble materials, bitter in taste as would be expected of the desired product, consistently gave high nitrogen analyses and none of the crude products showed any neuromuscular blocking activity on the cat<sup>29</sup>. Subsequent work by another worker<sup>30</sup> in this laboratory directed towards the preparation of 30, 17 Q- diazido-50(- androstane (V) employing azide ion as the attacking nucleophilic species on 3,  $\beta$ , 17,  $\beta$  -dimethanesulphonyloxy - 5 $\alpha$  androstane in accordance with the known superior yields resulting in the steroid field when this ion is employed in place of an amine<sup>31</sup> has shown that in the course of the reaction the N-methyl pyrrolidone is destroyed. Thus it could well be that the products with the high nitrogen content obtained in the present work are formed from the N-methyl pyrrolidone. In view of the promising results obtained with the azide ion<sup>30</sup> the current project was abandonned since the successful preparation of 306, 1702-diazido- 502androstane gives a ready route to the desired bisquaternary salt via reduction to the di-primary

-59-

amine<sup>32</sup> and exhaustive methylation.



## EXP RIPRIAL

Melting points were taken on a hot-stage melting point apparatus and are uncorrected. N-methyl pyrrolidone was dried over sodium hydroxide pellets and distilled, the fraction b.p. 201-203° being collected. Tertiary-butyl alcohol was distilled from sodium and the fraction b.p. 32-33° collected. The 3 $\beta$ , 17 $\beta$ -dimethanesulphonyloxy-5 $\alpha$ -androstane employed in these experiments was prepared by the sodium in ethanol reduction of 5 dandrostane - 3,17dione as described by Alauddin<sup>28</sup> Authenticity of the dimethanesulphonate so prepared was confirmed by comparison of infra-red spectra and undepressed mixed melting point, 15)-152°, with an authentic A number of experiments were performed sample. of which the following is typical.

Attempted Su Displacement of 3A 17A-dimethanesulphonyloxy-5- A- Androstane by Trimethylamine.

 $3\beta$ ,  $17\beta$ -dimethanesulphonyloxy-5 $\alpha$ -androstane (0.45g) in N-methyl pyrrolidone (2) ml) containing tertiary butyl alcohol (0.5 ml.) and anhydrous trimethylamine (20 ml.) was placed in a pressure bomb (95 ml. capacity) the bomb closed and then heated at 206°C for 25 hr. After cooling for 14 hr. the bomb was opened and the excess trimethylamine allowed to spontaneously evaporate. The reaction mixture was then transferred to a flask and the excess N-methyl pyrrolidone evaporated under reduced pressure. The brown residue thus remaining was extracted with distilled water (5x5 ml.) and the combined aqueous wash taken to dryness on a rotary film evaporator. The hygroscopic, bitter-tasting, tan-coloured crystalline residue was re-crystallized from othanol/ether to m.p. 233-240° (softens ca 130°) (Found; N,6.44;  $C_{27}H_{54}N_2O_6S_2$  requires: N,4.94%) Treatment of the quaternary compound with saturated ethanolic picric acid readily afforded the picrate m.p. 310° (decomp.) (Found: N,17.39;  $C_{37}H_{55}N_8O_{14}$  requires: N,13.46%)

Other experiments employing modified conditions also afforded material showing high nitrogen analyses.

-62-

Frlich and Morgenroth, "Studies In Immunity" 1. Wiley. New York, 1910, p. 24; cited by Albert in "Selective Toxicity" Methuen 2nd ad., London, 1960, p.23. Fischer, Ber., 1894, 27 2985; Langley, Proc. Roy. Soc., 1906, 376, 170; Lucas, J. Physiol 1907, 36, 113. 2. Martin, Biological Antagonism. Blakiston, New York, 1951; Woolley, "A Study of Antimetabolites", Wiley, New York 1952; Work and Work, "The Basis of Chemotherapy". Oliver and Boyd, Edinburgh, 1948; Albert, "Selective Toxicity", Methuen, London 1951. Erlenmeyer, Bull. Soc. Chim. Biol., Paris, 3. 1948, 30, 792; Friedman, Nat. Res. Counc. Wash. Publication 1951, 206;

Friedman, Data Presented at Section on Microbiological Deterioration, Gordon Research Conferences, New Hampton, New Hampshire, 19533

-63-

Meunier, <u>Bull. Soc. Chim. Fr.</u>, 1945, <u>12</u>, 517; Bradlow Vanderwerf and Kleinberg, <u>J. Chem. Educ.</u>, 1947, <u>24</u>, 433; Burger, <u>J. Chem. Educ.</u>, 1956, <u>33</u>, 362; Burger, "Medicinal Chemistry" Vol. I Interscience, New York, 1951; Schatz in "Medicinal Chemistry" ed. Burger 2nd ed. Interscience, New York 1960, p.72.

- 4. Cavallini, Farmaco, 1955, 10, 644. Cavallini and Massarini, [Medicin Pharmaceut. Chem., 1959, 1, 365.
- 5. Harper in "Fortschritte der Arzneimittelforschung", ed. Jucker Vol. 4, Birkhauser Basle, p.p. 221-294; Harper, J. Medicin Pharmaceut. Chem., 1959, 1, 467.
- 6. Woolley in "Fortschritte der Arzneimittelforschung",
   ed. Jucker, Vol. 2. Birkhauser, Basle, 1960, p.p.613-636.
- 7. Clinton, Manson, Stonner, Neumann, Christiansen, Clark, Akerman, Page, Dean, Dickinson and Caratabeas, J. Amer. Chem. Soc., 1961, 83, 1478.
- 8. Crum Brown, and Fraser, <u>Trans. Roy Soc. <u>Edinburgh</u> 1869, <u>25</u>, 151, 693.</u>
- 9. Barlow and Ing, Brit. J. Pharmacol., 1948, 3, 298.

Gill, Proc. Roy. Soc., 1959, <u>B150</u> 381. Schueler. <u>Arch. Int. Pharmacodyn</u>, 1953, <u>93</u>, 417; Barlow, "Introduction to Chemical Pharmacology" Methwen, London, 1955;

Barlow in "<u>Steric Aspects of the Chemistry and</u> <u>Biochemistry of Natural Products</u>", Biochemical Society Symposia No. 19, University Press, Cambridge, 1960;

- Paton and Zaimis, <u>Nature</u> 1948, 162, 810;
   Paton and Zaimis, <u>Brit. J. Pharmacol</u>, 1949, 4, 381;
- 11. Del Castillo and Katz, J. Physiol. 1955, 128. 157; Del Castillo and Katz, Progress in Biophysics and Biophysical Chemistry, 1956, 6, 121; Del Castillo and Katz Proc. Roy. Soc., 1957, B146, 339.
- 12. Waser, In "Curare and Curare-Like Agents", ed. Bovet, Bovet-Nitti and Marini-Bettolo, Flsevier, Amsterdam, 1959, p.p. 219-229; Waser, <u>Pflugers Arch. Gcs. Physiol</u>, 1962, 274, 431;

13. Loewe and Harvey, Arch. Fxp. Path. Pharmakol., 1952, 214, 214; Fakstorp, Pederson, Poulsen and Schilling, Acta Pharm. Fox. Kobh., 1957, 13, 52.

- 14. Ehrenpries, Science, 1959, 129, 1613;
  Ehrenpries, Fed. Proc., 1959, 18, 220;
  Ehrenpries, <u>Biochim. Biophys. Acta.</u> 1960, 44, 561;
  Nistratova and Turpaev, <u>Biokhimiya</u> 1961, 26, 952
  (Through <u>Chem. Abs.</u>, 1962, 56, 1860c).
- 15. Bernauer, Berlage, von Philipsborn, Schmid, and Karrer, Helv. Chim. Acta 1958, 41, 2293; Battersby and Hodson, Proc. Chem. Soc., 1958, 287; Battersby and Hodson, J. Chem. Soc., 1960, 736.
- 16. Haining and Johnston Brit. J. Pharm. Chemotherap.. 1962, 18, 275 and references cited therein.
- Lottringhaus, Kerp and Preugschas, <u>Arzneimittel-</u> <u>Forsch</u>, 1957, <u>7</u>, 222; cited by Burger, "Medicinal Chemistry" 2nd Ed. Interscience, London, 1967, p.499.
- Elworthy, Paper delivered to the British Pharmaceutical Conference, London, Sept. 2-6, 1963.
- 19. Long and Scheuler. J. Amer. Pharm. Assoc., Sci. Ed., 1954, 43, 79.
- 20. Corey and Pauling Proc. Roy. Soc., 1953, B141, 17.
- 21. Popovici, Gesheckter, Reinovsky and Rubin, Proc. Soc. Expt. Biol. and Med., 1950, 74, 415.

- 22. Koshland, Proc. Nat. Acad. Sci. Wash., 1958, 44, 98.
- 23. Dr. P.H. Elworthy, personal communication.
- 24. Cavallito and Gray in "Progress In Drug Research" ed. Jucker, Vol. 2, Birkhauser, Basle, 1960 p.p.135-226.
- 25. Janot, Laine, and Goutarel, <u>Ann. Pharm. France.</u> 1960, <u>18</u>, 673. Quevauviller and Laine, ibid., 1960, <u>18</u>, 678.
- Haworth, McKenna and Powell, J. Chem. Soc., 1953, 1110.
   Dodgson and Haworth, J. Chem. Soc., 1952, 67.
- 27. Henbest and Jackson, J. Chem. Soc., 1962, 954.
- 28. A. Alauddin Ph.D. Thesis, University of Glasgow, November, 1962.
- 29. Dr. T.C. Muir, personal communication.
- 30. Dr. B. Caddy, personal communication.
- 31. Bose, Kistner and Farker, J. Org. Chem., 1962, 27. 2925.
- 32. Boyer, J. Amer. Chem. Soc., 1951, 73, 5865; Adams and Blomstrom, J. Amer. Chem. Soc., 1953, 75, 3405; Vander Werf, Heisler and McFwen, J. Amer. Chem. Soc., 1954, 76, 1231;

Bretschneider and Hormann, <u>Monat</u>, 1953, <u>84</u>, 1021; Bretschneider and Karpitschka <u>Monat</u>, 1953, <u>84</u>, 1043.

-67-

The Structure of Aristolactone, A 10-Membered

Carbon Ring Sesquiterpene Lactone from

Aristolochia reticulata L. and

Aristolochia serventaria L.

## INTRODUCTION

-69-

The plant order Aristolochiales embraces three families, Rafflesiaceae, Hydnoraceae and Aristolochiaceae. The first two consist of parasitic species whilst the third, which is made up mainly of climbing plants with woody stems, is composed of some six genera containing about 200 species. Of these approximately 180 belong to the genus <u>Aristolochia</u>, a genus having world-wide distribution<sup>1</sup>.

Aristolochia species are known to have been used in folk medicine since about the fourth century B.C.<sup>2,3</sup>\_their use which was held in high esteem by the ancient Greek, Roman, and Hebrew physicians being varied. Thus they are reputed to have been employed in child-birth, in the treatment of cancer, wounds, ulcers, abscesses, fevers, asthma and epilepsy, as bitter tonics and purgatives, and for treating snake-bite<sup>14</sup>. The Chinese are reported to have used <u>A. contorta</u> and <u>A. kemmferi</u> in treating similar afflictions. Various species of <u>Aristolochia</u> were apparently also used by the early North American Indians as a snake-bite remedy<sup>6</sup>?<sup>7,8</sup> and to this day Aristolochia species are still employed in this way in Brazil <sup>9</sup> and Mexico<sup>10</sup>, although examination of the extractives of a number of species showed them to be ineffective as antidotes to snake poisoning<sup>11</sup>.

An indication of the possible value of Aristolochia species in child-birth was given by Shaw's isolation of an unidentified alkaloid from A. elegans which produced uterine contractions12. Later the aporphine alkaloid magnoflorine (I) was isolated from A. debilis13, A. Kempferilt and A. clematiting 15,16 but no reports of its pharmacological properties appear to have been published. Extracts of Aristolochia species have also been shown to be bacteriostatic towards Staphylococcus aureus<sup>17</sup> and other bacteria<sup>18</sup>. Very recently Kupchan and Doskotch19 have reported that aristolochic acid. (II) ( a constituent of a number of Aristolochia species) exhibits tumour-inhibiting activity against adrenocarcinoma 755, in mice, thus providing a renewed interest in the genus.

-70-







Another interesting compound recently isolated from the genus by Furukawa and Soma<sup>20</sup> is the sesquiterpenoid Metone, aristolone (III) which occurs in <u>A. debilis</u> and which is probably derived from the maliane skelcton(IV) via a 1,2 - methyl shift. The stereochemistry of aristolone was determined by Buchi, Greuter, and Tokoroyama<sup>21</sup> employing nuclear magnetic resonance techniques.





Apart from the compounds just mentioned, the very extensive chemical studies which have been performed on the genus <u>Aristolochia</u> have resulted in the identification of a large number of other organic constituents. A summary of the constituents so far isolated from <u>Aristolochia</u> species is given in Table 1.

Α	D	T		
 <b></b>	2	4.4	2.1	

			1		*
SPECIES	CONSTITUENT	FIG.	FORMULA	m.p.oc	REF.
<u>A</u> .	aristolochine		C15H28N03	265	
argentina Griseb.	palmityl phytosterolin		C42H7402	82	
42.200.00	aristinic acid	175	C18H13N07	275	722
	aristolic acid	4	C15H11N07	260-270	
	aristidinic acid		C18H13N07	260	
	aristolochic acid I.	II	C17H11NO7	290	J
Ao					
<u>bracteata</u> Retz.	Aristolochic acid I.		C17H11N07	287-292	23,24. *
A	magnoflorine	I	C20H24 NO4	21+8-21+9 (10d1d.e)	15,16
<u>clematitis</u>	aristolochine		C32H22N2013	215	2
L.	choline	V	C5H14NO		15
	aristolochic acid I.	II	C17H11N07	287-292	25
	aristolochic acid II.				26
	nor-aristolochic acid	VI	C16H19N06	209	27

SPECIES	CONSTITUENT	FIG.	FORMULA	п.р.ос	REF.
Α.	neutral compound		C18H280	137	2
<u>cymbifera</u> Mart.	crocetin dimethylester	VII	C22H2804	211-212	+
(A grandia	isobixin		C25H3004	215	228
flora Gomes)	allantoin	VIII	C4H3N403	221	20
60mp 67	A. <u>cymbifera</u> acid		С <sub>20</sub> H <sub>32</sub> O2 ог С21H34O2	107	J
	A-sitosterol	IX	C29H500	140	森
40	aristolochic				
debilis	acid I	11	C17H11N07	290	29,30
Sieb at Zuco	debelinic acid		C18H13N07	350	31
	aristolochic acid "C"		C16H19N07	280	32
	aristolactam	X	C17H11N07	305	32
	magnoflorine	I	C20H24N04	Spire	14
	aristolone	III	C15H220		20
4.	aristolochine		C17H19NO3	215	
<u>indica</u>	iso aristolochic acid		C17H11NO7	275	
	phytosterolin				>33
	1 shwarene		C15H24		
	1 shwarone		C15H220		
	ishwarol		C15H24 0		J

Const Ituent	FIG	FORMULA	л.р.°С
allantoin oleic acid	VIII	СцНаМ403 Станано	

RE.F.

<u>A</u> .	allantoin	VIII	CLH8N403		7
indica	oleic acid	-	C18H3402		
L.	linoleic acid		C18H3202		
	palmitic acid		C16H3202		
	stearic acid		C18H3602		
	lignoceric acid		C24H2+802		>33
	cerotic acid		C26H5202		
	glycerol		Сзн803		
	ceryl alcohol				
	phytosterol		5.224	137-138	J
	aristolochic acid I.	II	C17H11N07	290	33 <b>, 3</b> 4
A- kempfer1	magnoflorine	I	C20H24NO4		13
Willd.	aristolochic acid I.	II	C17H11NO7	290	14
<b>A</b> .	aristolochic			v . *	
L.	acid I	II	C17H11N07	290	34
A. <u>maxima</u> Jacq.	aristolochic acid I	II	C17HnN07	290	35

SPECIES

SPACIES	CONSTITUENT	FIG	FORMULA	ш.р. ос	REF.
A. pandurata Jacq.	aristolochic acid I	II	C17H11N07	290	35
<b>A</b> -	Terpene		C10H16		36
reticulata	acetic acid		C2B+02	-	7
Lo	malsic acid		_		
	ogalic acid				
	aristolochine		+		-37
	D/+ Glucose				
	water-insoluble acid		C5H902	approx 65	
	(-) - borneol	XI	C10H180		
	$(-) -\Delta^4$ carene	XII	C10H16		-38
	aristolactone		C15H2002	111	
	quatornary alkaloid		C17H20N03C1		
÷	isc-rhammetin	XIII	C16H1207	318-322	34
	aristolochic acid I	II	C17H11N07	290	
	aristo-red	XIV	C19H15N06		
	allantoin	VIII	C1+H8M1+03	221	]
	B-sitosteryl-1-A -D-glucoside		<sup>C</sup> 35 <sup>H</sup> 60 <sup>O</sup> 6		> 39
	reticulene	XV	C15H24	_	J

SPECIES	CONST IT UF.NT	FIG,	FORMULA	m.p°C	REF.
A					
rotunda	aristolochine		C32H22N2O13	215	40
<u>A</u> .	borneol	XI	C10H180		41
<u>serpentaria</u>	<b>B-sitosterol</b>	IX	C <sub>29</sub> H <sub>50</sub> O	140	42
Lo	B-sitosteryl -l-B-D- glucoside		с <sub>35<sup>H</sup>60<sup>0</sup>6</sub>		39,42
	aristolochic acid I	II	C17H11N07	290	34,35
	aristo-red	XIV	C19H15NO6		39
	aristolactone		C15H2002	111	39
A. sipho L'Herit	aristolochic acid	II	C17H11NO7	290	43

## # present work

f quoted as aristolochic acid I by Boit, "Ergebnisse Der Alkaloid-Chemie Bis 1960," Akademie Verlag, Berlin, 1961, p. 271.



v



HO

VII

IX



VIII







X

XI

XII



XIII

VIX.



XA

In the course of a general investigation of representative <u>Aristolochia</u> species, Stenlake and Williams isolated from the petroleum ether extractives of <u>A. reticulata<sup>38,144</sup></u> and later from the corresponding fraction of<u>Aserbentaria<sup>55</sup></u> a crystalline lactone which they designated aristolactone. Although their investigations resulted in the tentative assignment of a structure for aristolactone, there were several inconsistencies in the experimental data necessitating further work on the problem. Accordingly the present investigation was undertaken and in the event resulted in a revised structure for this compound.

Aristolactone, m.p. 110.5 -111°, [a],14+156.4 (BtOH) was assigned the empirical formula C15H2002 on the basis of elemental analyses and Raast molecular weight determination. Hydrogenation established the presence of three double bonds. Calculation of double bond equivilents (C. H. 2012 15 32 32-20 =6) thus showed that aristolactone could contain only one ring other than the lactone ring. The presence of the lactone ring was established by the hydrolysis of aristolactone to an hydroxyacid containing all the carbon atoms of eristolactone in quantitative yield. Aristolactone was initially thought to exhibit an ultraviolet absorption maximum at 211 mm (E,11,500) 30 14 12 but later Steele using more accurate techniques showed aristolactone to exhibit an absorption maximum with E,11,500 at 205 mu. Ozonolysis of aristolactone yielded formaldehyde (collected as the dimedone derivative, 5,46 thus indicating the presence of at least one vinylidene group.

Careful hydrogenation of aristolactone over palladium on charcoal in ethanol<sup>1,1,4,47</sup> afforded a dihydrolerivative, m.p. 79-80.50, ap17-77, (EtOH), 209 mµ (£,7800) which on ozonolysis no longer gave formaldohyde. When hydrogenation of aristolactone was carried out using Adam's catalyst, 2 moles of hydrogen were rapidly absorbed yielding an oil whose properties were indicative of a mixture of double bond isomers. Exhaustive hydrogenation of aristolactone over prereduced platinum oxide 44,47 gave a total hydrogen uptake of 3 moles, to yield a crystalline compound, C15H2602, m.p. 103.5-1040, [A] 1743.0° (EtOH) initially termed hexahydroaristo lactone but later termed isohexahydroisoaristolactone45. The hexahydrocompound, wilike aristolactone, was found to be stable to cold alkali, but was hydrolysed by hot alkali to a mixture of two hydroxy-acids. The major product C15H2803 exhibited m.p. 86-87°, and [0] 16+16° (Et OH), while the minor hydroxy-acid, which was isolated in only trace amounts, had m.p. 121-122°. The hexahydro-compound itself could be isolated in two forms, needles, m.p. 99-100° and platelets m.p. 101-102°. Since both showed identical

-80-

infrared spectra and optical rotations, and on hydrolysis afforded the same mixture of hydroxy acids it was concluded that the two forms of the hexahydrolactone (which on admixture gave an elevated melting point, 103-104°) were dimorphs and not stereoisomers, which might well be expected to arise on reduction of the three double bonds present in aristolactone. The large shift in optical rotation encountered in the reduction products of aristolactone was attributed to one or more asymmetric centres being in close association with a double bond <sup>144,48</sup>,

Williams<sup>44,47</sup> observed that treatment of aristolactone with cold potassium hydroxide in ethanol also produced a marked change in optical rotation. The initial increase which reached a maximum of  $(\mathcal{A}_D^++317^\circ)$  in 50 minutes, was shown to occur without the consumption of alkali, whereas the subsequent decrease in rotation was accompanied by an uptake of one mole of base. Work-up of the reaction mixture at the point of maximum rotation afforded a keto-ester which was designated ethyl oxoaristate. Subsequently it was discovered that the use of methanolic potassium hydroxide afforded the corresponding methyl ester in superior yield  $^{44}$ ,  $^{47}$ . Formation of the keto-ester was attributed to attack by alkowide ion on a proposed a $\beta$ -unsaturated lactons this being followed by hydrolysis of the ester function under the further influence of hydroxide ion in accord with scheme A (partial structures XVI to XX).

The keto ester, ethyl oxoaristate,  $C_{17}H_{26}O_{3}$  had m.p. 56-57°,  $[a]_{D}$ + 317° (EtOH) //max 1726 and 1168 cm<sup>-1</sup> (ester) and 1704 cm<sup>-1</sup> (shoulder, ketone) in paraffin mull, and exhibited a low intensity ultraviolet ketone maximum at 291 mp ( $\mathcal{E}$ ,250). Methyl oxoaristate had m.p. 68-69°,  $[a]_{D}$ + 342° (EtOH)<sup>47</sup>. Hydrogenation of methyl oxoaristate over prereduced platinum oxide afforded methyl dihydroomoaristate,  $C_{16}H_{26}O_{3}$ , m.p. 68-68.5°,  $[a]_{D}^{17}$  + 152°, (EtOH) Å max 290 mp ( $\mathcal{E}$ ,160) and end absorption at 210 mu ( $\mathcal{E}$ ,3600)<sup>45</sup>. Further reduction gave the fully saturated tetrahydro derivative as an oil<sup>44,46</sup>. Similarly hydrogenation of ethyl oxoaristate afforded a dihydro derivative, m.p. 65-66°,  $[a]_{D}$ + 131°, max 287 mp ( $\mathcal{E}$ ,52) and end absorption at 208 mp

 $(\mathcal{E},3570)^{45}$ . On further reduction it gave the fully saturated tetrahydro derivative, again as an oil<sup>44,46</sup>.

Steels 45,46 questioned the original proposals

-82-





XVI

XVII





XX

XIX

SCHEME A

in which aristolactone was considered to be an  $q\beta$ unsaturated lactone, and pointed out that  $a\beta, \gamma$  unsaturated lactone would be more consistent with the observed ultraviolet maximum at 205 mp, since and A-unsaturated lactone would be expected to exhibit a maximum at 215-225 mu49. This reassignment of the lactone double bond of aristolactone  $\beta >$  to the lactone, as shown in partial structure XVII still permitted rationalization of the formation of methyl- and ethyloxoaristates (XVII to XIX in scheme A). In addition, this new assignment of the double bond explained the negative Legal test<sup>50</sup> and the nonformation of an ammonia adduct from aristolactone, (a reaction typical of Af-unsaturated lactones 51, 52). Further, the presence of a  $\beta\delta$ -double bond explained the absence of pyruvic acid in the oxonolysis products of aristolactone 44,46 and the partial iodine values given by this compound 53.

Steele<sup>45,46</sup> adduced what he considered to be better evidence for the presence of a  $\beta$ , 3unsaturated  $\beta$  lactone system in aristolactone when he obtained what was concluded to be a



1,4- ketal from the lithium aluminium hydride reduction of aristolactone. Anog- unsaturated, S-lactone would have given rise to an allylic 1,4- diol. The presumed ketal was assigned the name oxoaristaldehyde<sup>45</sup>. Analysis indicated the empirical formula C15H2202. The compound had m.p. 197-198°, [x] 16. 5+ 92° (EtOH) /max 1752, 1733 cm-( shoulder ) and max 284 mp (£,58), and was assigned nartial structure XXI (scheme 3). Lithium aluminium hydride reduction of aristolactone under slightly more vigorous conditions yielded a compound named oxoaristool, C15H2402, m.p. 245-246° which was assigned the partial structure XXII (scheme 3). On lithium aluminium hydride reduction of the hexahydro-derivative of aristolactone, Steele46 obtained the expected 1,4-diol, C15H3002, m.p. 106-107°, [] 20 +18.7° (EtOH) partial structure XXIII (scheme B).

-244m

On treating aristolactone with a number of acidic reagents both Williams<sup>1,1,1</sup> and Steele<sup>146</sup> obtained isoaristolactone,  $C_{15H_{20}O_2}$ , m.p. 90-91°,  $[O]_D$ . (EtoH) max 209 mp (£,11,200) and 272 (£,640). This isomer was affected by base only under very vigorous conditions, a

fact which was interpreted to mean that the double bond considered to be  $\beta$  -to the lactone carbonyl in aristolactone was the site of isomerization. The large change in optical rotation accompanying the formation of isoaristolactone ([], +156-[], 44°) was assumed to arise from the generation of a new asymmetric centre. Hydrogenation of isoaristolactone was found to afford the same dihydro- and hexahydroderivatives as were obtained from aristolactone. This was taken as support for the hypothesis that rearrangement of the double bond  $\beta$  to the lactone carbonyl in aristolactone also occurred on hydrogenation of this compound as illustrated by the conversion of partial structure XVII into XXIVa or XXIV.



XXII

XXIVa

XXIVb

Partial structure XXIVb was eliminated on the grounds that the action of lithium aluminium hydride

on isoaristolactone afforded a compound,  $C_{15}H_{22}O_{2}$ ;  $[\alpha]_{D}$ - 50,1(EtOH) considered to be a ketone as it exhibited max 290 (£,30) and end absorption at 209 mµ (£,4960). Such a product might be expected

to arise from XXIVa but not from XXIVi which would form a diol.

In view of the identical reduction products obtained from aristolactone and isoaristolactone the nomenclature of the dihydro- and hexahydroderivatives was amended to dihydroisoaristolactone and hexahydroisoaristolactone respectively.

Williams<sup>44</sup> carried out a number of dehydrogenation experiments on aristolactone employing palladium on charcoal, but was unable to demonstrate the presence of either azulenic or naphthalenic products. However, benzenoid absorption in the ultraviolet was observed in the dehydrogenation products of both aristolactone and hexahydroisoaristolactone, although none of these products was isolated in pure form. Steele<sup>46</sup> using a slightly modified dehydrogenation procedure on certain semisolid residues obtained as by-products from the preparation of various derivatives of

-86-
aristolactone, isolated a purple azulenic product. This azulene, aristazulene, exhibited ultraviolet absorption maxima at 245, 279, 289, 333 and 348 mµ and visible light absorption maxima at 553, 561, 585, 593 and 642 mµ. In 50% sulphuric acid aristazulene showed absorption maxima at 227, 269, and 374 mµ. Steele drew the firm conclusion that aristazulene was a 2,4,8- trisubstituted azulene since its spectrum was virtually identical with the published spectra of typical 2,4,8- trisubstituted azulenes, in both the ultraviolet<sup>54,55</sup> and the visible<sup>55,56</sup>, <sup>57</sup> ranges. He also drew attention to the possibility that aristazulene was identical with vetivazulene (XXV)<sup>54,57</sup>.

XXV

-87-

When Williams treated aristolactone with potassium permanganate in ice-cold acetone he obtained a neutral crystalline product, C15H2204, m.p. 158.5 - 160°, 10+128°, concluded to be a 1,2- diol. Since sodium metaperiodate oxidation of this product afforded no formaldehyde it was concluded that the diol had not formed at the site of a vinylidene double bond but that it was derived from the double bond placed \$, 8-20 the lactone carbonyl. Further support for this contention was considered to be provided by the facts that there was little change in the optical rotation in passing from aristolactone to the diol, that the lactone ring was intact, and that the iodine values were in good agreement with the presence of two double bonds. The diol was accordingly represented by partial structure XXVI.



## Subsequent hydrogenation of the diol by Steele<sup>46</sup> afforded two compounds, termed dihydrodihydroxyaristolactone, m.p. 135-136°, end absorption at

-- 88--

210 mµ (£,30)0), and the fully saturated tetrahydrodihydroxyaristolactone,  $C_{15}H_{26}O_{4}$ , m.p. 123-124°,  $\left[\alpha\right]_{D}^{20}$ +32.2 (EtOH).

From the chromic acid oxidation of aristolactone, Williams isolated acetic acid in quantity indicative of the presence of two

C-CH<sub>3</sub> groups. Also formed were an unidentified keto-acid, and succinic acid. Since the presence of glutaric acid, an expected oxidation product of partial structure XXVII, could not be detected in the oxidation products, structures possessing this unit were eliminated from consideration as possible structures for aristolactone

> C-CH<sub>2</sub>-CH<sub>2</sub>-CH<sub>2</sub>-C XXVII

From the chromic oxide oxidation of isoaristolactone Steele<sup>46</sup> was able to isolate formic acid (derived from a vinyl group), acetic acid (derived from C-CH<sub>3</sub> groups), and succinic acid. In addition an unidentified dibasic acid thought to contain 8 to 10 carbons on the basis of its behaviour on paper chromatography was found to

-- 89--

be present in the reaction products.

After treating methyl tetrahydro-oxoaristate with hot nitric acid, Steele isolated another dicarboxylic acid which he characterized as the silver salt,  $C_{12}H_{20}O_4$  Ag<sub>2</sub>. This dibasic acid was considered to be 3,7-dimethyldecanedioate (XXVIII) solely on the grounds that it had an Rf. value greater than that of pimelic acid(XXIX) when subjected to paper chromatography.  $^{146}$ 



Both Williams<sup>44</sup> and Steele<sup>46</sup> found that ozonolysis of aristolactone yielded formaldehyde (60% of theoretical yield based on one vinyl group) as the only volatile fragment. The non-volatile residues contained two keto-acids, one of which was suggested by Williams<sup>44</sup> on the basis of paper chromatographic studies to be laevulenic acid(XXX). Steele<sup>46</sup> however pointed out the similarity of its Rf. value to that of acetoacetic acid(XXX), although the known

-90-

instability of this compound would make its isolation highly unlikely.



XXX XXXI The second keto-acid was not characterized, but was found to yield acetaldehyde on reozonolysis. No acetone or pyruvic acid could be detected in the ozonolysis products.

Similarily ozonolysis of isoaristolactone<sup>44</sup> gave formaldehyde (53% of theoretical yield based on one vinylidene group) as the only volatile product. Dihydroisoaristolactone, however, gave no volatile carbonyl containing fragment on ozonolysis, indicating the absence of a 1.1- disubstituted double bond in that compound.

On ozonolysis, ethyl oxoaristate<sup>1,1,1</sup> also yielded formaldehyde (4,3% of theoretical yield based on one vinylidene group) as the only volatile product. The non-volatile residues were not characterized. Ethyl dihydro-oxoaristate afforded no volatile fragments on ozonolysis indicating the absence, as in dihydroisoaristolactone, of any vinyl double bond. Identical observations were made for methyl dihydrooxparistate, although here the oily non-volatile product, b.p. 22)-24)  $^{\prime}$ .4 mm/showing positive reactions for a methyl ketone gave good analyses for C<sub>16</sub>H<sub>24</sub>O<sub>5</sub>.

The non-formation of acetone or 2,2-dimethylacetaldehyde in the ozonolysis of all these aristolactone derivatives precluded the presence of an isopropylidene(XXXII) or a 2-methylpropylidene (XXXIII)group.



RECUI

#### XXXIII

The infrared spectrum of aristolactone in carbon tetrachloride solution was stated<sup>144</sup> to exhibit a band at 1770 cm<sup>-1</sup> typical of a  $\chi$ -lactone and this absorption was said to shift to 1780 cm<sup>-1</sup> in hexahydroisoaristolactone but the present work (<u>vide infra</u>) employing a Unicam S.P. 100 instrument showed that these frequency assignments were not accurate. The fact that the ester stretching vibrations at 1064 and 1034 cm<sup>-1</sup> in aristolactone were replaced by a single band in the hexahydroisoderivative at 1167 cm<sup>-1</sup> was taken<sup>144</sup> to indicate that hydrogenation had caused marked changes in the lactone environment. Bands at 1650 and 890 cm<sup>-1</sup> in the infrared spectrum of aristolactone were attributed to vinylidone absorption, and peaks at 840, 800 (weak) and 782 cm<sup>-1</sup> were assigned to the two other double bonds. These bands were absent in the fully saturated hexahydroisoaristolactone<sup>144</sup>.

The infrared spectrum of isoaristolactone in carbon tetrachloride solution also showed typical vinylidene bands at 1656 and 392 cm<sup>-1</sup> equal in intensity to those of the parent lactone. However peaks attributed to the other double bonds in aristolactone were absent in the iso-compound, being replaced by new bands at 833 and 815 cm<sup>-1</sup> which were assigned to a newly formed trisubstituted double bond. Dihydroisoaristolactone contained these new bands but the vinylidene absorption was

-93-

absent, an observation consistant with the nonformation of formaldehyde on ozonolysis of this compound<sup>1,1</sup>.

Ethyl oxoaristate ( liquid paraffin mull) was found to exhibit infrared maxima at 1726 and 1186 cm-1 (ester group) - the former peak showing a shoulder at 1704 cm<sup>-1</sup> (ketone carbonyl). Methyl oxoaristate showed similar absorptions. Examination of the ethylenic absorption region of methyl oxoaristate in carbon tetrachloride solution showed the presence of vinylidene absorption at 1650 and 890 cm<sup>-1</sup> and a new trisubstituted double bond neak at 813 cm<sup>-1</sup>. In addition the bands at 840, 800, and 782 cm<sup>-1</sup> in the infrared spectrum of aristolactone and isoaristolactone were present in the infrared spectrum of methyl oxoaristate though reduced in intensity".

Williams<sup>1,1</sup> observed that the infrared spectrum of hexahydroisoaristolactone showed marked similarity to the spectrum of tetrahydroalantolactone (XXXIV), especially with respect to the absorptions in the 1000 and 950 cm<sup>-1</sup> regions, but the two compounds were found not to be

-94-

identical (although of course this could have been attributable to stereoisomerism).



#### XXXXIV

Absorption in these regions had been regarded by Marrison<sup>58</sup> as typical of cyclohexane rings leading Williams to conclude that aristolactone contained a cyclohexane ring. Further evidence that aristolactone might contain a six-membered ring system was adduced by Williams on the grounds that the two bands at 1008 and 951 cm<sup>-1</sup> in aristolactone were replaced by a single band at 1100 cm<sup>-1</sup> in methyloxoaristate, observations paralleling those of Lecompte<sup>59</sup> on cyclohexanone derivatives.

Williams<sup>4,4</sup> accordingly examined all possible C15 structural formulae possessing cyclohexane ring systems and a X-lactone which obeyed the isoprene rule<sup>60</sup> but found no structure for aristolactone among these which would accommodate all the accumulated experimental evidence.

-95-

Steele<sup>46</sup> reexamining the infrared evidence drew attention to the work of Prelog and his group 61,62 wherein bands quoted by Williams as being typical of a cyclohexane ring system were found also to be present in the spectra of cyclononane or cyclodecane and their derivatives. This information coupled with the discovery at about that time of a number of sesquiterpenoid lactones containing a cyclodecane ring<sup>63,64,65</sup> suggested to Steele<sup>46</sup> that aristolactone might well be derived from a 9 or 10 membered ring system. The monocyclic nature of aristolactone together with the formation of a 2,4,8- trisubstituted azulene on dehydrogenation made the cyclodecane system seem more likely, in light of the isoprene rule<sup>60</sup>.

-96-

Steele<sup>46</sup> accordingly postulated aristolactone as XXXV, isoaristolactone as XXXVI, dihydroisoaristolactone as XXXVII, hexahydroisoaristolactone as XXXVIII, methyl oxoaristate as XXXIX, and dihydro- and tetrahydromethyl oxoaristate as XL and XLI respectively. -97-







XXXX

NOVE VIE

TIVICOL







XXXXVIII







XLI

From a consideration of the Klyne-Hadson lactone rule<sup>66</sup> Steele<sup>46</sup> further suggested the partial stereochemistry shown in XXXVIIIa for hexahydroisoaristolactone.



Upon a re-examination of the evidence the structure of aristolactone XXXV was revised by Stenlake, Steele and Williams45 to(XLII) which contains two vinylidene groups. Since dihydroisoaristolactone (XXXVII) contains no vinyl groups. formation of the same dihydro-compound from both aristolactone and isoaristolactone was suggested to involve reduction of one vinyl group with concurrent rearrangement of the second vinyl group to a trisubstituted double bond, as in Thus the nomenclature of structure XXVII. the dihydro- and hexahydro-lactone was altered to isodihydroisoaristolactone and isohexahydroisoaristolactone. A similar rearrangement was postulated to occur in the formation of methyl

dihydro- and ethyl dihydro-oxoaristates.



### XLII

However, acceptance of structure XLII leaves certain conflicting evidence. For example, the anomalous infrared carbonyl absorption maxima in oxoaristaldehyde (XLIII) and isooxoaristaldehyde<sup>45</sup> is unexplained. Similarly the low infrared absorption values for the proposed enol lactone function in aristolactone XLII coupled with the absence of a frequency shift in the fully saturated isohexahydroisoaristolactone XXXVIIIa is not readily rationalized on the basis of structure XLII. It is of interest that a degradation product of arctiopicrin of proven structure XXXVIII<sup>67</sup> is obviously not identical with the hexahydro-lactone. However as there are five asymmetric centres in this compound giving rise to 2<sup>5</sup> theoretical stereoisomers

This evidence is not necessarily conclusive. Low formaldehyde values on ozonolysis of aristolactone and its derivatives are not explained on structure XLII<sup>68</sup> nor are structures alternative to XLII precluded. Thus it was felt necessary to continue work on this problem so that the structure of aristolactone could be firmly established. This work has resulted in the assignment of structure XLIV to aristolactone.





MILLII

XLIV

#### DISCUSSION

As outlined in the introduction. certain unsatisfactory features in the evidence upon which the earlier structural proposals for aristolactone and its derivatives had been based necessitated further studies on the constitution of these compounds. Initially, all known derivatives were subjected to a reinvestigation of their infra-red spectra employing a Unicam model S.P. 100 double beam spectrophotometer equipped with an S.P. 130 sodium chloride prism - grating double monochromator operated under Vacuum conditions. These studies brought to light certain errors in frequency values in the earlier studies. Accordingly, all the infra-red frequencies quoted in this section (unless specification is made to the contrary) are those of the present determinations wherein measurements were made in dilute carbon tetrachloride solution. Aristolactone and certain of its derivatives were also subjected to nuclear magnetic resonance (n.m.r.) study in deuterated chloroform employing tetramethylsilane as internal standard. The instrument used was the Perkin-Elmer nuclear magnetic resonance spectrometer run at

40 megacycles per second.

The spectral studies quickly proved the structure XLII previously assigned to aristolactone to be untenable, and when taken in conjunction with new chemical evidence permitted structure XLIV to be assigned to aristolactone.

Despite the extensive nature of the earlier work on aristolactone and its derivatives 38, 39, 44-47, 50 no attempt had been made to secure the parent hydrocarbon. The assumption that aristolactone was derived from the germacrane skeleton, inferred from considerations of double bond equivalents and from hydrogenation and dehydrogenation evidence, had never been unequivocally proven. It therefore seemed that degradation of a suitable derivative, to the parent hydrocarbon would be a logical first step. The obvious line of attack appeared to lie in the preparation of the di-ptoluenesulphonate (ditosylate) or the di-methanesulphonate (dimesylate) of the so-called tetrahydroisoaristo-6,12-diol which had been previously prepared by the action of lithium aluminium hydride on hexabydroisoaristolactone<sup>45</sup> Hydrogenolysis of either disulphonate

-102-

by means of lithium aluminium hydride would then be expected to afford the parent hydrocarbon. This was indeed found to be the case although the hydrogenolysis of the disulphonate proved more complex than had been The diol. m.p. 106-107° was prepared anticipated. as previously described<sup>45</sup> and converted into the dimesylate. m.p. 79-81°. in good yield by treatment with a 2.2 molar ratio of methanesulphonyl chloride in pyridine at 0°. This derivative was chosen in preference to the ditosylate because of the greater ease of mesylate formation under mild conditions<sup>69</sup>. Further, the mesylate function is known to be a poorer leaving group than the tosylate group<sup>70</sup> and so is less likely to suffer nucleophilic replacement by a pyridine molecule to give a water soluble pyridinium salt as an unwanted by-product during the sulphonate ester formation - an important consideration in the present case where the starting diol was available only in spall quantity<sup>45</sup>. The dimesylate was found to be highly unstable quickly, exhibiting a pronounced fall in melting point which was accompanied by the appearance of a new absorption band at ca 1660 cm<sup>-1</sup>

-103-

in the infra-red, attributable to the generation of a double bond via elimination of methanesulphonic acid. Lithium aluminium hydride reduction of the freshly prepared dimesylate afforded a colourless oil which unexpectedly gave a positive tetranitromethane test for the presence of a double bond. Unsaturation was confirmed by the presence of a sharp band at 1660 cm<sup>-1</sup> in the infra-red spectrum of the oil. Gas-liquid chromatography (g.l.c.) employing a 50 metre capillary column coated with polypropylene glycol as the stationary phase showed the oil to consist of three components (table 2). After catalytic hydrogenation of the unfractionated oil over platinum oxide in ethanol (hydrogen uptake: 0.49 mole based on C15H30), g.l.c. employing the same capillary column showed the complete disappearance of the first peak (the numbering of peaks is in order of increasing retention time), diminution of the second, enhancement of the third, and the appearance of two new peaks (table 2). The infra-red spectral examination of the crude product showed that unsaturated material was still present. Accordingly the material was

-104-

subjected to further hydrogenation over highly active Adam's catalyst in ethanol (hydrogen uptake: 0.39 mole based on C15H30). This afforded a product which g.l.c. indicated to be free of the first two peaks present in the original product obtained from the dimesylate (table 2). Since the infra-red spectrum also showed the absence of unsaturation the first two peaks just discussed can now be assigned to unsaturated components. Further, it can be deduced that the third peak present in the original mixture is due to a saturated component, more of which is formed by catalytic reduction of the unsaturated material appearing as peak number one. At the same time reduction of the compound represented by peak number two gives rise to the saturated isomers represented by peaks four and five. Thus treatment of the dimesylate with lithium aluminium hydride gives rise both to hydrogenolysis and to elimination products in the ratio 15.5:84.5. Gas-liquid chromatography of an authentic sample of synthetic germacrane (XLV) employing the same capillary column under the same conditions showed the presence of three isomers, the retention times of which each coincided with that of one of the saturated hydrocarbons derived from

-105-

aristolactone. Admixture of the saturated hydrocarbons derived from aristolactone with the synthetic germacrane followed by g.l.c. confirmed that both materials contained the same three isomers. Further the infra-red spectra of both mixtures run as liquid films were superposable.

SAMPLE		RFLATIVE PERCENTAGE OF PEAKS* (BY TRIANGULATION)				
		1	2	3	4	5
1	LIALE, Hydrogenolysis Product	14.4	70.1	15.5	-	-
2	Partial Reduction	-	36.5	29	12	22.5
3	Full Reduction	-		30.1	20.8	49.1
4	Authentic Synthetic Germacrane	-	-	25.7	50.0	24.3

\*Peak numbers assigned in order of increasing

retention time.

TABLE 2



XLV

-106-

Thus it was unequivocally established that aristolactone unlike the sesquiterpene ketone aristolene, occurring in <u>Aristolochia debilis</u><sup>20</sup> and having structure XLVI<sup>21</sup>, is not a derivative of calarane (XLVII) (which can be envisaged as arising in nature <u>via</u> a 1,2-methyl shift in a maliane (XLVIII) derivative). Since rationalization of the isomers of Table 2 is best made in terms of the structure of aristolactone, account of their formation is left until page 125.



Comparison of the n.m.r. spectra of isoaristolactone<sup>47</sup> and methyl oxoaristate<sup>47</sup> with those of their dihydro derivatives shows the conversion of an isopropenyl group into an isopropyl group. Thus absorptions representing overlapping doublets which appear as barely resolved triplets with intensity 3 protons at 8.14 and 8.33? respectively in the former compounds are replaced by

a pair of superposed doublets (Jabc.p.g) of total intensity 6 protons at 8.94 and 9.11 Trespectively in the dihydro compounds. As is to be expected the complex absorption pattern in the 57 region is reduced in intensity by 2 protons. The values of 8.14 and 8.33 for the isopropylene methyl protons in isoaristolactone and methyl oxoaristate is in good agreement with the range 8.1 - 8.47 previously assigned to the methyl group of an isopropenyl function<sup>71,72</sup>. Furthermore this evidence for the presence of an isopropenyl group in isoaristolactone and methyl oxoaristate is in good agreement with the n.m.r. absorptions observed with geijerene (XLIX) 72 and the decalin derivative 121. The methyl group of the isopropenyl function in aristolactone itself is seen as a barely resolved apparent triplet of intensity 3 protons at 8.172.

XLIX

L

-108-

The superposed doublets at 8.947in dihydroisoaristolactone and at 9.117in the dihydro derivative of methyl oxoaristate assigned to the gem dimethyl group protons are also in excellent agreement with the absorption to be expected from an isopropyl group<sup>71,73,74</sup>. The n.m.r. evidence is also in full agreement with the infra-red evidence. Thus the absorptions in aristolactone, isoaristolactone, and methyl oxoaristate at 3075 cm<sup>-1</sup> and 895 cm<sup>-1</sup> which are characteristic of 1,1,-disubstituted ethylenes<sup>72,75a,76,77</sup> are absent in the corresponding dihydro compounds which instead exhibit doublets in the 1385 and 1375 cm<sup>-1</sup>

The presence of the isopropenyl function in aristolactone necessitates that the lactonic carbonyl function be derived from one of the ring methyl groups of germacrane. This conclusion, incidentally is consistent with the observations of Professor V. Herout<sup>78</sup> who has pointed out that the only known lactones derived from the germacrane skeleton in which the lactone carbonyl function is derived from the isopropyl group occur in plants of the

region which are typical of gem dimethyl groups 75b.

-109-

family Compositae to which <u>Aristolochia</u> species do not belong.

The infra-red absorption of the lactone carbonyl at 1765 cm<sup>-1</sup> in both aristolactone and isoaristolactone, at 1764 cm<sup>-1</sup> in dihydroisoaristolactone, and at 1769 cm<sup>-1</sup> in hexahydroisoaristolactone, enables the lactone ring to be designated as saturated and  $\chi$ -(75c. 79-82) in all four compounds. These absorption frequencies rule out the possibility that any of the three compounds, aristolactone, isoaristolactone, or dihydroisoaristolactone, is an ap-unsaturated lactone or unsaturated lactone having an exocyclic double bond from the *d*-position (such compounds being known to absorb in the range 1760 to 1740 cm-1 75c,83. Further confirmation is given by the lack of ultra-violet absorption attributable to and B-unsaturated lactone in these compounds 45,47. All unsaturated lactones having a double bond from the a-carbon atom must exhibit true conjugation as a consequence of the planar nature of the 5-membered lactone ring. Similarly the lactone carbonyl absorption frequencies of aristolactone and its derivatives mentioned above are inconsistent with any of these compounds being an enol lactone. Unsaturated

-110-

lactones with the double bond in the  $\beta_1$ %- position or having an exocyclic double bond from the  $\beta$ - position are known to absorb near 1800 cm<sup>-1</sup>, while the absorption frequency is lowered 20-40 cm<sup>-1</sup> to the range 1780-1760 cm<sup>-1</sup> on reduction to the corresponding saturated  $\aleph$ lactone<sup>75c</sup>. For example,  $\aleph$ - angelica- lactone (LI) exhibits lactone carbonyl absorption at 1799 cm<sup>-1</sup> when the infra-red spectrum is measured in chloroform (1806 cm<sup>-1</sup> when measured in carbon tetrachloride), whereas the derived saturated  $\aleph$ - lactone exhibits carbonyl absorption at 1775 cm<sup>-1</sup>, measured in chloroform<sup>84</sup>. Again parasantolide (LII) exhibits lactone carbonyl absorption at 1792 cm<sup>-1</sup> whilst its saturated derivativos show this maximum at 1764 cm<sup>-1</sup> <sup>85</sup>.



LII

That the lactone ring of the aristolactone series is derived from a secondary alcohol is confirmed on two counts. Firstly aristolactone gives rise to the keto-ester methyl oxoaristate<sup>45,47</sup> and secondly an apparent triplet (J=8 c.p.s.) of intensity 1 proton at 5.27 is present in the n.m.r. spectrum of hexahydroisoaristolactone, and this can be assigned to a single proton on the carbon atom bearing the alcoholic oxygen atom . This same absorption is discernible in the n.m.r. spectra of aristolactone, isoaristolactone and dihydroisoaristolactone although it merges with vinylic proton absorption. This absorption is absent from the n.m.r. spectrum of methyl oxoaristate.

The absence of any proton resonance above 8.6 tin the n.m.r. spectra of aristolactone, isoaristolactone, and methyl oxoaristate indicates that none of these compounds contain a methyl group bound to a carbon atom bearing hydrogen. Thus the second ring methyl group of germacrane must form part of a trisubstituted double bond system in all three compounds (no tetrasubstituted double bond in association with a ring methyl group being possible in the germacrane skeleton). Resonance of the protons of this methyl group

-112-

as expected 72-74, 82;88,89 appears as barely resolved doublets of intensity 3 protons at 8.527, 8.417, and 8.457 in aristolactone, isoaristolactone, and dihydroisoaristolactone respectively, at 8.467 in methyl oxoaristate, and at 8.527 in the latter's dihydro-derivative. In the fully saturated hexahydroisoaristolactone absorption from the protons of this methyl group appears as part of the doublet (J=6 c.p.s.) of intensity 9 protons at 9.067.

The 5 region of the n.m.r. spectra of aristolactone and isoaristolactone shows absorption from three Vinylic protons in both compounds. A fourth vinylic proton absorbing at <u>ca</u> 3.2 A is also present. On the basis of the total n.m.r. absorption of 20 protons there must be a third double bond present in addition to the isopropenyl double bond and the double bond associated with the ring methyl group in both compounds which, this n.m.r. evidence now shows, must be trisubstituted. Since the infra-red and ultra-vielet spectra, as already discussed, have eliminated the possibility that either compound could be and  $\beta$  - unsaturated or an enol lactone, this last double bond must lie in association with the isopropenyl group. The absence of conjugated double bond absorption in the ultra-violet similar to that found for 3,8(9)-pmenthadiens (LIII; Åmax 233.5;  $\mathcal{E},19000$ )<sup>90</sup>, must depend upon the inability of the diene system in the aristolactone series to assume a planar nature due to non-bonded interactions arising from the geometry of the 10-membered ring system. This point will be discussed in more detail on p. 146-147,



LIII

The n.m.r. evidence would further indicate that it is this last double bond in association with the isopropenyl group which is involved in the rearrangement of aristolactone to methyl oxoaristate<sup>45,47</sup>. Thus the n.m.r. spectrum of this keto-ester shows that the isopropenyl group and the methyl group trisubstituted double bond system are unchanged from aristolactone, whilst the very low field proton appearing as a triplet (J=3 c.p.s.) at 3.26 in aristolactone is not present in methyl oxoaristate.

The evidence thus far therefore reduces the possible formulae for aristolactone to six <u>Viz</u> LIVa to LIVf (numbering refers to the germacrane skeleton ) without taking into account geometrical isomerism about double bonds, while hexahydroisoaristolactone would be LVa or b without taking stereochemistry into account.







LIVa



LIVC

LIVE









LVa

LVD

-115-

Structures LIVc to LIVf inclusive for aristolactone, and with them structure LVb for hexahydroisoaristolactone, may be immediately eliminated on the basis that no rational mechanism is apparent which would account for the facts that the keto-ester, methyl oxoaristate  $is\beta, \delta$  - and  $not \alpha\beta$  - unsaturated<sup>45</sup> and that it still retains a double bond in association with the ring methyl group as shown by its n.m.r. spectrum. Moreover, the isopropenyl double bond is known not to be involved in the conversion of methyl oxoaristate under the influence of base into and  $\beta$ - unsaturated keto-acid45 since its dihydro derivative in which this function is converted into an isopropyl group on treatment with base also affords an dB- unsaturated keto-acid45.

The formation of methyl oxoaristate from LIVa or LIVb can be rationalized on the basis of parallel transannular hydride shifts as shown in LIVa going to LVI wherein the ring ethylenic link associated with the methyl group remains  $\beta_{1}$  to the ketonic carbonyl group. Such transannular interactions are not without parallel in 10-membered ring systems<sup>91,92</sup>

-116-

although such transannular hydride shifts are normally acid catalysed, not base catalysed, and are completed by an elimination, not an addition. The present mechanism, however, has certain features akin to those of  $S_N^{A'}$  reactions with the eliminated hydride ion resubstituting in the molecule across the ring this geometrically favoured substitution being offered as a driving force for the original elimination.



The alternative structure LIVb for aristolactone cannot be eliminated on the basis of the absence of diene conjugation in the ultra-violet since should the 10-membered endocyclic diene be cis-cis no true conjugation is possible as has already been shown for the cis-cis isomer of cyclodeca - 1,3-diene itself<sup>93</sup> although for the cyclodeca - 1,3-diene in which one double bond is cis and the other trans true conjugation was demonstrated<sup>93</sup>. In the case of aristolactone however the lactone bridge could so twist the molecule from the planar that electron delocalization would not be possible with respect to an endocyclic cits, trans diene. A somewhat analogous effect is observed for the "K" band of biphenyls. Biphenyl itself exhibits Mmax 249 522 ( $\xi$ ,15,000)<sup>144</sup>, while 2-methylbiphenyl (LVII), where the methyl group forces the two benzene rings slightly out of plane with resultant lowering of band position and intensity, has  $Mmax 237mp(\xi,10500)^{94}$ . In a more extreme case the two <u>ortho</u>-methyl groups of ditolyl (LVIII) force the two rings into planes at right angles to one another so that no high wave length absorption apart from benzenoid absorption appears<sup>95</sup>



# 1

LVII

LVIII

The n.m.r. spectra of none of the three compounds, aristolactone, isoaristolactone, and dihydroisoaristolactone show vinylic protons with coupling constants in the range 8-14 c.p.s., as would be

expected from adjacent protons such as on C-5 and C-6 in LIVb. Indeed the J value of 1 c.p.s. in the low field vinylic proton triplet in these compounds would point to structure LIVa for aristolactone, since either of the isolated ethylenic links therein could reasonably give rise to such a coupling pattern 71, 73, 82, 89 The placing of the double bond bearing the isopropenyl group as in LIVa and not as in LIVb can, however, be done with virtual certainty as a result of a reinvestigation of the so-called "oxoaristaldehyde" and "isooxoaristaldehyde" which are formed by treating aristolactone and isoaristolactone respectively with lithium aluminium hydride<sup>45</sup>. The n.m.r. spectra of these two compounds clearly show that they are not aldehydes as previously suggested<sup>45</sup> whilst the absence of hydroxyl absorption in the infra-red together with a carbonyl absorption band at 1765 cm<sup>-1</sup> in both derivatives indicate them to be saturated X-lactones. Their nomenclature has accordingly been changed to dihydroneoaristolactone and dihydroneoiscaristolactone respectively. This is in accordance with their lower oxidation state as indicated by the presence of n.m.r. absorption from only 3 vinylic protons

in the 4.5 to 5.5 Cregion. The formation of these two new lactones, like the formation of methyl oxoaristate is accompanied by the loss of the low field vinylic proton appearing at 3.26 Tin the n.m.r. spectrum of aristolactone. Inspection of molecular models shows that in certain of the geometrical isomers possible with structure LIVa the vinylic proton on C-8 lies over the lactone carbonyl group. The de-shielding effect of the carbonyl group on this proton could therefore account for its low field absorption at 3.267. That a new point of attachment has arisen for the acyloxy oxygen in going to the dihydroneo-lactone series (and that this point may be on C-8) is certainly not contraindicated by the n.m.r. spectra. In aristolactone, as already mentioned. the oxygen bearing methine proton gives rise to a triplet at ca 5.27 (J=8 c.p.s.) whereas in the dihydroneoaristolactone series the corresponding proton affords a doublet (J=8 c.p.s.) at ca 67 which would indicate a change in the lactone ring system.

The formation of dihydroneoaristolactone by the action of lithium aluminium hydride on aristolactone as represented by LIVa may be considered to occur by one



of schemes GaD or R, making the product one of LIX to LXI. Alternative structure LIVb for aristolactone could only give rise to dihydroneoaristolactone via a scheme analogous to 2 in order to account for the formation of a &- lactone. Scheme may be eliminated entirely and with it alternative structure LIVb for aristolactone since in addition to the mechanism involving the highly unlikely loss of a hydride ion from an aldehyde, the mechanism is completely parallel to that involved in the formation of methyl oxoaristate from aristolactone, - the attacking species being a hydride ion in one case and a methoxide ion in the other - and it is known that isoaristolactone, which undergoes an analogous reaction with lithium aluminium hydride, will not undergo keto-ester formation 45. Thus the structure of aristolactone is confirmed as LIVa, and dihydroneoaristolactone must be one of the products arising from either scheme C or D.

The carbon and hydrogen analytical values as well as the proton count in the n.m.r. spectrum of dihydroneoaristolactone indicates that it contains 22 hydrogen atoms (as is also true for dihydroneoisoaristolactone). The decalin derivative LX which would arise from the
mechanism in scheme D, can only accommodate 20 protons, and so scheme D must be ruled out and structure LXI, which fits all the available physical evidence, accepted for dihydroneoaristolactone.

When aristolactone (LIVa) is treated with neutral potassium permanganate<sup>45</sup> two products are obtained. That formed in major yield is an oily carboxylic acid containing a methyl ketone group, whilst the minor product which is obtained crystalline can be concluded to be a 1,2- diol since it forms an isopropylidene derivative C18H260 m.p. 120-121°, E 210, =103, on treatment with excess day acetone in the presence of p-toluenesulphonic acid96. That the formation of this 1,2- glycol (which was originally termed "6,7dihydroxyaristolactone" 45) does not involve the isopropenyl double bond of aristolactone was shown by the absence of formaldehyde as a product when the diol was treated with sodium meta-periodate and from the retention of vinylidene absorption at 3068 and 895 cm<sup>-1</sup> in the infra-red. Thus the 1,2- glycol can be formulated as LXII or LXIII.



LXII

IXIII

Structure LXIII would <u>a priori</u> appear to be the more likely alternative since it is reported<sup>45</sup> that the diol is stable to mineral acid. Even if a rearrangement analogous to that involved in the conversion of aristolactone into isoaristolactone (which concerns the 4:5 double bond - see page141) does not occur as it would be expected to do, diol LXII would still be expected to undergo allylic rearrangement in acid medium by a mechanism such as shown in partial structures LXIV and LXV and LXVI. Product LXV is not a 1,2- glycol and product LXVI is the same compound as would result if initial diol formation had taken place on the isopropenyl vinylidene group - known not to be the case. Hence it would appear that the crystalline product of neutral permanganate oxidation is LXIII.

-123-



-124-

Final proof of structure LXIII follows from the fact that the oily carboxylic acid formed as the major product in the permanganate oxidation of aristolactone (LIVa) possesses a methyl ketone function in accordance with structure LXVII which is readily derived from LXIII. Interestingly the infra-red spectrum of the 1,2- glycol (measured in potassium chloride disc) shows a well resolved split lactone carbonyl peak, the two component peaks being of equal intensity and occurring at 1768 cm<sup>-1</sup> and 1745 cm<sup>-1</sup>, which could well arise from hydrogen bonding in the crystal lattice.



IXVII

The present studies have shown that hexahydroisoaristolactone (LVa) when subjected to g.l.c. over 0.5% Apiezon M on acid washed celite, exhibits a main peak with a shoulder on the high retention time side. Unfortunately, however, complete resolution could not be obtained with this column at other temperatures or with a column packed with silicone on celite and further work employing other columns is necessary in order to achieve complete separation of the isomers. It is nevertheless clear that two stereoisomers are present in hexahydroisoaristolactone (LVa), thus confirming previous evidence wherein hydrolysis of hexahydroisoaristolactone afforded both a crystalline and an oily hydroxy acid - the latter being present in only small amounts<sup>45</sup>. The occurrence of two isomers in hexahydroisoaristolactone can be rationalized if stereospecific reduction of one double bond and non-stereospecific reduction of the other were to occur during its formation from dihydroisoaristolactone. The stereoisomerism would be expected at C-4 since inspection of models of various geometrical isomers fitting formula LIVa indicates that the high steric impedence of the lactone ring on one side of the 7:8 double bond would lead to stereospecific

-125-

reduction at this site. The formation of two germacrenes and one germacrane from the action of lithium aluminium hydride on the two diastereoisomeric dimethanesulphonates prepared from the mixed hexahydroisoaristo - 2,14- diols (LXVIII) can be explained by invoking a transannular hydride shift and elimination of the elements of methanesulphonic acid as an alternative to hydrogenolysis giving rise to germacrenes LXIX. Analogous eliminations during complex metal hydride reductions have been encountered in the steroid field<sup>97</sup> and with phenyl substituted butanols<sup>98</sup>.



Should one of the diastereoisomeric dimethanesulphonates give rise solclyto one germacrene isomer whilst the other gives rise to the second germacrene and the germacrane, the occurrence of three hydrocarbons in the lithium aluminium hydride reduction product as proven by gasliquid chromatography would be completely accounted

-126-

for in terms of the results presented in table 2.

The elements of symmetry present in the germacrane molecule reduces the number of possible isomers from the usual  $2^n$ , where n = the number of asymmetric carbon atoms. Indeed examination of models indicates that the germacrane structure can exist as a total of two non-optically active stereoisomers LXX and LXXII and one pair of enantiomorphs LXXII and LXXIII. The latter would of course not be resolved by gas-liquid chromatography and so synthetic germacrane should exhibit three peaks on g.l.c., as was indeed observed.



XX

LOUI

IXII



IXXIII

Complete reduction of the mixture of the two germacrenes and the germacrane resulting from the action of lithium aluminium hydride on the mixed diastereoisomeric methanesulphonates prepared from the mixed hexahydroisoaristo- 2,14- diols would then give germacranes in which stereoisomerism was possible on both C-4 and C-7 - the stereochemistry at C-10 being uniquely defined as that in aristolactone. Stereoisomerism at both C-4 and C-7 would, however, give rise to three germacrane isomers viz LXX, LXXI and one of the two enantiomorphs LXXII or LXXIII, thus accounting for the observed composition of the germacranes derived from aristolactone (LIVa). Moreover as can be seen from the results summarized in table 2, the germacrene with lowest retention time gives rise to the germacrane with the lowest retention time. whilst the germacrene with the highest retention time gives rise to the two germacranes having the higher retention times.

So far nothing has been said concerning the possibilities of geometrical isomerism about the endocyclic double bonds of the aristolactone series.

-128-

However important evidence concerning the configurations of these double bonds is perhaps available from the differences in behaviour of methyl oxoaristate and its dihydro-derivative towards the action of refluxing acetic anhydride, as discovered in the present work. Thus methyl oxoaristate on treatment with boiling acetic anhydride affords a neutral crystalline compound of m.p. 112-113°,  $[\alpha]_{D}$  +164° (EtOH) analysing for  $C_{15}H_{20}O_{2}$ . This compound showed weak end absorption in the ultra-violet (8,950 at 210 mp) and a low intensity maximum at 275 mp (E, 30). Catalytic hydrogenation of this derivative (hydrogen uptake: 2 moles) afforded the fully saturated tetrahydro-derivative, m.p. 70-72°. The infra-red spectrum of the unsaturated compound, in dilute carbon tetrachloride solution, was devoid of hydroxyl absorption and exhibited a split carbonyl peak of near equal intensities at 1788 and 1779 cm<sup>-1</sup> (in potassium chloride disc the lower frequency band is only barely apparent) which would indicate the presence of a saturated 8-lactone function 75c and the absence of keto or ester functions. Since the spectrum was measured in dilute carbon tetrachloride solution (4 mg /ml) the split carbonyl band cannot have its origin in

-129-

intermolecular hydrogen bonding of the type demonstrated in LXXIV or in dipole interactions of the type shown in LXXIV.

The carbonyl doublet would not be expected to arise from the presence of configurational isomers since any lactone formation by the action of acetic anhydride on methyl oxoaristate could reasonably be expected to be stereospecific. However, the split carbonyl peak could be expected to arise from either Fermi resonance (the coupling between the overtone of a low frequency absorption with the fundamental stretching frequency of the lactone carbonyl or from a "hot transition" (wherein a low frequency vibration is excited from both the ground state and an upper energy level)99. In so far as Fermi resonance is solvent dependent and concentration independent, the carbonyl absorption of the product obtained from the action of acetic anhydride on methyl oxoaristate, was studied in tetrachloroethylene solution and in acetonitrile solution at different concentrations. The results

-130-

clearly indicated that the double carbonyl peak does indeed arise from Fermi resonance. Thus in place of the two peaks at 1788 and 1779 cm<sup>-1</sup> in carbon tetrachloride solution, the spectrum measured in tetrachloroethylene solution (0.26 mg/70 µl; 0.49 mm microcell) showed a high intensity peak at 1791 cm<sup>-1</sup> with a shoulder on the low intensity side at 1782 cm<sup>-1</sup> whilst the spectrum run in acetonitrile solution showed a high intensity peak at 1772 cm<sup>-1</sup> with a shoulder on the high frequency side at 1783 cm<sup>-1</sup> - the relative intensities being independent of concentration. The small frequency shifts observed are in accord with other studies of Fermi resonance<sup>100</sup>.

The n.m.r. spectrum of the unsaturated product from the action of acetic anhydride on methyl oxoaristate shows the presence of 20 protons so it can be concluded on the basis of double bond equivalents ( $C_nH_{2n+2}$ ;  $\frac{32-20}{2} = 6$ ) two being present as double bonds as evidenced by quantitative hydrogenation and absence of unsaturation in the product as shown spectroscopically; a third being utilized in lactone ring formation, with a fourth being present in the lactone carbonyl) that the

derivative is bicyclic in addition to the lactone ring. The infra-red spectrum of this compound exhibited in addition to the typical vinylidene C-H stretching band of aristolactone and its derivatives at 3073 cm<sup>-1</sup> a new band at 3091 cm<sup>-1</sup> which may be assigned to a second vinylidene group 72,75a76,77, the C-H stretching of other ethylenic groups being known to absorb in the range 3055-3010 cm<sup>-1</sup> 75a,102. The presence of two vinylidene groups would appear to be confirmed by the n.m.r. spectrum which exhibits 4 vinylic protons in the range 5.1 to 5.37. The absence of absorption in the range 8.4 to 8.5 Ttypical of the trisubstituted double bond methyl group on C-4 in other aristolactone derivatives, and the absence of any absorption from protons of a methyl group bound to a carbon atom bearing hydrogen would further support the conclusion that the C-4 methyl group had undergone isomerization to a vinylidene group. At the same time a barely resolved triplet (J=0.5 c.p.s.) of intensity 3 protons at 8.27 [Comparable to the analogous absorptions from the methyl group of the isopropenyl function at 8.33 7 in methyl oxoaristate (LVI), at

8.17 (in aristolactone (LIVa) and at 8.14 (in isoaristolactone, would indicate that the isopropenyl function was intact in the new lactone. Further, the absence of resonance attributable to a proton on an oxygen bearing carbon atom would indicate that the lactone acyloxy oxygen is bound to a tertiary carbon atom. In summation then, the action of acetic anhydride on methyl oxoaristate (LVI) affords a bicyclic compound possessing in addition a &- lactone ring derived from a tertiary hydroxyl group, and two vinylidene groups.

The dihydro-compound obtained by catalytic hydrogenation of methyl oxoaristate (LVI), and which still retains a trisubstituted double bond in association with the methyl group on C-4, on treatment with boiling acetic anhydride yields a neutral crystalline product, m.p. 166-167°,  $[\alpha]_p-74°$  (EtOH) analysing for  $C_{17}H_{26}O_4$ , and exhibiting max 215 mm (E,154) and 254 mm (E,69). The analytical figures for this compound thus indicate a net replacement of methoxide ion in the starting material by acetoxy ion, but the infra-red spectrum (measured in potassium chloride disc with a Perkin-Elmer Infracord) clearly shows that it is not an anhydride by the absence of peaks in the 1850-1800 cm<sup>-1</sup> region. Moreover there is no generation of a vinylidene group corresponding to that observed in the case of methyl oxoaristate as evidenced by the absence of absorption at <u>cs</u> 3075 cm<sup>-1</sup> and 990 cm<sup>-1</sup>. Further, no absorptions ascribable to a hydroxyl group are present but the infra-red spectrum exhibits peaks at 1768 cm<sup>-1</sup> and 1735 cm<sup>-1</sup>. The first of these is assignable to a saturated  $\check{A}$ -lactone function<sup>75c</sup> whilst the second when taken in conjunction with a C-0 stretching mode at 1245 cm<sup>-1</sup> can be assigned to an acetate group.

The generation of such an acetoxy lactone from a J-koto ester by the action of acetic anhydride would appear to have its closest analogy in the formation of enol lactones by the action of acetic anhydride on keto acids. Woodward<sup>103</sup> has suggested that the latter reaction involves attack by an acetoxy ion on a proton attached to a carbon atom  $\emptyset$ - to the keto group with concerted elimination of acetoxy ion from the mixed anhydride which is first formed, as is shown in the conversion of partial structure LXXVI into LXXVII.





At the same time he expected the compound represented by partial structure LXXVIII to be formed as a second product on the grounds of alternative attack by acetoxy ion on the carbon atom of the carbonyl group  $10^4$  although in actual fact none of this substance could be detected in his reaction product.



If however in the case of the dihydro derivative of methylexceristate(LOCIX), attack on the carbonyl carbon atom were to occur in this very manner satisfactory account of the properties of the product can be given as shown in the conversion of LXXIX into LXXX.



Further evidence in support of a structure such as LXXX for this compound comes from the nature of the product formed by the action of glacial acetic acid on the dihydro derivative of methyl oxoaristate, which in turn under the influence of boiling acetic anhydride is converted into the same compound as represented by LXXX (mixed m.p., infra-red spectrum and optical rotation). When treated with glacial acetic acid at room temperature for 48 hours the dihydro derivative of methyl oxoaristate afforded a neutral crystalline compound C18H3005, m.p. 90.5-91°, [x]\_- 73.91° (EtOH) and E,600 at 210 mp. The infra-red spectrum (in potassium chloride disc measured on the Perkin-Fimer Infracord) indicated hydroxyl O-H stretching at 3400 cm<sup>-1</sup> and C-O stretching at 1130 cm<sup>-1</sup>, indicative of the presence of a tertiary hydroxyl group 75d.

-136-

A carbonyl band at 1720 cm<sup>-1</sup> is assignable to the methyl ester, while a second band at 1705 cm<sup>-1</sup> when taken in conjunction with a C-O stretching mode at 1260 cm<sup>-1</sup> would indicate the presence of an acetate group<sup>75d</sup> - that is the elements of acetic acid appear to have been added to the starting compound. This can be rationalized by formation of the monoacetate of the gem diol derived from the keto carbonyl group (LXXIX -> LXXXI). The inductive effect of the hydroxyl group in LXXXI would be expected to lower the infra-red absorption frequency of the acetate carbonyl<sup>85</sup> in agreement with the observed facts.



IXXIX An interesting reaction which may be taken for an analogy is given by the formation of LXXXIV by the action of lactic acid(LXXXII) on cyclohexanone (LXXXIII) in the presence of an acid catalyst. This cyclic ketal (LXXXIV) may be envisaged as arising by some such mechanism as shown below.



## LXXXII

LXXXIV

An alternative mechanism for the net addition of the elements of acetic acid to the oxo group of the dihydro derivative of methyl oxoaristate (LXXIX) involving enolization and a double bond shift is portrayed in the sequence LXXIX LXXXV LXXXVI.



IXXIX IXXVI Such nucleophilic attack by the acetoxy ion on the enol carbon atom is not unlike the hemiacetal acetate formation found with the dihydrofuran ring in clerodin (LXXXVII going to partial structure LXXXVIII)<sup>105</sup>, or the addition of carboxylic acids to dihydropyran (LXXXIX) giving rise to tetrahydropyranyl -2- esters (XC)<sup>106</sup>.



H-G-G-GH3

TXXXAII

IXXXVIII

LXXXIX

As is true of such tetrahydropyranyl esters, the hydroxy acetate derived from the dihydro derivative of methyl oxoaristate (LXXIX) was found to be unstable to acids, dilute acetic acid itself causing decomposition. Until degradation studies have been carried out on the hydroxy acetate however, distinction between structures LXXXI and LXXXVI cannot be made, although the latter would be expected to readily generate and g-unsaturated ketone through loss of the elements of acetic acid. Nevertheless either structure on treatment with acetic anhydride would be capable of losing the elements of methanol to give a lactone acetate of the type represented by LXXX.

The formation of a new carbocyclic ring by methyl oxoaristate under the influence of acetic anhydride whilst the same reagent gives rise to a reaction of a completely different type in the case of the dihydro derivative can only have its explanation in differences in the nature of the 4:5 double bond in the two compounds. Certainly direct attack on either the oxo or carbomethoxy groups would not take such different courses without an underlying reason.

-140-

The presence of an isopropenyl group in methyl oxoaristate (LVI) and its absence in the dihydro compound (clearly shown by infra-red and n.m.r. spectroscopy) can in no way affect the issue since the isopropenyl group is retained unchanged in the product formed from methyl oxoaristate as evidenced by infra-red and n.m.r. studies. Moreover the double bond must be 4:5 in methyl oxoaristate and in its dihydro compound since both compounds are not  $\alpha\beta$ -unsaturated ketones and both are isomerised to  $\alpha\beta$ - unsaturated ketones by base 45, 47.

Since the dihydro derivative of methyl oxoaristate is prepared under conditions strictly analogous to those known to convert aristolactone into dihydroisoaristolactone, it is to be concluded that an identical isomerization is occurring in both cases and that "methyl dihydroixoaristate" is more properly to be termed methyl dihydroisooxoaristate. In the original preliminary communication announcing the revised structure of aristolactone<sup>107</sup> (a reprint of which is included as appendix 2), it was assumed that the conversion of the aristolactone series into



the isoaristolactone series involved migration of the 4:5 double bond into the 3:4 position in the absence of other evidence. However it is now clear that the iso series must differ from the original series through geometrical isomerism about the 4:5 double bond although unambiguous assignment of configuration to this double bond in the aristolactone and isoaristolactone series cannot be made until the actual bicyclic skeleton present in the product of the action of acetic anhydride on methyl oxoaristate has been determined. Should the ring closure to the bicyclic compound involve loss of a proton from C-6, C-7, or C-8 (scheme F), inspection of models shows that a transannular reaction is only possible with the 4:5 - double bond in the cis configuration and so it would be cis in methyl oxoaristate (LVI), aristolactone (LIVa) and dihydroneoaristolactone (LIX), and trans in methyl dihydroisooxoaristate (EXXIX), isoaristolactone, dihydroisoaristolactone and dihydroneoisoaristolactone. However. should the generation of the bicyclic ring system involve attack by the 11 electrons of the 4:5 double bond on the carbonyl carbon atom as shown in scheme & (an attractive mechanism in so far as it involves spontaneous

-142-



scheme G

generation of the second vinylidene group, inspection of models would show that no distinction can be made between a cis double bond or a trans double bond in the 4.5 position of methyl oxoaristate, and so the configurations as listed above could be reversed.

Should scheme ? be operative, in view of the known greater stability of double bonds exocyclic to a 5-membered ring<sup>108</sup> it might be considered that structure XCII is more likely than XCI or XCIII. Should the compound prove to be XCI, the generation of the exocyclic vinylidene group in this decalin derivative might possibly be rationalized by analogy with the direction of enolization in 3-keto steroids. In the case of these latter compounds, where the A/B ring fusion is trans (XCV) enolization generates the double bond in the 2:3 position<sup>109</sup> as shown in XCVI. whilst where the A/B ring fusion is cis, enolization generates the double bond in the 3:4 position (XCVII -> XCVIII)<sup>109</sup>. The same phenomenon is also evidenced by the 2-bromination of 3-oxo trans A/B steroids and the 4-bromination of 3-oxo cis A/B steroids110

-143-



XCV

XCVI



## XCVII

## XCVIII

Thus should the intermediate in pathway 6 of scheme F be a cis fused decalin, an unfavourable configuration would result<sup>111</sup> with the double bond being endocyclic, so that allylic rearrangement to the exocyclic position might not be unexpected.

Although J values are of little assistance in distinguishing between the two possibilities, an

indication that the 4:5 double bond may be trans in aristolactone and cis in isoaristolactone (i.e. the mechanism shown in scheme G is involved in the action of acetic anhydride on methyl oxoaristate) follows from the positions of the resonance from the 3 protons of the methyl group on C-4 in aristolactone and isoaristolactone (at 8.527 and 8.417 respectively) since Bates and Gale<sup>112</sup> have shown that the resonance from the protons of the methyl group substituted to a cis ring double bond in costunolide (XCIX) absorb some 0.07? lower than the corresponding methyl group protons in germacrone (C) where the methyl group is substituted on a trans double bond. Other examples with acyclic compounds and caryophyllene derivatives further support the generalization that protons of a methyl group on a cis double bond absorb at a lower field than protons of methyl groups on trans double bonds. However the values of the C-4 methyl proton absorption at 8.487 in methyl oxoaristate and at 8.52 (in methyl dihydroisooxoaristate show a shift in the opposite direction to that with aristolactone Unfortunately superposition of and isoaristolactone. both methyl absorption peaks in the n.m.r. spectrum

-145-

of dihydroneoaristolactone prevents the acquisition of further evidence from the neolactone series. Thus no definite assignment of double bond configuration to the aristolactone and isoaristolactone series can be made with any degree of certainty from consideration of the position of the absorption of the protons on the C-4 methyl group.



The inability of isoaristolactone to undergo the transannular hydride shift involved in the formation of methyl oxoaristate (LVI) from aristolactone (LIVa) can of course be explained by the changes produced in the geometry of the 10-membered ring as a result of the isomerization of the 4:5 double bond, whatever the configurations in the two compounds.

Unambiguous assignment of configuration to the 7.8 double bond in aristolactone (LIVa) and the various derivatives in which it occurs, is again not possible since examination of molecular models

-146-

ŧ

shows that whether this 7.8 double bond be cis or trans, certain conformations are possible with the 4.5 double bond cis or trans in both series (when variation in the fusion of the lactone ring is also considered) which are relatively strain free and permit of the transannular hydride shift involved in the formation of methyl oxoaristate (LVI), and the deshielding of the C-8 proton by the lactone carbonyl group. However, it would appear that there is greater restriction to true conjugation between the 7:8 double bond and the isopropenyl double bond where the former is trans than where it is cis. It may be noted that models can be made where both the 4:5 and 7:8 double bonds have the trans configuration.

Incidentally, the assignment of new formulae to the aristolactone series as reported in this thesis necessitates a complete re-interpretation of the ultraviolet spectral comparisons made earlier by Steele, Stenlake, and Williams<sup>113</sup> but this will not be attempted here.

From the above it can be seen that in order to complete the aristolactone problem, the following need to be done:-

Conclusively establish the constitution of the diol 1. formed by potassium permanganate oxidation of aristolactone. Establish the carbon skeleton of the neoaristolactone 2. series. This may be determined by first catalytically reducing the neo lactones to their corresponding fully saturated compounds and then comparing this product with hexahydroisoaristolactone (although because of the possibility of stereoisomerism non identity of the products with hexahydroisoaristolactone does not mean that they necessarily have the new lactone system, as portrayed in LIX). By treating these fully saturated compounds with lithium aluminium hydride, the corresponding diol would be formed which after mesylation and lithium aluminium hydride hydrogenolysis (and if necessary catalytic hydrogenation as employed in the conversion of hexahydroisoaristolactone into germacranes) would yield the parent hydrocarbons which could then be identified.

3. Isolate once more and characterise the second product from the lithium aluminium hydride reduction of aristolactone i.e. the compound previously termed "oxoaristool"45.

The infra-red spectrum in carbon tetrachloride

-148-

solution of a specimen of this compound prepared by earlier workers on the aristolactone problem showed carbonyl absorption at 1759 cm<sup>-1</sup> which can only be assigned to an ester or lactone carbonyl function, and two hydroxyl stretching peaks at 3605 cm-1 (non hydrogen bonded) and 3544 cm<sup>-1</sup> (hydrogen bonded). Since it is impossible to have both an ester function and a hydroxyl function in this compound which analyses for (C15H2LO2), 45 when n=1, it is apparent that the compound must be polymeric - the most likely case is that it is dimeric (i.e. n=2) and that it possesses one ester function and two hydroxyl groups in agreement with the double hydroxyl absorption in the infra-red. Moreover the relative intensity of the C=O stretching absorption as compared with those of the C - O stretching absorptions in the 1100-100) cm-1 region would further support this interpretation. Its dimeric nature would also be indicated by its melting point of 245-246° which is considerably higher than the melting points of the other known aristolactone The n.m.r. spectrum does not permit derivatives. an accurate proton count, but it is not inconsistent

-149-

with the presence of 44,46 or 48 protons of which 12 are present as methyl groups attached to vinylic positions (superposed absorption at 8.42) and of which 6 or 7 are present as vinylic protons (complex absorption in the 5 (region). At least 2 protons (and perhaps 3) attached to carbon atoms bearing oxygen are present as indicated by complex absorption at <u>ca</u> 6 (.

Lack of material prevented further study of this compound and until more is available it would seem unwise to engage in predictions of its structure. It would appear however, that its formation does not involve a simple partial reduction of lactone groups to afford hemiacetals as is observed in the picrotoxinin series<sup>114</sup>. 4. Establish the carbon skeleton of the product of the action of acetic anhydride on methyl oxoaristate employing similar methods to that described in point 2. Should this compound prove to be XCII it is to be noted that selenium dehydrogenation would be expected to give 2,5-dimethyl-8-isopropylazulene (which would not appear to be so far known).

-150-

5. Establish the carbon skeleton of products arising from the action of glacial acetic acid and acetic anhydride on methyl dihydroisooxoaristate, employing similar methods.

6. Completely resolve hexahydroisoaristolactone by gas-liquid chromatography and determine the percentage of each isomer present as a check that they are in the ratio of 7:3 as calculated from the results shown in table 2.

7. All these compounds should be subjected to n.m.r. spectroscopy as well as infra-red spectroscopy in dilute carbon tetrachloride solution on the Unicam model S.P.100, where such work has not been previously performed.

8. It might also be profitable to convert methyl oxoaristate through into germacrane as a double check on its carbon skeleton.

To fully establish the stereochemistry of aristolactone and its derivatives (both with respect to the geometry of the double bonds and with respect to the fusion of the lactone ring) it would appear that the best available method involves an X-ray crystallographic study of certain

-151-

key compounds as heavy atom derivatives. For instance the presence of 3 double bonds in aristolactone and isoaristolactone might conceivably lead to crystalline silver nitrate adducts<sup>115</sup>. Again the addition of the elements of acetic acid to methyl dihydroisooxoaristate to afford a crystallineproduct would suggest that the corresponding compound prepared from bromoacetic acid might well be a suitable derivative for X-ray study. Other suitable derivatives could conceivably be prepared by the addition of such electrophilic reagents as nitrosyl bromide, hydrogen bromide, or iodine monochloride to a variety of unsaturated aristolactone derivatives available.

In addition to the work on aristolactone certain preliminary screenings were also performed in the present work on two Aristolochia species previously studied by other workers<sup>23,24,28</sup> <u>viz A. cymbifera</u> (syn.A. grandiflora) and A. bracteata, to ascertain whether these species, as well as A. reticulata and A. serpentaria, contained aristolactone. It was shown that they did not. However certain other constituents were isolated. From A. bracteata,

-1520

aristolochic acid (VI) was obtained in 0.008% yield whilst A. cymbifera afforded what appeared to be A-sitosterol (DK) on the basis of mixed melting point and infra-red spectra (measured as the potassium chloride disc). However as sitosterols nearly always occur in admixture in nature116 it is necessary to carry out further work for example, conversion into the methyl ether and then gas-liquid chromatography as described by Clayton<sup>117</sup> before identification of this material can be made with certainty, Also some preliminary work was done on the "water insoluble acid" previously reported by Stenlake and Williams . This was shown to be dibasic and to fit the formula C20H2804, by molecular weight determinations employing boiling point elevation measurements. Kuhn-Roth determination indicated that the compound contained 3 C-CH<sub>3</sub> groups. Generation of the dimethyl ester by treatment of the acid with diazomethane gave a compound analysing for C22H32O4. Preliminary ozonolysis studies on both the dibasic acid and its dimethyl ester gave indistinct and variable results so that no conclusions as to the nature of the diacid can be made. Further work on

this compound should initially be directed towards confirming its chemical individuality by employing gasliquid chromatography of the dimethyl ester.

Melting points were determined on a hot-stage melting point apparatus and are uncorrected. Ultra-violet spectra were measured on an Optica CF-4 recording spectrophotometer and on a Hilger and Watts U-V spek. Infra-red spectra were measured on an Unicam model S.P 100 instrument as described in the discussion, or on Perkin-Elmer 237 or Infracord instruments. Optical rotations were determined on a Bellingham and Stanley polarimeter employing a 1 decimetre cell. Gas-liquid chromatographic analyses were carried out on a Perkin-Elmer Fractometer run at 138° (employing a 50 metre capillary column coated with polypropylene glycol as stationary phase). Microanalyses were carried out by the microanalytical laboratory of the Royal College of Science and Technology and by Drs. Weiler and Strauss, Oxford. Catalytic hydrogenations were performed in a Towers microhydrogenation apparatus at atmospheric pressure and room temperature. Horizontal tube distillations were carried out in a Towers heating unit.

Isoaristolactone, dihydroisoaristolactone and hexahydroisoaristolactone were prepared as previously
described in the literature<sup>47</sup>, as were dihydroneoaristolactone ("oxoaristeldehyde") and dihydroneoisoaristolactone (isooxoaristaldehyde")<sup>45</sup>.

Hexahydroisoaristo-2.14-dimethanesulphonate (LXVIII)

To hexahydroisoaristo-2,14-diol, m.p. 106-107° (formerly termed "tetrahydroisoaristo-6,12-diol") prepared as previously described<sup>45</sup> (1.00g; 3.66 m mole) in dry pyridine (10 ml) at 0°, was added dropwise methanesulphonyl chloride (1.15g: 8.05 m mole) and the reaction mixture allowed to stand at room temperature for 36 hr. Addition of crushed ice precipitated crystals which were collected by filtration and dried. Recrystallization from ether/petroleum ether afforded colourless needles of the dimethanesulphonate m.p.79-81° (0.90g), which quickly suffered a pronounced fall in melting point (Found: C, 51.19; H, 7.62, 8.15.45. C17H3406S2 requires C, 51.23; H, 8.60; S, 1609%). Ymax ( inKCl disc) 1330 and 1175 cm<sup>-1</sup> (sulphonate). Conversion of the Dimethanesulphonate into

Diastereoisomeric Germacranes (XLV)

The dimethanesulphonate prepared as described above (0.85g) in dry ether (50 ml) was treated with excess of

lithium aluminium hydride (0.8g) and the mixture heated under reflux for 6 hours. Excess of complex metal hydride was decomposed by the slow addition of dil. HCl to the reaction mixture. The resultant acidic solution was exhaustively extracted with ether and the ethereal solution washed with water and dried over anhydrous sodium sulphate. Concentration of the ethereal solution afforded a colourless oil (0.321g) a sample of which gave a strong positive reaction to tetranitromethane indicating the presence of unsaturated material (confirmed by a band at ca 1660 cm<sup>-1</sup> in the infra-red spectrum). G.l.c. (see table 2) indicated that the oil was composed of three compounds. The unfractionated oil (0.32g) in ethanol (11 ml) was then hydrogenated over platinum oxide (0.08g) until hydrogen uptake ceased (total uptake: 0.49 mole hydrogen). The catalyst was removed by filtration and after concentration of the filtrate the oily residue remaining was distilled in a horizontal tube at 90°/0.05 mm Hg, the entire distillate being collected as one fraction. Since the infra-red spectrum and the g.l.c. trace (see table 2) indicated that unsaturated material was still present, the crude product was subjected to rehydrogenation this

time employing highly active Adam's catalyst (uptake: 0.39 mole hydrogen). Distillation of the fully saturated oily product at 90°/0.05 mm Hg afforded a mixture of three components as shown by g.l.c. The three peaks present were each found to correspond in retention time to one of the three peaks obtained from a sample of synthetic germacrane (confirmed after admixture of the two samples).

### Aristolactone diol Acetonide.

Aristolactone diol (formerly termed 6,7-dihydroxyaristolactone") prepared by neutral potassium permanganate oxidation of aristolactone as previously described<sup>45</sup> (0.090g) in dry acetone - benzene (50 ml to 20 ml) to which <u>p</u>-toluenesulphonic acid (100 mg) had been added, was heated under reflux for 20 hours following a procedure developed by Takeda, Kubota and Shimaoka<sup>96</sup> for the preparation of the acetonides of sapogenins. The reaction mixture was neutralized by 15% sodium bicarbonate solution and the organic solvents were removed by distillation under reduced pressure. The aqueous residue was exhaustively extracted with benzene, the benzene solution washed with water, dried (NatiSO<sub>4</sub>), and concentrated affording crystalline material

~158~

m.p. 126-128° (0.1g.) Chromatography over ferric oxide<sup>118</sup> with chloroform as eluant gave nearly colourless crystals which upon recrystallization from methanol exhibited m.p. 128-129° Subsequent recrystallizations from methanol induced a fall in m.p. to 120-122° which could not be elevated by repeated recrystallization or sublimation. The <u>acetonide</u> melting at 120-122° was subjected to analysis (Found: C, 69.69; H, 8.85. C<sub>18</sub>H<sub>36</sub>O<sub>4</sub> requires C, 70.55; H, 8.85%) £103 at 210 mµ. Formation of the Bicyclic Derivative of Methyl oxoaristate Under the Influence of Acetic Anhydride.

Methyl oxoaristate prepared from aristolactone by the action of methanolic potassium hydroxide as previously described  $^{47}$  (0.12g) in acetic anhydride (10 ml) was heated under reflux for 2 hours. The yellowish crystalline mass (0.11g) remaining after distillation of the acetic anhydride under reduced pressure was recrystallized from aqueous acetone affording the <u>cvclization product</u> as colourless needles m.p. 112-113°,  $[\alpha]_D^{18.5}$ +164.1° (c,1.02 in ethanol) (Found: C,76.99, 77.07; H,8.40, 8.67. C<sub>15</sub>H<sub>20</sub>°<sub>2</sub> requires C,77.54; H,8.68% ymax (in C CI4 solution) 3091 and 3073 (vinylidene)
and 1788 and 1778 (&-lactone) 1656 cm<sup>-1</sup> (vinylidene).

Catalytic reduction of the cyclized product in ethanol over Adam's catalyst (uptake: 1.8 moles hydrogen) afforded the tetrahydro derivative m.p. 70-72° shown spectroscopically to be fully saturated. Lactone Acetate (LXXX) Derived from Methyl dihydroisooxoaristate.

Methyl dihydroisooxoaristate previously termed methyl oxoaristate prepared as in the literature<sup>45</sup> (0.97g.) in acetic anhydride (20 ml) was heated under reflux for 4.5 hr. after which time the acetic anhydride was removed in vacuo. The crystalline residue was recrystallized from petroleum ether (40-60°) giving colourless needles of <u>lactone acetate</u> m.p. 166-167°,  $[\alpha]_D^{25} - 63$ °(c 1.23 in ethanol) (0.31g.) (Found: 69.39, 68.84; H, 8.95, 8.80. C<sub>17</sub>H<sub>26</sub>O<sub>4</sub> requires C,69.36; H,8.90%). /max 215 mµ (£,154.) /max (ACELdisc) 1765 (¥-lactone) 1725 (acetate) and 1248 cm<sup>-1</sup> (acetate).

Carbon and hydrogen values were highly variable duplicate analyses of the same sample giving different values. Other analyses were: C, 68.32, 67.90, 68.15, 68.23 H, 8.87, 9.13, 8.74, 8.87%

Hydroxy Acetate Derivative (LXXX or LXXXVI) of Methyl dihydroisooxoaristate.

Methyl dihydroisooxoaristate<sup>45</sup> (0.68g) containing traces of colloidal platinum oxide) in glacial acotic acid (40 ml) was allowed to stand at room temperature until the fall in optical rotation had ceased (48 hr.). Careful addition of water precipitated the product which was collected by filtration, washed with water and dried under vacuum. The <u>hydroxy acetate derivative</u> m.p.80-82 was recrystallized from aqueous ethanol to constant m.p. 90.5-91°,  $[29_D^{18} - 73.91$  (c,1.0 in ethanol) (Founds C,65.82, H,9.21.  $C_{18}H_{30}O_5$  requires C, 66.24; H,9.24%) E,239 at 210 mµ max (KCldisc) 3350 (hydroxy1) 1726 (methyl ester) 1703 and 1260 (acetate) and 1135 cm<sup>-1</sup> (tertiary alcohol).

The hydroxy acetate when treated with refluxing acetic anhydride was converted into the same lactone acetate derivative of methyl dihydroisooxoaristate as described previously as shown by mixed m.p., optical rotation and infra-red spectra. Isolation of B- Sitosterol From Aerial Parts of A. cymbifera.

Dry powdered aerial parts of A. cymbifera (500g.) were percolated with petroleum ether (40-60°) until the percolate was colourless (total volume of menstruum, 31). Concentration of the solution yielded a heavy green oil (3.2g), which after seeding with aristolactone and storing in a refrigerator for an extended period of time, failed to deposit crystalline aristolactone as was always observed with corresponding extractives of A.serbentaria and A. reticulata. The oil was chromatographed over a ferric oxide118: cellulose (1:2.5) column eluting with petroleum ether (40-60°). Concentration of the various fractions obtained showed that later fractions contained a crystalline material. This was collected and recrystallized from petroleum ether (40-60°) until colourless and then sublimbed. The material showed m.p. and mixed m.p. 138-140° with B-sitosterol. The . substance gave a positive Liebermann-Burchard test. Isolation of Aristolochic Acid from A. bracteata.

Powdered dried A. bracteata (270g) was percolated with petroleum ether (40-60°) until the menstruum was colourless (total volume: 2 1) Removal of the solvent and seeding the residue with aristolactone failed to afford any of this lactone. The marc was air dried and then percolated with ethanol after maceration for 24 hr. Concentration of the percolate (IL) afforded a dark green oil (4.4g) which was treated with hot water. Trituration of the water-insoluble material with ethanol dissolved the oily matter present and left out of solution tacky crystals. Recrystallization to constant m.p. 276-278° gave aristolochic acid (0.1g) as bright yellow needles, identical with an authentic sample (mixed m.p., U-V., 1-R.).

## "Water-Insoluble Acid"

Water insoluble acid<sup>38</sup> obtained as an oil (0.3g) was distilled six times in a horizontal tube at 105°/0.05 mm Hg affording a <u>pale vellow oil</u> (Found: C, 71.80, 72.00; H,8.87, 8.96; molecular weight 303; C-CH<sub>3</sub> 10.05. C<sub>20</sub>H<sub>30</sub>O<sub>4</sub> requires C,71.85; H, 9.04%; 334.44; equivalent to 2.23 C-CH<sub>3</sub>). V max (liquid film) 1700 (carboxylic acid) 1630 (shoulder) and 940 cm<sup>-1</sup> (broad; carboxylic acid). Esterification of "Water Insoluble Acid"

Water insoluble acid (1.2g) in ether (15 ml) was

treated with excess of ethereal diazomethans. Concentration of the ethereal solution and distilling the resulting oil in a horizontal tube at 90°C/0.05 mm Hg afforded the <u>dimethyl ester</u> (Found, C,73.18; H,9.17. C<sub>22</sub>H<sub>34</sub>O<sub>4</sub> requires C,72.90; H,9.45%).

Ymax (liquid film) 1715 (ester) and 1630 cm<sup>-1</sup> (double bond).

REFERENCES

- 1. Trease, <u>A Textbook of Pharmacognosy</u>, 7th ed., Bailliere, Tindall and Cox, London, 1957, p.221.
- 2. Dawson, Pharm.J., 1927, 119, 396.
- 3. Dawson, Pharm. J., 1927, 119 427.
- Urdang, Goldat, Queller and Sonnedecker,
   <u>An Examination of Old Literature (Especially</u> <u>HERBALS) For Drugs with Supposed Effects on</u> <u>Cancer.</u> Report to The National Institutes of Health, On Contract No. C-2089, with the University of Wisconsin, 1956.
- 5. Porter-Smith, <u>Contribution Towards the Materia</u> Medica and Natural History of China, American Presbyterian Mission Press, Shanghai, 1871 p. 22.
- 6. Fluckiger, <u>Pharmacognostic</u>. (Through Ber. <u>Dtsch</u>. <u>Pharm. Ges.</u>, 1920, <u>30</u> 43).
- 7. Murray and Apparat, <u>Medicaminium</u>, 1759, p. 563 (Through Bar. Dtsch. Pharm. Ges., 1920, 30, 43).
- 8. Miller, <u>Gartnerlexikon</u>, 1759, p. 151 (Through Ber. Dtsch. Pharm. Ges., 1920, 30 43).
- 9. Rosenmund and Reichstein, Pharm. Acta Helv., 1943, 18 243.

-165-

- 10. United States Dispensatory, 24th ed., Lippincott and Company, Philadelphia, 1949.
- 11. Bosmann, Arch. exp. Path. Pharmak., 1942, 200, 414.
- 12. Shaw, Aust. J. Pharm., 1947, 28 857.
- 13. Tomita and Kura, J. Pharm. Soc. Japan. 1957, 77, 812.
- 14. Tomita and Kugo, Pharm. Bull. (Japan), 1956, 4, 121.
- 15. Pilarczyk, Planta Medica, 1958, 6 258.
- 16. Pailer and Pruckmayr, Monat., 1959, 90. 145.
- 17. Mesa, Garcia, Cravioto, and Calvo de la Torre, <u>Cencia e Invest.</u>, Buenos Aires, 1950, <u>6</u>, 471 (Through <u>Chem. Abs.</u>, 1951, <u>45</u>, 702).
- 18. Goncalves de Lima, Larios, Zapata, and
  Dzienozielewsky, <u>Cigncia</u> (Mex.), 1952,
  12 31 (Through <u>Chem. Abs.</u>, 1953, <u>47</u> 6492).
- 19. Kupchan and Doskotch, J. Med. Pharm. Chem., 1962, 5 657.
- Furukawa and Soma, <u>J. Pharm. Soc. Japan</u>,
   1961, <u>81</u>, 565, 559.
   Furukawa, <u>J. Pharm. Soc. Japan</u>, 1961, <u>81</u>, 570.
- 21. Buchi, Greuter, and Tokoroyama, <u>Tetrahedron</u> Letters 1962 No. 18, 827.
- 22. Hesse, Arch. Pharm., 1895, 233, 684.

23.	Dutta and Sastry, Ind. J. Pharm., 1958, 20 (10), 302.
24.	Rao, Row, and Murty, Current Sci., 1958, 27 168.
25.	Pailer, Belohav, and Simonitsch, Monat.,
	1955, 86, 676.
26.	Frickhinger, Buchn. Rep. Pharm., 1851, 3, 7, 1.
27.	Pailer and Schleppnik, Monat., 1957, 88. 367.
28.	Green, Eugster, and Karrer, Helv. Chim. Acta.,
	1954, <u>37</u> , 1717.
29.	Ryo, Folia Pharmacol. Japan., 1927, 4, 123.
30.	Tseng and Ku, Acta Chim. Sinica, 1957, 23 (2), 157.
31.	Tseng and Ku, Acta Pharm. Sinica. 1958, 6, 33.
32。	Tomita and Sagasawa, J. Pharm. Soc. Japan,
	1959, 72, 1470.
33.	Krishnaswamy, Marnjunath, and Veukato Rao,
	J. Ind. Chem. Soc., 1935, 12, 476.
34.	Coutts, Stenlake, and Williams, J. Pharm.
	Pharmacol., 1959, 11, 607.
35.	Ganshirt, Pharmazie, 1953, 8, 584.
36.	Peacock, Amer. J. Pharm., 1891, 63, 257.
37.	Ferguson, Amer. J. Pharm., 1887, 59, 481.
38.	Stenlake and Williams, J. Pharm. Pharmacol.,
	1954, 6, 1005.
39.	R.T. Coutts, Ph.D. Thesis, Glasgow University,
	September 1959.

-167-

- 40. Ryo, Ber. Ges. Physiol. Fontl., Pharmacol., 1927, 40, 462.
- 41. Spica, Gazz. Chim. Ital. 1887, 17, 313 (through J. Chem. Soc. Abs., 1888, 82).
- 42. Calentano and Kind, J. Org. Chem., 1953, 18, 1473.
- 43. Castille, J. Pharm. Belg., 1922, 4, 125, 141, 569.
- 44. W.D. Williams, Ph.D. Thesis, Glasgow University, December, 1955.
- 45. Steele, Stenlake, and Williams, J. Chem. Soc., 1959, 3289.
- 46. J.W. Steele, Ph.D. Thesis, Glasgow University, September, 1958.
- 47. Stenlake and Williams, J. Chem. Soc., 1955, 2114.
- 48. Chanley and Polgar, J. Chem. Soc., 1954, 1003.
- 49. van Tamelen, Osborne, and Bach, J. Amer. Chem. Soc., 1955, 77. 4625.
- 50. Stenlake and Williams, Unpublished Work.

51. Hansen, Ber., 1931, <u>64</u>, 67. Ruzicka and van Melsen, <u>Helv. Chim. Acta.</u> 1931, <u>14</u>, 397. Arth, J. <u>Amer. Chem. Soc.</u>, 1953, <u>75</u>, 2413. Stenlake and Williams, <u>J. Chem. Soc</u>., 1959, 2627.

52. Ruzicka and Pieth, Helv. Chim. Acta, 1931, 14, 1690.

- 53. cf. Cavallito and Haskell, J. Amer. Chem. Soc. 1946, 68, 2332.
- 54. Sorensen and Hougen, Acts Chem. Scand., 1948, 2, 447.
- 55. Herz, J. Amer. Chem. Soc., 1951, 73, 4923.
- 56. Birch, Collins, and Penfold, Chem. and Ind., 1955, 1773.
- 57. Pattner, Helv. Chim. Acta, 1941, 24, 283.
- 58. Marrison, J. Chem. Soc., 1951, 1614.
- 59. Lecompte, Compt. rend., 1945, 221, 50.
- 60. Ruzicka, Imperientia. 1953, 2: 357.
- 61. Prelog, Schenker, and Klung, <u>Helv. Chim.</u> Acta, 1952, <u>36</u>, 471.
- 62. Prelog, Sohenker, and Gunthard, Helv. Chim. Acta, 1952, 35, 1598.
- 63. Goissman, Deuel, Bonde, and Addicott, J. Amer. Chom. Soc., 1954, 76, 685. Geissman and Deuel, J. Amer. Chem. Soc., 1957, 79, 3778.

Geissman and Deuel, Chem. and Ind., 1957, 328.

- 64. Barton and de Mayo, J. Chem. Soc., 1957, 150, and references cited therein.
- 65. Suchý, Horák, Herout, and Sorm, Coll. Czech. Chem. Comm., 1957, 22, 1902.

-169-

- 66. Klyne, <u>Chem. and Ind.</u>, 1954, 1198. James and Shoppee, <u>J. Chem. Soc.</u>, 1956, 1059.
- 67. Suchý, Horák, Herout, and Sorm, <u>Chem. and Ind.</u>, 1957, 894. Suchý, Horák, Herout, and Sorm, <u>Croatica Chem.</u> <u>Acta</u>, 1957, <u>29</u>, 247.
- 68. <u>cf</u> Suchy, Herout and Sorm, <u>Coll. Czech. Chem. Comm.</u>, 1962, 27, 1905. and references cited therein.
- 69. Von Fuw and Reichstein, Helv. Chim. Acta, 1946, 29, 654.
- 70. Leffek, Robertson and Sugamori, <u>Canad. J. Chem.</u>, 1961, <u>39</u>, 1989. Llewellyn, Robertson and Scott, <u>Canad. J. Chem.</u>, 1960, <u>38</u>, 1505. Leffek, Llewellyn and Robertson, <u>Canad. J. Chem.</u>, 1960, <u>38</u>, 222.
- 71. Tiers, J. Phys. Chem., 1958, 62, 1151.
- 72. Birch, Grimshaw, Penfold, Sheppard and Speake, J. Chem. Soc., 1961, 2286.
- 73. Rowland and Roberts, J. Org., Chem., 1963, 28, 1165.
- 74. Barton and Gupta, J. Chem. Soc., 1962, 1961.

-170-

- 75. Bellamy, <u>The Infra-Red Spectra of Complex Molecules</u>. Methuen, London, 1962(a) p.34; (b) p.13; (c) p. 179;
  (d) p. 96.
- 76. Jones, Humphries, Herling and Doleringer, J. Amer. Chem. Soc., 1952, 74, 2820, 6319.
- 77. Jones, Chem. In. Canad., 1950, 2, 26 (94).
- 78. Professor V. Herout, Private Communication.
- 79. Rao, Kelkar and Bhattacharyya, Tetrahedron, 1960, 9, 275.
- 80. Kanzawa, Kamis, Sumi, Nishikawa, J. Amer. Chem. Soc., 1958, 80, 3705.
- 81. Dauben, Hayes, Schwartz and McFarland, J. Amer. Chem. Soc., 1960, 82, 2232. Dauben, Schwartz, Hayes and Hance, J. Amer. Chem. Soc., 1960, 82, 2239.
- 82. Geissman and Ellestad, J. Org. Chem., 1962, 27, 1855.
- 83. <u>inter alia</u>: Doleys and Herout, <u>Coll. Czech. Chem. Commer</u> 1962, <u>27</u>, 2654. Suchy, Herout and Sorm, <u>Coll. Czech. Chem. Commer</u> 1962, <u>27</u>, 1905.

de Villiers, J. Chem. Soc., 1961, 2049.

- 84. Jones, Angell, Ito, and Smith, <u>Canad. J. Chem.</u>, 1959, <u>37</u>, 2007.
- 85. Woodward and Kovach, J. Amer. Chem. Soc., 1950, 72, 1009.

86.	Herz, de Vivar, Roma and Viswanathan, J. Amer. Cham. Soc.,
	1963, 85, 119.
87。	Herz, Veda and Inayama, Tetrahedron, 1963, 19, 483.
88。	Pinhey and Sternell, Tetrahedron Letters, 1963, No. 4, 275.
89.	Prochazka, Čekan and Bates, Coll. Czech, Chem. Comm.,
	1963, 28, 1202.
90.	Pines and Eschenazi, J. Amer. Chem. Soc., 1955, 77, 6314.
<b>91</b> 。	Prelog and Scharker, Helv. Chim. Acta, 1952, 35, 2044.
92.	References cited by Raphael, Proc. Chem. Soc., 1962, 97.
93.	Bloomquist and Goldstein, J. Amer. Chem. Soc.
	1955, <u>77</u> , 998.
94.	Braude, Fawcett and Webb, J. Chem. Soc., 1954, 1049.
95。	O'Shaughnessy and Rodebush, J. Amer. Chem. Soc.9
	1940, 62, 2906.

96. cf. Takada, Kubota and Shimaoka, Tetrahedron, 1959,7,62.

97. Bancroft, Haddad and Summers, J. Chem. Soc., 1961, 3295.

98. Cram, J. Amer. Chem. Soc., 1952, 74, 2149.

99. Allen, Filington and Meakins, J. Chem. Soc. 1960, 1909.

100. Fates, Yoda and Mann, J. Amer. Chen. Soc., 1958, 80, 202.

101. Bellamy and Williams, Trans. Farad. Soc., 1959, 55, 14.

- 102. cf. Bladon, Fabian, Henbest, Koch and Wood, J. Chem. Soc., 1951, 2402. Johnson, Idler, Meloche and Baumann, J. Amer. Chem. Soc., 1953, 75, 52.
- 103. Woodward, Sondhelmer, Taub, Heusler and McLamore, J. Amer. Chem. Soc., 1952, 74, 4223.
- 104. Turner, J. Amer. Chem. Soc., 1950, 72, 579.
- 105. Barton, Cheung, Cross, Jackman and Martin-Smith, J. Chem. Soc., 1961, 5061.
- 106. Bowman and Fordham, J. Chem. Soc., 1952, 3945.
- 107. Martin-Smith, Smith, Stenlake and Williams, Tetrahedron Letters, 1963, No. 24, 1639.
- 108. Brown, Brewster and Schechter, J. Amer. Chem. Soc., 1954, 70, 467.
- 109. Inhoffen, Becker and Kolling, Ann., 1950, 568. 181. Moffett and Anderson, J. Amer. Chem. Soc., 1954, 76. 747.
- 110. Butenandt and Wolff, Ber., 1935, <u>68</u>, 2091. Feiser and Feiser, "Steroids" Chapman Hall, London, 1959, pp. 282-283.
- 111. Dreiding, Chem. and Ind., 1954, 1419.
- 112. Bates and Gale, J. Amer. Chem. Soc., 1960,82,5749.

- 113. Steele, Stenlake and Williams, Chem. and Ind., 1959, 1384.
- 114. Holker, Holker, AcGookin, Robertson, Sargeant and Hathway, J. Chem. Soc., 1957, 3746. Burkhill, Holker, Robertson Taylor J. Chem. Soc., 1957, 4945.

- eta 2 1 --

- 115. Araus and Stern J. Amer. Chem. Soc., 1962, 84, 2893.
- 116. Barton in Rodd, "Chemistry of Carbon Compounds" vol. IIB, Chapter XII, Elsevier, Amsterdam, 1953, p. 884.
- 117. Clayton, Nature, 1961, 190, 1071.
- 118. Glemser and Rieck, Angew. Chem., 1957, 69, 91.

The Structure of Petaline Chloride,

an Alkaloid Isolated From

Leontice leontopetalum L.

#### HISTORICAL INTRODUCTION

Certain plants belonging to the genus Leontice (family Berberidaceae) have a long history of application in folk medicine, which in the case of two species can be traced back to the early Greeks. One of these species, referred to by Guntherl as Leontice chrysogonum (syn., leontopetalon) was employed by the ancients as a remedy for snakebite and in the treatment of sciatica. The other, which was probably Leontice leontopetalum (syn., chrysogonon), they reputedly used for treating "bitings of the shrew mouse". In more recent times the people of eastern . Mediterranean countries have employed the tubers of Leontice leontopetalum as a soap substitute<sup>2,3,4,5</sup>, a snake-bite remedy<sup>3</sup>, a corrective for overdoses of opiates<sup>2</sup> and in the treatment of epilepsy3,6. Yet despite this long established use in folk medicine few scientific studies of Leontice species, apart from some Russian work, appear to have been undertaken. Accordingly in 1955 an investigation of Leontice leontopetalum a species thought to have been first introduced into Great Britain in 15977 - was undertaken in these

laboratories<sup>3</sup>,9,10,11

Using a modification of a method developed by Power and Salway<sup>12</sup>, McShefferty<sup>10</sup> isolated a saponin from the tubers of L. <u>leontopetalum</u>. This saponin, which is doubtlessly the agent responsible for the soap-like activity of the tubers<sup>2,3,4,5</sup>, analysed for  $C_{69}H_{112}O_{36}\circ 5H_{2}O_{3}$  and had m.p. 236-238°, and  $[O_{30}+15.1^{\circ}$ . It gave a cosanacetate,  $C_{109}H_{152}O_{56}$ , m.p. 155-156°,  $[O_{30}+19.9^{\circ}$ . Acid hydrolysis of the saponin afforded the sapogenin,  $C_{30}H_{48}O_{4}$ , m.p. 333-334° and  $[O_{30}+78^{\circ}$ , which was shown to be identical with hederagenin (I). The sugar portion of the saponin was found to consist of four D-glucose and three L-arabinose units<sup>10</sup>.



From the petroleum ether extractives of the tubers of L. leontopetalum, McSheffertyl0 isolated ceryl alcohol, and a compound concluded to be a  $3-\beta$ -hydroxy- $\Delta^7$ -sterol, C<sub>29</sub>H<sub>460</sub>, m.p. 155.5-156°. Examination of the fatty acids in the extractives indicated the presence of palmitic, stearic, oleic and linoleic acids, and probably **n**-hexacosanoic acid. Glycerol was also shown to be present. In addition, AcShefferty<sup>10</sup> isolated three alkaloids from the aqueous acid extraction of the defatted plant material.

Alkaloids had been isolated previously from other species of the genus Leontice. In 1932 Orekhov and Konovalova<sup>13,14</sup> demonstrated the presence of at least four alkaloids in the tubers of L. ewersmanni Bge. (which earlier had been mistakenly identified by Hooker Jr. and Thomas as L. leontonetalum<sup>15</sup>), a plant found in Persia and Turkestan. These authors showed that the total basic fraction of L. eversmanni amounted to 0.4% of the dry weight of tubers. One alkaloid which was designated leontamine was isolated as an oil, b.p. 118-119°/ 4mm, [a], +2.53 and 1, 1.5113 and gave an analysis in agreement with the empirical formula C14H20N2. The second base, the crystalline leontidine had m.p. 116-118°, but Orekhov and Konovalova quote no empirical formula for this compound in their papers13,14. The third alkaloid was obtained only as a picrate, m.p. 176-178°, whilst the fourth was not isolated, but only inferred to be present.

Later Yunusov and Sorokina<sup>16</sup> re-examined the alkaloids from the tubers of <u>L</u>. <u>eversmanni</u> and the formula  $C_{15}H_{20}N_20$  was assigned to leontidine and its [04] was quoted as - 188.7°. Attempts were made to show that leontamine was the optical antipode of sparteine (syn., pachycarpine) (II) a base present in the aerial parts of the plant<sup>16</sup> but not in the tubers. However the experimental evidence failed to confirm this assumption.



II

Yunusov and Sorokina<sup>16</sup> isolated a further alkaloid,  $C_{15}H_{24}N_{20}$ , m.p. 103-104°, from the tubers, which they designated leontine. The aerial portions (0.87% alkaloids) were found to contain leontidine, sparteine (II), and d-lupanine (III).



Working on the same species, Platonova, Kuzovkov, and Massagetov<sup>17,18</sup> isolated two further bases, tapsine (VI) the dilactone of tapsinic acid (V) (which biogenetically may be derived from the aporphine alkaloid magnoflorine (IV) by a Hofmann-type degradation and oxidation), and isoleontine,  $C_{15}H_{24}N_{2}O$ , m.p. 107-108°,  $[\propto]_p$ -78.2°, thought to be related to matrine (VII).



-180-

No further alkaloids were detected in any part of the plant<sup>19,20</sup>.

Recently, Rulko and Proskurnina<sup>21</sup> confirmed that leontine was a stereoisomer of matrine by dehydrogenation to octodehydromatrine (VIII).





The pharmacological properties of the alkaloids from L. <u>eversmanni</u> appear not to have been reported, although the saponins from the tubers of this plant, like all saponins, have been shown to exhibit in <u>vitro</u> haemolytic activity<sup>22</sup>,

Yunusov and Sorokina also investigated L. <u>Alberti</u> Bge.<sup>16</sup> a rare plant found in the mountainous regions of central Asia. From it they isolated an unidentified liquid base, a small amount of a crystalline alkaloid, m.p. 180-183° and methylcytisine (IX).

-181-





TX.

To explain the frequent occurrence of quinclizidine alkaloids belonging to the lupanine, cytisine, sparteine, and matrine series in the same plant, Yunusov and Sorokina<sup>16</sup> suggested a scheme (akin to that portrayed in scheme I) of skeletal interconversions using lupinine (X) as the basic unit.

If an intermediate of type XI were to be formed from the condensation of lupinine with a piperidine derivative, it is readily seen that ring closure via route A would give rise to the matrine skeleton (XII) whilst ring closure via route B would lead to the sparteine skeleton (II).



sparteine (II). Thus a preformed piperidine ring is not involved in the formation of rings C and D in this alkaloid.



A recent examination of <u>Leontice odessane</u> by Kolisnichenko<sup>27</sup> led to the demonstration of the presence of five alkaloids, comprising 2.6% of the plant material. However, none of the individual alkaloids were characterized chemically.

Of the three bases isolated from L. leontonetalum during the earlier work in these laboratories<sup>8</sup>,10 one, a non-crystalline base had physical constants and empirical formula  $C_{14}H_{26}N_{2}$ , (b.p. 118-120°/4 mm,  $[\swarrow]_{p}+2.78^{\circ}$ , tion), by 1.5117) in agreement with those cited by Orekhov and Konovalova for leontamine<sup>13</sup>,14, the constitution of which is still unelucidated. Direct comparison of specimens was not made but it was assumed that they were identical. The specimen obtained by McShefferty was shown to be fully saturated both by chemical and physical means<sup>10</sup>. The second base isolated was named leonticine,

-184-

C<sub>20</sub>H<sub>25</sub>NO<sub>3</sub>, m.p. 118.5-119.5°, [a]\_0(StoH) and was obtained in 0.018% yield from the tubers. Although it exhibited the same melting point, leonticine was obviously not identical with leontidine, C<sub>25</sub>H<sub>20</sub>NO<sub>2</sub>, isolated from <u>Leontice ewersmanni</u>13,14,16. Leonticine slowly decolourized acidic potassium permanganate but attempts at catalytic hydrogenation were unsuccessful.

The third alkaloid, a water-soluble quaternary base, was isolated as the pink micro-crystalline reineckate, C<sub>20</sub>H<sub>22</sub>O<sub>3</sub>N [Cr (SCN)<sub>1</sub>(NH<sub>3</sub>)<sub>2</sub>], m.p. 179-181°. Decomposition of the reineckate by the silver sulphate barium chloride method of Dutcher<sup>28</sup> gave the quaternary chloride as deep yellow deliquescent scales, m.p. 140-143°. This quaternary salt, designated petaline chloride, was shown by elemental and functional group analysis to be C18H16ONCL (OCH2)2. It was found to be optically active,  $[\alpha]_D^{20} + 11.3^{\circ}$ (H20), and exhibited ultraviolet absorption maxima at 224 mp (E, 20,576), 280 mp (E, 11,600) and 328 mµ ( $\epsilon$ , 334). Petaline chloride was reported to give indistinct spot tests for the presence of a phenolic hydroxyl group but it gave positive tests typical of the berberine type of alkaloid<sup>29</sup>. Petaline

-185-

chloride was readily converted into the corresponding quaternary picrate, chloroplatinate and sulphate. It was reported to undergo reduction with zinc and dilute hydrochloric acid, and with hydrogen over Adam's catalyst in aqueous acetic acid (uptake: one mole) to yield dihydropetaline chloride С20H24N03C1 · 2H20, п.р. 122-125 (1)-16.7 (H20) with Max 224 (8,15,364) and 280 mp (8,11,580). McShefferty<sup>10</sup> concluded, on evidence which was by no means unequivocal, that petaline chloride was a pseudoquaternary base (partial structure XVII) whose double bond appeared not to be affected by hydrogenation although isomerization appeared to have taken place about the optical centre, as indicated from the inverted optical rotation.



# The oxidation of petaline chloride with alkaline potassium permanganate was stated to give inconclusive results. Petaline chloride was found to be extremely unstable to alkali, yielding black tars. In one experiment<sup>10</sup> on treatment with excess 4% barium hydroxide, petaline chloride

-186-

afforded a 30% yield of leonticine, m.p. 117.5-118.5°, together with another substance, m.p. 105-110°, exhibiting reactions typical of a ketone or aldehyde. This carbonyl compound, designated oxypetaline chloride gave what were regarded as unsatisfactory analyses, both as the free base and as the 2,4dinitrophenylbydrozone and no formula was assigned to it. It was further suggested that leonticine was tetrahydroanhydropetaline<sup>10</sup>, For analogy, comparison was made with the work of Gadamer 30,31 and Perkins<sup>32</sup> as modified by Faltis<sup>33</sup>. In this work three molecules of berberine (ammonium form, XVIII) are converted, in the presence of alkali, into two moles of tetrahydroanhydroberberine (XIX) and



XVIII



-38Em

YTY

302

On the basis of this analogy the second product of the action of barium hydroxide on petaline chloride should have been a lactam incapable of 2,4-dinitrophenylhydrazone formation and not a ketone or aldehyde. On the basis of ultraviolet spectral comparisons, McShefferty<sup>10</sup> suggested that petaline chloride was an isoquinoline alkaloid. Further he suggested that the tertiary base, leonticine was an artefact resulting from the action of alkali on petaline chloride during the work-up of the plant, a conclusion supported by the present investigation (vide infra).

Petaline chloride was tested pharmacologically by Ahmad and Lewis<sup>34</sup> who claimed it to be a more potent convulsant than leptazol although at low dosage levels petaline chloride was found to reduce the convulsant activity of leptazol and exhibited muscle relaxant activity.

1

#### DISCUSSION

A survey of the work previously performed with petaline and leonticine as outlined in the Introduction revealed the conspicuous absence of any attempted Hofmann degradation of any quaternary salt derived from leonticine. Since this reaction has proven to be of great value in the structural elucidations of a variety of alkaloids (as with aporphine alkaloids such as isocorydine methchloride<sup>35</sup>, the erythrina alkaloids such as tetrahydroerysodine  $(XXI)^{36}$  and in the structural elucidation of the interesting alkaloid protostephanine (XVII)37 isolated from Stephania japonica<sup>38</sup> in addition to the classical structural elucidations of the tropine and pomegranate alkaloids), it seemed a logical first step to apply the Hofmann degradation to leonticine which the carlier work would indicate was the methine base of petaline. This last deduction follows from a consideration of the method of formation of leonticine from petaline, by the action of base on the latter compound and from its analytical figures.

-190-




# XXI

XXILL

Indeed employing milder conditions than used by the earlier workers<sup>10</sup> that is passage over Amberlite IRA-400 anionic exchange resin, the conversion yield of petaline chloride into leonticine was raised to 65% while petaline reirockate was converted directly into leonticine in <u>ca</u> 40% yield.

Quaternization of leonticine by the action of .methyl iodide in dry acetone readily afforded the corresponding quaternary iodide C<sub>21</sub>H<sub>23</sub>O<sub>3</sub>NI, a.p. 169-171° (addition of 1 mole methyl iodide) in good yield. This was subjected to passage over IRA-400 anion exchange resin but Hofmann degradation failed to occur - the product being the corresponding leonticine

-191-

methohydroxide m.p. 150-155°. Treatment of the methohydroxide with 5% sodium ethoxide in ethanol afforded a nitrogen-free methine m.p. 111-113°,  $[\alpha]_{n} = 0^{\circ}$  analysing for  $(C_{6}H_{6}O)_{n}$ . Molecular weight determination by the mass spectographic method showed the molecular weight to be 282 thus fixing the formula of the nitrogen-free product as This product therefore has been formed C18H18020 from leonticine by the net elimination of the elements of trimethylamine and water. That the lost nitrogen containing fragment was indeed trimethylamine was shown by the isolation of the volatile fragment as its pierate. The picrate thus isolated was identical with authentic trimethylamine picrate (mixed m.p., infra-red spectra, and analyses).

The infra-red spectrum of the nitrogen-free fragment from the Hofmann degradation of leonticine methohydroxide measured in carbon tetrachloride solution (2.07 mg/6 ml) clearly showed an intramolecularly hydrogen bonded hydroxyl group as evidenced by the position of the O-H stretching frequency at  $3545 \text{ cm}^{-1}$  ( $\xi$ , 144)<sup>39a</sup>. That the same intramolecular hydrogen bonding was present in leonticine was shown by the retention of this absorption at  $3535 \text{ cm}^{-1}$  ( $\xi_{2}137$ ) in the infra-red spectrum of this compound measured in carbon tetrachloride (2.44 mg/6ml). These two results taken in conjunction with the fact that methoxyl determinations show the presence of two methoxyl groups in both leonticine and the nitrogen free degradation product (thus accounting for all the oxygen atoms present in both compounds) shows that the -O-H group is adjacent to a methoxyl group. Petaline, leonticine and the nitrogen-free Hofmann degradation product all give a positive Gibb's test<sup>40,41</sup> for a phenol possessing a free para position, thus showing the fragment XXIII to be present in all three compounds.



#### XXLIX

The nuclear magnetic resonance spectra of leonticine and its nitrogen-free degradation product (as acetate) clearly shows the conversion of the system XXIV into XXV

# $\begin{array}{c} \text{Ar-CH}_2\text{-CH}_2\text{-N} \xrightarrow{\text{CH}_3} \longrightarrow \text{Ar-CH} = \text{CH}_2\\ \text{XXIV} \qquad \text{XXV} \end{array}$

Thus absorption in leonticine of intensity 6 protons at 7.65 typical of methyl groups substituted on nitrogen<sup>35,42</sup> and complex methylene absorption of intensity 4 protons in the range 7.0-7.61 are replaced in the nitrogen-free degradation product (as acetate) by doublets of intensity 1 proton at 4.57-(J=17 c.p.s.) and 4.97 (J=10 c.p.s.), each peak showing fine splitting, typical of the vinylidene protons on the styryl double bond<sup>35,42,43</sup>. Further the infre-red spectrum of the nitrogen-free degradation product (in potassium chloride disc) exhibits maxima at 3086 cm<sup>-1</sup> and 905 cm<sup>-1</sup> typical of a vinylidene double bond<sup>39b</sup>. Such absorption was absent in the infrared spectrum of leonticine.

The ultra-violet spectrum of leonticine exhibited  $\lambda \max 216 \ \mathrm{mp} (\xi, 23, 400) \ \mathrm{and} 299 \ \mathrm{sp} (\xi, 21, 100) \ \mathrm{which}$ would not be incompatible with a stilbene derivative, being very similar to the absorption from the cismethine base XXVI of laudanosine which has  $\lambda \max 215 \ \mathrm{mp}$ ( $\xi, 23400$ ) and 294  $\ \mathrm{mp} (\xi, 10, 960)^{\mathrm{h}4}$  and 3, 3', 4, 4'tetramethoxystilbene (XXVII) which has 214  $\ \mathrm{mp} (\xi, 26, 900)$ and  $\lambda \max 303_{\mathrm{m}}(\xi, 13, 500)^{\mathrm{h}5}$ .





XXVIII

#### XYZCK

The ultra-violet spectrum of leonticine run in N. HCl was virtually unchanged showing maxima at 216 mp ( $\xi$ ,29,100) and 239 mp ( $\xi$ ,23,500) thus indicating that leonticine is not any 2-unsaturated amine (enamine)<sup>46</sup>.

The ultra-violet absorption spectrum of the Hofmann degradation product of leonticine exhibits Mmax 209 mp ( $\varepsilon$ 27,000), 269 mp ( $\varepsilon$ ,23,900) and 305 mp ( $\varepsilon$ ,21,100). The first and last maxima may be attributed to a cis-stilbene system whilst the maximum at 269 mp is not unlike the contribution of the styryl vinylidene group in the phenanthrene derivative (XXVIII), of the aporphine alkaloid chakranine, which accounts for the maximum in its spectrum at 259 mp ( $\varepsilon$ ,37,150). As with the degradation product YXVIII, the maximum at 269 mp in the ultra-violet spectrum of the Hofmann degradation product of leonticine measured in

-195-

N KOH is virtually unaltered appearing at 272 mp (E,21,700)



#### XXVIIII

Ozonolysis of the nitrogen-free Hofmann degradation product of leonticine afforded  $\underline{p}$  - methoxy benzaldehyde, identified by comparison of its infra-red spectrum with that of authentic material and by lack of mixed melting point depression in the 2,4-dintrophenylhydrazones prepared from the two specimens, thus identifying one half of the stilbene system. However the nature of the remainder of the molecule could not be determined from this experiment owing to the intractability of the material. The formation of  $\underline{p}$ -methoxybenzaldehyde on ozonolysis, incidentally, would explain the strong odour of  $\underline{p}$ -methoxy benzaldehyde, encountered during attempted Kuhn-Roth estimation of C-methyl groups in leonticine.

Ozonolysis of leonticine methiodide in ice-cold ethanol readily permitted the separation of the water soluble quaternary salt formed from one half of the stilbene from the p-methoxybenzaldehyde (identity again confirmed) formed from the other. This salt was not characterized as such but was subjected to Hofmann degradation to generate the substituted styrene and the resulting nitrogen-free oily product was hydrogenated over Adam's catalyst in ethanol. The resulting oil thus obtained showed no vinylidene absorption in the infra-red (measured in carbon tetrachloride solution; 2.85 mg/10 ml) indicating full reduction of the styryl double bond generated during the Hofmann eligination. Horeover no carbonyl absorption was present but two O-H stretching bands were present. The high frequency hydroxyl absorption at 3615 cm<sup>-1</sup> (E,22) can be assigned to the benzylic hydroxyl group formed by the reduction of the aldehyde group generated from the stilbene double bond on ozonolysis, whilst the low frequency absorption at 3547 cm<sup>-1</sup> (£,112) can be assigned to the phenolic hydroxyl group intramoleculary hydrogen bonded to the adjacent methoxyl group.

-197-

The constitution of this hydroxy-, methoxy-, hydroxymethyl- ethyl- benzene is limited to the one formula XXIX.



This stems from the fact that it has a free position para to the hydroxyl group (positive Gibbs test in accord with the similar positive tests given by petaline, leonticine and the methine derived from leonticine methiodide) and that there is no intramolecular hydrogen bonding of the benzylic hydroxyl group as would occur if this function were ortho to either the phenolic hydroxyl group on the aethoxyl group (as evidenced by its absorption frequency of 3615 cm-1) when the need for the ethyl group and the hydroxymethyl group to be ortho to one another is taken into account. From this it follows that the structure of the nitrogen free degradation product of leonticine methohydroxide is XXX, that of leonticine is XXXI and that of petaline, XXXII.

-198-





XXX

#### XXXX

TIXXXX

Petaline (XXXII) being the only compound in the above series with a centre of asymmetry would be expected to be the only compound showing optical activity as indeed is the case.

The substitution pattern of the benzene ring in the tetrahydroisoquinoline molety of petaline (XXXII) is of interest biogenetically. Normally the substitution pattern is such that oxygen functions occur at two, three or four of the positions, 6,7,3,3' and 4' of the benzyl tetrahydroisoquinoline nucleus as for example in the alkaloids corpavorine (6,7,8,4') (XXXIII) laudanosine (6,7,3',4') (XXXIV) magnocurarine (6,7,4) (XXXV) and the dioxygenated (6,4') alkaloid XXXVI<sup>47</sup>, reflecting their biogenesis from hydroxyphenylalanines<sup>43</sup>.

-200-





XXXIII



XXXIA



YOCKY

JECKAI

Indeed it would appear that no benzyltetrahydroisequinoline alkaloid bearing an oxygen function at C-5 has so far been reported 53, 54 and so petaline would seem to be of a unique substitution pattern.

A possible explanation of the oxygenation pattern in the benzene ring of the tetrahydroisoquinoline nucleus of petaline would perhaps be that loss of hydroxyl from C-7 of a precursor could occur through a quinonoid intermediate (compare corresponding loss of hydroxyl from the precursor of volucrisporin<sup>55,56</sup>) followed by the introduction of oxygen ortho to the phenolic group on C-6 prior to methylation of the latter<sup>56</sup>.

It is perhaps also of some interest in connection with the unusual oxidation pattern present in petaline that recently otobain (XXXVII)<sup>57</sup> and hydroxyotobain (XXXVIII)<sup>58</sup> have been shown to possess a 5,6-dioxygenated substitution pattern, which is different from that normally pertaining in lignans which are usually 6,7dioxygenated (plus, occassionally, 5-oxygenated)<sup>59</sup> as in, for example, conidendrin (XXXIX)<sup>60</sup>.



XXXXIII

XXXXX

-201-



Otobain, which thus appears to be a case where the hydroxyl on C-7 responsible for ring closure at  $C_{10}$ , para to it, has been lost and an oxygen function has been inserted at C-5, represents a situation strictly paralleling that in petaline-viz loss of an oxygen (from porhaps C-7) and insertion of oxygen on the other side of a second oxygen ortho to that lost.

In order to clinch the evidence for the structure of petaline (XXXII) it was necessary to confirm the structure of XXIX by chemical means. It was therefore decided to attempt the synthesis of authentic material of this structure by an unambiguous route. The pathway chosen is shown in scheme 2. Thus isovanillin (XL) was chosen as a suitable starting material and was converted into the corresponding allyl ether XLI by boiling under reflux with a 1 molar ratio of allyl bromide in acctone containing excess of anhydrous potassium carbonate The product XLI b.p. 136°/1mm Hg was subjected to Claisen rearrangement 50 by heating to 300°C for 5 minutes (oil-bath) and fractionally distilled the fraction b.p. 142-143%/1mm Hg of product XLII being collected.

-202-

The authenticity of XLII was established by catalytic reduction of the vinylidene double bond followed by decarbonylation of the resultant 2-m-propyl isovanillin XLIII over 5% Palladium on charcoal by adaptation of a method described by Hawthorne and Wilt<sup>51</sup>. The product XLIV was identical (infra-red spectra of liquid films) with that obtained by the catalytic reduction of Q-eugenol (XLVII) (possessing the correct physical properties prepared from guaiacol (XLV) by Glaisen rearrangement of its allyl ether XLVI<sup>49</sup>.

Lithium aluminium hydride reduction of 2allylisovanillin (XLII) in dry ether smoothly converted this compound into the corresponding alcohol XLVIII which after chromatography on neutral alumina was acetylated by the action of acetyl chloride in dry acetone in the presence of excess of anhydrous potassium carbonate. Acetylation was shown to be quantitative by the absence of hydroxyl absorption in the infra-red spectrum of the oily diacetate (XLIX. Ozonolysis of the diacetate in ice-cold ethanol followed by reductive cleavage of the ozonide with

-203-

sodium borohydride<sup>52</sup> afforded the phenyl ethyl alcohol L. This compound was readily converted into the methanesulphonate derivative LI which was directly hydrogenolysed by the action of lithium aluminium hydride in dry tetrahydrofuran. Considerable quantities of high boiling material were present in the product which might indicate the occurrence of side reactions such as that depicted in LIII, and no material corresponding in properties to that obtained from the degradation of petaline could be isolated.



LITT

#### EXPERIMENTAL

Melting points were determined on a hot-stage melting point apparatus and are uncorrected. Ultraviolet spectra were measured on an Optica CF-4 recording Infra-red spectra were measured spectrophotometer. on an Unicam model SP100 equipped with a model S.P.130 sodium chloride prism-grating double monochromator operated under vacuum conditions and the Perkin-Elmer model 237 Infra-Red spectrophotometer. Optical rotations were determined on a Bellingham and Stanley polarimeter employing a 1 decimetre cell. Microanalyses were carried out by the microanalytical laboratory of the Royal College of Science and Technology, and by Drs. Weiler and Strauss, Oxford. Catalytic hydrogenations were effected in a Towers hydrogenating apparatus and were performed at room temperature and pressure.

Petaline reineckate was isolated from dry powdered Leontice leontopetalum extractives as previously described<sup>B</sup>.

-205-

# Conversion of Petaline Reineckate Leonticine (XXXI)

Petaline reineckate (5g.) in acetone (2) ml) was 1. introduced onto a column of anhydrous Amberlite resin IRA-400 (in OH form) (250g) and eluted with dry ethanol until all coloured material had been washed through. After concentration of the eluate, the brown semi-solid mass was taken up in benzone (50 ml) and the resultant solution filtered through neutral alumina (10g.) to afford a colourless solution. Concentration of this solution under reduced pressure followed by recrystallization of the solid residue from acetone afforded leonticine m.p. 121-123°, A = 0, as colourless needles (Found: C, 73.58; H, 7.76; N, 4.613 N-CH3, 13.24, 11.72; -OCH, 19.40, 17.56, 16.98; active H, 0.330. Calculated for C20H250 N: C, 73, 38; H, 7.70; N, 4.28; N-CH3, 17.72; - OCH3, 18.96; 1 active hydrogen, 0.308%). Max 216 mu (E, 28, 400) and 299 mu (E,21,100); in N HCl 216 mm (E,29,100) 289 mm (E,23,500); Ymax (C Cl<sub>4</sub>) 3535 cm<sup>-1</sup> (E, 137) (intramolecularly bonded hydroxy1).

11. Petaline chloride (7.2g.) in ethanol (25 ml) was eluted through an Amberlite IRA-400 anionic exchange column (OH form). Work-up of the eluate as described above afforded leonticine m.p. 121-123° (4.8g; 65%).
Leonticine Methiodide.

Leonticine (3.1%) in dry acetone (50 ml) was treated with methyl iodide (4 ml) and the solution heated under reflux for 2.5 hr. The crystals which were deposited were collected by filtration and dried affording 4.02g product. The methiodide was recrystallized to constant m.p. 169-171° (Found: C, 53.69; H,6.89; N,3.1  $C_{21}H_{28}O_{3}NI$  requires c,53.73; H,6.2g; N,2.98%) max 217 mµ (£,63,000) and 297 mµ (£,24,200) Hofmann Degradation of Leonticine Methiodide.

Leonticine methiodide (3.6 g.) in methanol (60 ml) was slowly eluted down an Amberlite IRA-400 anionic resin column (OH form) (60 g.). The dark brown eluate was taken to dryness yielding dark needles shown to be leonticine methohydroxide m.p. 150-155° (2.66).

Methohydroxide (1.0g.) in 5% sodium ethoxide in ethanol (25 ml) was heated for 2.5 hours under nitrogen in a closed system, the effluent gas passing through saturated ethanolic picric acid, Water (10 ml) was added to the reaction mixture which was then made acidic with acetic acid and then exhaustively extracted with benzene. Washing the benzene solution with water and drying over anhydrous sodium sulphate afforded orange crystals (0.63g.) Recrystallization from ether/ petroleum ether (40-60°) gave <u>Hofmann degradation</u> <u>product</u> (XXX) m.p. 111-113° (C1-0° (Found: C,76.66; H,6.78; 0-CH<sub>3</sub>, 13.42, 11.98; molecular weight by mass spectrograph 282.  $C_{18}H_{18}O_3$  requires: C,76.58;  $H_76a42$ ; 2-0CH<sub>3</sub> require 2195% mol. wt. 282). Amax 209 mµ ( $\xi_227,000$ ) 269 mµ ( $\xi_23,900$ ) and 305 mµ ( $\xi_21,100$ ) in N KOH, 254 mµ ( $\xi_21,800$ ), 272 mµ ( $\xi_21,700$ ) and 365 mµ ( $\xi_9300$ ).

Ymax (CCl<sub>4</sub> solution) 3545 cm<sup>-1</sup> (E,144) (intramolecularly bonded hydroxyl.

Treatment of the Hofmann degradation product (1.lg) with 10% BF<sub>3</sub> in mothanol complex (35 ml) on a water-bath for 20 minutes and decantation of the reaction mixture into crushed ice precipitated crystals which were collected, dried, and recrystallized from benzene/ petroleum ether (40-60°) to constant melting point 168-170°, thus affording the <u>nitrogen-free product</u>

=208m

<u>methyl other</u> (0.84g.) (Found: C,77.28; H,6.57. C19H2003 requires, C,77.00; H,6.80%).

The crystals present in the ethanolic pieric acid solution employed as trap for the volatile nitrogenous fragment eliminated during the Hofmann degradation of leonticine methohydroxide were collected by filtration and recrystallized from ethanol and exhibited m.p. 209-211° undepressed on admixture with authentic trimethylamine picrate (Found: C,37.85; H,4.22; N, 19.26; Calculated for  $C_9H_{12}N_4O_7$  C,37.51; H,4.20; N, 19.45%). As expected the infra-red spectrum of the trimethylamine picrate derived from the Hofmann degradation of leonticine methohydroxide was completely superposable with that of authentic trimethylamine picrate.

# Ozonolysis of the Hofmann Degradation Product of Leonticine Methohydroxide

Employing the published procedure of Bauer, Birch and Ryan<sup>61</sup> mitrogen-free methine (0.3g.) in ethanol(25ml). containing conc. sulphuric acid (0.5 ml) at 0°C was rapidly ozonized for 20 minutes and the flask was then flushed with oxygen for 25 minutes. Excess 5% aqueous ferrous sulphate (25 ml) was added to decompose the ozonide and

then the reaction mixture was exhaustively extracted with ether (6 x 15 ml). After washing with water, the ethereal extractives were dried over anhydrous sodium sulphate and concentrated to dryness. The resultant brown oil (0.1g.) exhibited an infra-red spectrum superposable with that of authentic pmethoxybenzaldehyde whilst the m.p. on admixture of the two 2,4 - dinitrophenylhydrazones, from derived material and authentic p-methoxylbenzaldehyde, was undepressed, m.p. 260-261°. On standing the oil deposited crystals which were collected by filtration and washed with ice-cold ethanol giving m.p. 1840, undepressed on admixture with authentic p-methoxy-The infra-red spectrum of the crystals benzoic acid. was superposable on that of <u>p-methoxybenzoic</u> acid.

Chromatography of the oil, over neutral alumina failed to afford successful iselation of a second fragment. Ethyl acetate extraction of the aqueous residue from the ether extraction also failed to afford further material.

# Ozonolysis of Leonticine Methiodide

Leonticine methiodide (0.3g.) in ethanol (25 ml) at O°C was rapidly ozonized for 20 mins., and the flash then flushed with oxygen for 25 mins. The ozonide was hydrogenolysed over Adaa's catalyst. Removal of the catalyst by filtration and dilution of the reaction mixture with distilled water (25 ml) precipitated a brownish oil which was removed by extraction of the aqueous solution with ether. Work w up of the ethereal solution as above resulted in the isolation of further <u>p-methoxybenzaldehyde</u> again identified as above. The aqueous solution was taken to dryness on a rotary-film evaporator. The hydroscopic crystalline residue was treated with 5% sodium ethoxide in ethanol (20 ml) and the mixture heated under reflux for 1.5 hr. After dilution of the reaction mixture with water and acidification with glacial acetic acid, the reaction mixture was exhaustively extracted with other. Washing and drying the othereal solution followed by concentration to dryness afforded a yellowish oil (ca 50 mg.). The oil was hydrogenated over highly active Adam's catalyst in ethanol. The catalyst was

+211-

removed by filtration, the ethanolic solution concentrated under reduced pressure and the resultant oil twice distilled in a horizontal tube at 85°/0.04 mm Hg,affording a yellow oil (15 mg)./max (carbon tetrachloride solution) 3615 (£,22) (benzylic hydroxyl) and 3545 cm<sup>-1</sup> (£112) (intramolecularly hydrogen bonded hydroxyl), Conversion of Isovanillin (XL) into 2-Allyl-isovanillin

(XLVIII)

Isovanillin (120g.), allyl bromide (115g.) and powdered anhydrous potassium carbonate (90g.) in dry acetone (300 ml) placed in a 11 flask fitted with an efficient stirrer, were heated under reflux for 8 hours. Acetone (ca 200 ml) was taken off under reduced pressure and water (300 ml) was added. The reaction mixture was oxtracted with ether (6 x 100 ml) and the ethereal solution washed with 10% sodium hydroxide (2 x 50 ml) followed by water until the wash was neutral, then three times more. The ethereal solution was concentrated to dryness thus affording a yellow oil(147.5 g; 97.4%).

The infra-red spectrum indicated the absence of

-212-

hydroxyl absorption in the 3500 cm<sup>-1</sup> range with the appearance of a new ether (allyl) band at 1030 cm<sup>-1</sup>

The <u>allyl ether</u> (XLI) (147.5g.) was heated to 300°C (oilbath temperature 312°C) for 5 minutes. After cooling the rearrangement product was distilled under vacuum, the fraction b.p.  $142-143^{\circ}$ / 1 mm Hg. being collected thus affording 2 - allylisovanillin (XLII) (108g; 73.3%) as a pale yellow oil. Chromatography of the product over neutral alumina with ether as eluant and distillation of the oil so obtained b.p.  $142^{\circ}/1$ mm afforded a low melting solid (Found, C.68.21; H,6.40; C<sub>11</sub> H<sub>12</sub>O<sub>3</sub> requires C,68.74; H,6.30%)/max (liquid film) 3400 (hydroxyl) 3080 (vinylidene) 1085 cm<sup>-1</sup> (hydroxyl).

Elution of the column with ether: ethanol (10:1) afforded a small amount of a second aldehyde, presumed to be 6-allyl-isovanillin, which was not characterised. <u>Authentication of 2-Allyl-isovanillin (XLII)</u>

2-Allyl-isovanillin (lg.) was hydrogenated over Adam's catalyst in ethanol (hydrogen uptake: 1 mole) Removal of the catalyst by filtration and concentration of the solvent afforded the dihydro derivative as evidenced by absence of the vinylidene band at 3080 cm<sup>-1</sup> and the ethylenic double bond peak at <u>ca</u> 1640 cm<sup>-1</sup> in the infra-red spectrum. Decarbonylation of the dihydro compound by the palladium on charcoal procedure developed by Hawthorne and Wilt<sup>51</sup> (CO collected: 0.8 mole) afforded 2-methoxy-6-n-propylphenol (XLIV). Identity of this compound was confirmed by infra-red spectral comparison with

dihydroeugenol (XLIV), prepared by reduction of synthesised Q-eugenol having the correct physical constants<sup>40</sup>, the spectra being superposable. <u>Conversion of 2-Allyl-isovanillin (XLII) into 2-allyl-3-</u> acetyloxy-3-methoxybenzylacetate (XLIX)

2-Allyl-isovanillin (48g.) in dry ether was treated with excess of lithium aluminium hydride (22g.) and heated under reflux for 19 hours. Destruction of the excess of complex metal hydride by the slow addition of dilute HCl and exhaustive extraction of the acidic reaction mixture with ether gave a dark yellow ethereal solution. After washing with water and drying over anhydrous sodium sulphate, the ethereal solution was taken to dryness. The resulting dark brown 2 - allylisovanyllyl sizehol (XLVIII) (47.5g; 98%) was filtered through neutral alumina. Concentration of the ethereal eluate afforded the product as a pale yellow oil. Attempts to purify the 2-allyl-isovanillyl alcohol by high vacuum distillation were unsuccessful due to apparent polymerization of the material. The chromatographed material had/max (in carbon tetrachloride solution) 3590 (£,36) (benzylic hydroxyl) 3555 (£,203) (intramolecularly hydrogen bonded hydroxyl), 3090 (vinylidene) and 1640 cm<sup>-1</sup> (vinylidene).

2-Allyl-isovanillyl alcohol (XLVIII) (7.5g.) acetyl chloride (10 ml) and anhydrous potassium carbonate (3.5g.) in dry acetone (125 ml.) were heated under reflux for 17 hr. After cooling, water (200 ml.) was slowly added and the mixture exhaustively extracted with ether. The ethereal solution was washed with water, dried (Na<sub>2</sub>SO<sub>4</sub>) and concentrated to dryness. The oily yellow diacetate XLEX could not be induced to crystallize. Thus the product was chromatographed on neutral alumina with ether as eluant. The pule yellow oil obtained after concentration of the eluate was negative to Gibb's reagent<sup>40,41</sup>.

// max (liquid film) 3090 (vinylidere) 1765 (shoulder; acetate) 1760 (acetate) and 1635 cm<sup>-1</sup> (vinylidere).

-215-

Conversion of 2-Allyl-3-acetyloxy-4-methoxybenzyl acetate XLIX into2 Chromoschyl 3-Acetyloxybenzyl acetate (L) by Reductive Cleavage of the Intermediate Ozonide.

Allyl compound (XLIX) 2.6g.) in ethanol (80 ml) at O°C was rabidly ozonized for 35 mins. and the reaction solution flushed with exygen for 40 mins. Excess of sodium borohydride (2.6g.) was added portionwise, in a procedure analogous to that developed by Souse and Blum, and the reaction mixture kept at 0°C for 14 hrs. Excess of borohydride was decomposed by the careful addition of dilute acetic acid. Water (150 ml) was added to the slightly acidic solution and the resultant solution extracted with ether. After washing the ethereal solution with water followed by drying over anhydrous magnesiul sulphate, the ethereal solution was concentrated to dryncss. The phenylethyl alcohol (L) (2.3g) gave a negative Gibb's test whilst the infrared spectrum exhibited no maxima in the ranges 3100-3050 cm<sup>-1</sup> and 1700-1630 cm<sup>-1</sup>.

Methanesulphonate ester (LI) Formation of 2-i-hydroxyethyl-3-acetyloxy-4-methoxybenzyl acetate (LI) and Hylrogenolysis to the Proposed 2-F.thyl-3-hydroxy-4-methoxy-benzyl alcohol

2-fa-hirdromysthyle-3-acetyloxy-4-methoxy-benzyl acetate (L) (1.63g.) in dry pyridine at-5°C was treated with methanesulphonyl chloride (0.66 ml) and the mixture kept at -5°C for 19 hr. The reaction mixture was then poured onto crushed ice. The semi-solid precipitate was extracted with ethyl acetate. Washing the ethyl acetate solution with dil HCl followed by water and drying (Na<sub>2</sub>SO<sub>4</sub>) afforded a clear brown solution. Concentration of the solution gave crystalline methanesulphonate derivative (LI) (1.5g.) m.p. 103-106°, // max (KCl disc) 1765 and 1740 (acetate) 1415 (methanesulphonate, 1280 (acetate), 1245 (acetate) and 1170 cm<sup>-1</sup> (methanesulphonate).

The methanesulphonate derivative (LI) was not purified as such but was dissolved in tetrahydrofuran (4) ml), excess of lithium aluminium hydride (1.3g) added portion-wise, and the mixture heated under reflux for 42 hr. Excess of lithium aluminium hydride was destroyed by the careful addition of dil HCl and

1021700

the resultant acidic solution was extracted with ether  $(7 \ge 25 \text{ml})$ . After washing with water the ethereal solution was dried and concentrated to dryness affording a brown resinous mass (l.lg). The resinous material was extracted with ether (10  $\ge$  1 ml), the ethereal solution concentrated and the brown-oily residue distilled at  $102^{\circ}/0.05$  mm Hg thus affording a yellowish oil, in very small yield, not showing identity of infra red spectra with the product (XIIX) obtained from petaline.

See

### REFERENCES

- 1. Gunther, The Greek Herbal of Dioscorides. Oxford University Press, Oxford, 1934, p.p. 340-1, 447.
- Lindley, <u>Vegetable Kingdom</u>, Bradbury and Evans, London, 1846, p. 438.
- Low, <u>Die Flora der Juden</u>, vol 1, Damask, Vienne and Leipzig, 1929-34, p. 288.
- 4. Perrot and Matieres, <u>Premières du Regne Vegetab</u>, Tom 1, Masson et Cie., Paris, 1943, p. 870.
- Temple, <u>Flowers and Trees of Palestine</u>, Stock, London, 1907, p. 114.
- Bailey, <u>Standard Cyclopoedia of Horticulture</u>,
   Vol. 2, Macmillan, New York, 1947, p. 1839.
- 7. Loudon, <u>An Encyclopoedia of Plants</u>, Longman, Rees, Orme, Brown, and Green, 1829, p. 286.
- 8. McShefferty, Nelson, Paterson, Stenlake and Todd, J. Pharm. Pharmacol., 1956, 8, 1117.
- Nelson and Fish, J. Pharm. Pharmecol., 1956,
   8, 113<sup>1</sup>+.
- 10. J. McShefferty, Ph.D. Thesis Glasgow University, March 1957.
- 11. Nelson and Fish, J. Pharm. Pharmacol. 1959, 11, 427.
- 12. Power and Salway, J. Chem. Soc., 1913, 191.

-219-

- 13. Orekhov and Konovalova, Arch. Fharm., 1932, 270, 329.
- 14. Orekhov and Konovaleva, Khim-Farm Prom., 1932, 10, 371.
- 15. Index Nevensis Vol. II. Clarendon Press, Oxford, 1895, p. 51.
- 16. Yunusov and Sorokina, J. Gen. Cham. (U.S.S.R.), 1949, 12, 1955.
- 17. Platanova, Kuzovkov, and Messagetov, Zbur. Obshchai. Khim., 1953, 23, 880.
- Platonova, Kuzovkov, and Sheinker, Zhur.
   <u>Obshchei</u>. Khim., 1956, 26, 2651. (<u>J. Gen</u>.
   Chem. U.S.S.R., 1956, 26, 2957).
- 19. Platonova and Kuzovkov, Zhur. Obshchei. Khim., 1954, 24, 2246.
- 20. Platonova and Kuzovkov, Zhur. Obshchei Khim., 1956, 26, 283.
- 21. Rulko and Proskurnina, Zhur. Obshchei. Khim., 1961, 31, 308.
- Baibekov, <u>Tashkent Akad. Nauk. Uzbec S.S.R.</u>, 1956, 45 (<u>Through Chem. Abs.</u>, 1958, <u>52</u>, 13098)
   Schutte and Nowack, Naturwiss. 1959, <u>46</u>, 493.
   Schutte and Nowack, and Schafer, <u>Arch. Pharm.</u>, 1962, <u>295/67</u>, 20.

- 25. Schutte, Aslandow, and Schafer, Arch. Phar., 1962, 295/67. 34.
- 26. Reifer and Wiewlorowski, Abstract "A" of XIX th International Congress of Pure and Applied Chemistry, London, 1963, p.p. 286-287.
- 27. Kolisnichenko, Farm. Zhur. (Kiev), 1960, 15, 36.
- 29. Dutcher, J. Amer. Chem. Soc., 1946, 68, 419.
- 29. Hirschhauser, Zeit. Anal. Cham., 1885, 24, 157.
- 30. Gadamer, Chem Zeit., 1902, 26, 291.
- 31. Gadamer, Arch. Fharm., 1905, 247, 31.
- 32. Perkins, J. Chem Soc., 1918, 505.
- 33. Faltis, Monat., 1910, 31, 557.
- 34. Ahmad and Lewis, J. Pharm. Pharmacol., 1960, 12, 163.

Katritzky, Jones and Bhatnagar, J. Chem. Soc., 1960, 1950.

- 36. Kenner, Khorana, and Prelog, Helv. Chim. Acta, 1951, 34, 196
- 37. Takeda, Bull. Agric. Chem. Soc. Japan. 1956, 20, 165.
- 38. Kondo, and Sanada, J. Pharm. Soc. Japan, 1927, 541, 31.
- 39. Bellamy, "The Infra-Red Spectra of Complex Molecules", Methuon, London, 1962, (a) 96, (b) 34.
- 40. Gibbs, <u>J. Biol. Chem.</u>, 1927, <u>72</u>, 649; King, King and Manning, <u>J. Chem. Soc.</u>, 1957, 563.

- 42. Bick, Harley-Mason, Sheppard and Vemengo, J. Chem. Soc., 1961, 1896.
- 43. N. A.R. Spectra Catalog, Varian Associates, 1962, Spectrum number 232.
- 44. Battersby and Harper, J.Chem. Soc., 1962, 35, 26.
- 45. Battorsby and Grenock, J. Chem. Soc., 1961,25, 92.
- 46. Leonard and Locko, J. Amer. Chem. Soc., 1955,77, 437.
- 47. Unitoma, J. Pharm. Soc., Japan, 1962, 82, 1577.
- 48. Battersby, Proc. Chem. Soc., 1963, 189.
- 49. "Organic Synthesis", ed. Bachmann, vol. 25, Chapman Hall, London, 1945, p. 49.
- 50. Claisen and Fislab, Ann., 1913, 401, 52.
- 51. Hawthorne and Wilt, J. Org. Chem., 1960, 25, 2215.
- 52. cf. Sousa and Blum, J. Org. Chem., 1960, 25, 108.
- 53. Boit, "Ergebnisse der Alkaloid Chemie bis 1960," Akadamie-Verlag, Berlin, 1961, p.p. 216-229.
- 54. Burger in "The Alkaloids" ed. Manske and Holmes, vol. IV, Academic Press, New York, 1954, p. p. 29-71.
- 55. Divekar, Read, Vining and Haskins, <u>Canad. J. Chem.</u> 1959, <u>37</u>, 1970; Read and Vining, <u>Chem. and Ind.</u>, 1959, 1547.

- 56. Birch, Proc. Chem. Soc., 1962, 3.
- 57. Gilchrist, Hodges and Porte, J. Chem. Soc., 1962,1780.
- 58. Wallace, Porte and Hodgos, J. Chem. Soc., 1963, 1445.
- 59. Karrer, "Konstitution und Vorkommen der Organischen Pflanzenstoffe, Birkhauser, Basle, 1958, p. p. 466-476.
- 60. Holmberg, Ber., 1921, 54, 2389.
- 61. Bauer, Birch and Ryan, Aust. J. Chem., 1955,8,534.